# Water management of canola under tile drainage in the Canadian Prairies

By

Emeka Leonard Ndulue

# A thesis submitted to the Faculty of Graduate studies to the University of Manitoba in partial fulfilment of the requirements of the degree of DOCTOR OF PHILOSOPHY

Department of Biosystems Engineering Price Faculty of Engineering University of Manitoba Winnipeg

Copyright © 2022 by Emeka Leonard Ndulue

# Abstract

An effective water management system is necessary to address global food shortages, water scarcity, and increasing climate extremes. Southern Manitoba is a major crop production area with highly productive soils, flat topography, and seasonally high water table. The objectives of this research were to (i) determine the impacts of controlled drainage (CD), free drainage (FD), and no drainage (ND) on canola yield and quality, (ii) simulate water table depth (WTD) under canola production using the DRAINMOD model, (iii) evaluate the potential impacts of climate change on the hydrology and canola yield using the DRAINMOD model and (iv) assess the performance of the standard ET model under limiting conditions and fourteen empirical ET models in the region. The result shows large differences in yield between the years, suggesting that canola yield is significantly influenced by weather variables. In 2019 with normal average temperature and rainfall close to the long-term average, CD plots consistently yielded higher yields than FD and ND, with a significant difference with FD. As rainfall decreased and temperature increased in the following years, the impact of drainage, especially CD, becomes diminished, with no significant differences between the treatments. Also, the prevailing weather may have masked the oil quality parameters and nutrient dynamics across the soil profile, as there were no significant differences between the treatments. The WTD collected from the PESAI (Prairies East Sustainable Agriculture Initiative) site in Arborg was used to test the ability of the DRAINMOD model to predict WTD using the 2019 and 2020 canola growing seasons. Statistical analysis and graphical plots showed close agreement between the measured and simulated WTD. Economic analysis using the simulation results suggests that the 10 m drain spacing maximized the return on investment. The parameterized DRAINMOD model was thereafter run with downscaled climate model projections from CANESM2 for historical (1981-2010), midcentury (2041-2070), and late-century (2071-2100) periods under three representative concentration pathways (RCP2.6, RCP4.5, and RCP 8.5). Results showed an overall decrease in average relative canola yield under FD and CD in the future. The ET (evapotranspiration) is an essential input to hydrologic models. However, the weather variables required to compute the recommended ET model (FAO PM) equation are not readily available. The results showed that the FAO-PM model decreased with increased missing data, yielding average to poor results. The best performing models were Valiantzas-1, Valiantzas-3, Irmak, Valiantzas-2, and Priestly-Taylor models. New empirical coefficients were also developed to improve selected empirical ET models. Overall, this research is important for developing optimal drainage design variables for increased crop yield.

# Acknowledgements

I am eternally grateful to God for seeing me through my studies and enhancing my career. I would like to acknowledge my advisor, Dr. Sri Ranjan for always being there to support and guide me through this study and to my committee members, Dr. Cicek and Dr. Cattani for their inputs.

I want to specially appreciate my wife, Nana Goodness, my father, Mr. Chukwulobe Ndulue, my Mother Mrs. Ogechukwu Ndulue, my siblings, and my wonderful cousin Nkoli, for their prayers, unflinching support and encouragement.

I would also like to thank Dr. Afua Mante, for insights and fruitful research discussions during this journey. I thank Mr. Sampson Boateng, for coming through for me at critical times, and my encouraging lab mates; Thushyanthy Akileshan, Prabakaran Santhanam, Mujibur Rahman, Rahul Makkar, Jerrold Rentz, Amadeus, and Baiyan for accompanying me to the field.

I acknowledge the funding support from UMGF (University of Manitoba graduate Fellowship), GETS (Graduate Enhancement of Tri-Council Stipends) award, NSERC, Mitacs, the Canadian Agricultural Partnerships Program, Manitoba Agriculture and Resource Development, the Manitoba Crop Alliance, Barley and Canola growers association, and Grant A. Harris Fellowship. I also want to thank Timi Ojo and Sass Alison of Manitoba Agriculture for providing the weather data used in Chapter 6. I would like to express my sincere gratitude to Richard Heide of Hespler Farms and all the staff of Hespler Farms. I greatly appreciate the support I received from the PESAI team especially Dr. Nirmal Hari, James Lindal, and Eugene DeLorme,

I also want to thank my friends and relations especially Eze Henry, Jude Eneje, Oluchi and Lizzy Lawrence, Mrs. Eyosa Chukwuka and family, Babatunde Alabi, Mr. Tunde Ogunje, Anita, Ikenna, Dr. Okechukwu, Ugochukwu, and Bequie.

# Dedication

To my family and parents.

# **Table of Contents**

Abstract		ii
Acknowledgen	ients i	ii
Dedication	i	V
Table of conter	nts	v
List of Tables .		X
List of Figures		ĸi
List of copyrig	hted materials xi	ii
Chapter One In	troduction	1
1.1 Backgr	ound of the research	1
1.2 Objecti	ves of the research	3
1.3 Justifica	ation of the research	3
1.4 Scope of	of the research	4
Chapter Two L	iterature Review	5
2.1 Can	ola	5
2.1.1	History – Origin, Statistics, Production, Uses	5
2.1.2	Lifecycle	7
2.1.3	Canola Planting and Management	7
2.1.4	Nutrients Requirements	8
2.1.5	Factors affecting canola yield	9
2.1.6	Water requirement 1	0
2.2 Water n	nanagement - Drainage 1	2
2.3 Types of	f Drainage 1	3
2.4 Water ta	able management techniques 1	4
2.4.1	Free Drainage (FD) 1	4
2.4.2	Controlled Drainage (CD) 1	5
2.4.3	Controlled Drainage-Subirrigation (CDSI) 1	6
2.5 Require	ements of Water Table Management (WTM) 1	7
2.6 Integrat	ed Water Management 1	8
2.7 Model	Assessment 1	9

Chapter Three Canola yield and quality under tile drainage in the Canadian Prairies	21
Abstract	21
3.1 Introduction	21
3.2 Materials and method	23
3.2.1 Study Area	23
3.2.2 Experimental Layout	24
3.2.3 Agronomic practices - Tillage, planting, chemigation, and harvesting	25
3.2.4 Instrumentation and Data Collection	26
3.2.4.1 Weather	26
3.2.4.2 Water table Depth	27
3.2.4.3 Soil nutrient	27
3.2.5 Canola Quality Analysis	27
3.2.6 Statistical analysis	28
3.3 Results and discussions	
3.3.1 Weather variables	28
3.3.2 Canola Yield	30
3.3.3 Nutrient analysis	33
3.3.3.1 Soil Nitrate-N content	33
3.3.3.2 Soil Test Phosphate (STP)	36
3.3.4 Canola quality assessment	38
3.3.4.1 Oil Content, Protein Content, and Glucosinolate contents	38
3.3.4.2 Fatty Acid (FA) constituents	40
3.4 Conclusion	42

Chapter Four DRAINMOD simulation of drain spacing impact on canola yield in heavy soils in the Canadian Prairies	clay 43
Abstract	43
4.1 Introduction	43
4.2 Materials and methods	46
4.2.1 Study Area	46

4.2.2 Field instrumentation and data collection	. 47
4.2.3 DRAINMOD MODEL	. 48
4.2.3.1 Model inputs	. 49
4.2.3.1.1 Weather data	49
4.2.3.1.2 Soil	49
4.2.3.1.3 Drainage design	. 50
4.2.3.1.4 Crop data	52
4.2.3.1.5 Trafficability	52
4.2.4 Model performance	53
4.2.5 Assessment of drain spacing on predicted relative canola yield	54
4.2.5.1 Economic analysis	54
4.3 Results and discussions	. 55
4.3.1 Weather	55
4.3.2 Model parameterization	. 55
4.3.3 Water table dynamics	. 56
4.3.4 DRAINMOD Hydrologic long-term assessment	59
4.3.5 Impact of drain spacing on canola yield	62
4.4 Conclusion	65
Chapter Five Assessment of the potential impacts of climate change on the hydrology and Canola yield using the DRAINMOD model	1 67
Abstract	67
5.1 Introduction	. 68
5.2 Methodology	. 70
5.2.1 Site description	70
5.2.2 Experimental layout	71
5.2.3 Climate projections	72
5.2.4 DRAINMOD MODEL	. 72
5.2.4.1 Model setup, inputs, and parameterization	. 73
5.2.5 DRAINMOD simulations for Climate change assessment	73
5.3 Results and Discussions	. 74

5.3.1 Projected changes in climate data	74
5.3.2 Projected changes in the hydrology under free drainage and controlled draina	ge
5.3.2.1 Subsurface drainage	77
5.3.2.2 Runoff	. 78
5.3.2.3 Evapotranspiration (ET)	. 82
5.3.2.4 Water table depth (WTD)	. 82
5.3.3 Projected changes in relative yield of canola	. 83
5.4 Conclusion	. 87

Chapter Six Performance of the FAO Penman-Monteith Equation under limiting conditions and fourteen Reference Evapotranspiration Models in Southern Manitoba	89
Abstract 8	89
6.1 Introduction	89
6.2 Materials and methods	92
6.2.1 Weather data	92
6.2.2 The FAO-Penman-Monteith (FAO-PM) Equation	93
6.2.3 Empirical ETo models	94
6.2.4 ET <sub>o</sub> estimates under limiting conditions	96
6.2.5 Model calibration and validation	96
6.2.6 Sensitivity analysis	97
6.2.7 Model performance assessment	97
6.3 Results and Discussions	98
6.3.1 Climatic conditions of the study area	98
6.3.2 Relationship between ETo and weather variables	99
6.3.3 ET <sub>o</sub> assessment under missing data 10	00
6.3.4 Comparison of ET <sub>o</sub> models 10	02
6.3.3.1 Combination-based models 10	03
6.3.3.2 Radiation-based models 10	04
6.3.3.3 Temperature-based models 10	04
6.3.3.4 Mass transfer-based models 10	05
6.3.5 Model calibration and validation	08

6.3.5.1 Determination of Hargreaves solar radiation adjustment	
coefficient, Krs	108
6.3.5.2 Determination of empirical coefficients for ETo models	108
6.3.6 Changes of climatic variables to changes in ETo	110
6.4 Conclusion	111
Chapter Seven Conclusion	113
7.1 Main findings	113
7.2 Practical applications and contributions	114
7.3 Recommendations for future studies	115
References	117

# List of Tables

Table 3.1: Agronomic practices during the study period	26
Table 4.1: Soil hydraulic parameters	50
Table 4.2: Model inputs in DRAINMOD	50
Table 4.3: Effective root depth of canola used in the simulation	52
Table 4.4: Summary of statistics indices on DRAINMOD performance for simulating	
WTD	58
Table 4.5: Summary of annual average water balance	60
Table 4.6: Relationship between predicted relative yield and objective functions under	
different spacing	64
Table 5.1: Summary of statistical indices used to assess the DRAINMOD model	
(Adapted from Ndulue and Sri Ranjan 2022)	73
Table 5.2: Projected changes in the annual climatic variables	75
Table 5.3: Summary of DRAINMOD hydrologic simulation	80
Table 5.4: Canola yield, SEW <sub>30</sub> , Dry days, and work days under different climate	
scenario and water management system	86
Table 6.1: ETo models and their input requirements	95
Table 6.2: Daily average climatic variables at the study site from 2012-2019	99
Table 6.3: ETo computed with complete data sets Versus ETo computed with missing	
parameters	103
Table 6.4: Statistics indices using original and calibrated Krs for Rs and ETo	
computation	108
Table 6.5: Statistical indices of calibrated empirical models	109

# **List of Figures**

Figure 3.1: Field layout and assignment of treatments	25
Figure 3.2: Comparison of monthly rainfall and average temperature during the study	
period and 30-year average	29
Figure 3.3: Average yield per treatment throughout the study period (2019-2021)	30
Figure 3.4: Soil Nitrate-N content within the rootzone during 2020 and 2021 growing	
seasons	35
Figure 3.5: Soil Test Phosphate within the rootzone during 2020 and 2021 growing	
seasons	37
Figure 3.6: Average annual oil, protein, and glucosinolate contents per treatment during	
the study period (2019-2021)	39
Figure 3.7: Average annual proportions of fatty acids (Palmitic acid (C16:0), Stearic	
acid (C18:0), erucic acid (C22:1), Linolenic acid (C18:3), Linoleic acid	
(C18:2), Oleic acid (C18:1) per treatment during the study period (2019-	
2021)	41
Figure 4.1: Layout of the study area	47
Figure 4.2: Comparison of monthly rainfall and average temperature during the study	
period and 30-year average	55
Figure 4.3: Comparison of measured and predicted WTD in response to rainfall during	
the 2019 and 2020 growing season	58
Figure 4.4: Regression of measured and simulated WTD over the study period	59
Figure 4.5: Monthly average drainage and runoff under different drain spacing	61
Figure 4.6: Impact of drain spacing on the average annual water balance components	61
Figure 4.7: Predicted relative yield from 1986 to 2015 under different drain spacing	62
Figure 4.8: Yield reduction factors caused by wet, dry, and planting delay under	
different spacing	63
Figure 4.9: Effect of drain spacing on relative canola yield and average annual return on	
investment	65

Figure 5.1: Seasonal distribution of minimum temperature $(T_{min})$ , maximum temperature	
$(T_{max})$ , average temperature $(T_{avg})$ , and precipitation (PCP) for baseline,	
midcentury (2041-2070), and end of the century (2071-2100) periods under	
three climate scenarios (RCP 2.6, RCP 4.5, and RCP8.5). Uncertainty in the	
data is represented by the standard error	76
Figure 5.2: Relative yield due to different stressors (planting delay, wet stress and dry	
stress) as a function of time (baseline, mid-century, and late-century	
periods) under free drainage and controlled drainage under three climate	
scenarios (RCP2.6, RCP4.5, and RCP8.5)	85
Figure 6.1: Location of the study area from Manitoba Agriculture	93
Figure 6.2: Linear regression of $ET_0$ and (a). solar radiation (b). wind speed (c). vapor	
pressure deficit (d). average temperature (e). relative humidity	100
Figure 6.3: Regression of ET <sub>o</sub> computed with complete and incomplete datasets	101
Figure 6.4: Regression analysis between empirical $ET_0$ models versus FAO PM equation	
	107
Figure 6.5: Statistical indices for ETo computed using original and calibrated ETo	
equations	110
Figure 6.6: Percent change in each climatic variable to percent change in ETo	111

# List of copyrighted materials

Ndulue, E., & Sri Ranjan, R. (2021). Performance of the FAO Penman-Monteith equation under limiting conditions and fourteen reference evapotranspiration models in southern Manitoba. *Theoretical and Applied Climatology*, *143*(3), 1285-1298.

Ndulue, E. & Sri Ranjan, R. (2022) DRAINMOD simulation of drain spacing impact on canola yield in heavy clay soils in the Canadian prairies. Irrigation and Drainage,1–17. Available from: <u>https://doi.org/10.1002/ird.2683</u>

# **Chapter One**

### Introduction

#### **1.1 Background of the research**

Water is perceived as the most important natural resource because human life and other living things depend on it for survival. While it is the most common and abundant substance on earth, its distribution is limited in time and space (Cosgrove and Loucks 2015). The earth's surface is occupied by more than 70% of water. Canada is home to about 7% of the world's freshwater resources, with an average annual water yield of 3,478 Km<sup>3</sup> (Statistics Canada, 2017). Canada has about 11.7% (1,169,561 Km<sup>2</sup>) of the land area covered with water (Statistics Canada, 2017). It has the world's most extensive coastline of 243,042 km, as it is bounded by three oceans (Atlantic, Pacific, and Arctic) (Statistics Canada, 2017). From the above statistics, one can conclude that Canada is excessively rich in water resources. However, these are aggregate values and could be misleading. Water distribution in Canada is uneven temporally and spatially. Most of the water resources drain to the north. Only about 40% of the freshwater is found in the south, which has about 85% of the country's population (ECCC 2013). Some regions in the Canadian prairies receive significantly low annual rainfall (McGinn, 2010). The Canadian prairies in western Canada consist of three provinces, Saskatchewan, Alberta, and Manitoba; Canada's food basket is known for growing grains, oilseeds, and animal products.

The Canadian prairies play a crucial role in the total food production in Canada, contributing more than 50% of Canada's grain (Bueckert and Clarke 2013). The region has more than 42% of Canada's farms and more than 80% of Canada's farmland (AAFC, 2000). It is also the largest producer of canola, an important source of edible oil. Saskatchewan boasts about 53.7% of the total production in Canada, followed by Alberta (28.86%), Manitoba (16.31%) and British Columbia (0.6%) (CCC 2021). Manitoba produced more than 3.3 million tons in 2018. Canola is suited to the Canadian prairies, although crop yield and oil quality are susceptible to environmental stresses frequently experienced in the prairies. In Manitoba, excess water, drought, and heat account for a combined crop loss of 71% (MASC, 2020). Projected future climate scenarios for the region predict wetter springs and dryer summers (Sauchyn and Kulshreshtha, 2008). The winter snow and early spring rainfall could have severe consequences for soil workability, trafficability, timeliness of farm operation, and, consequently, total crop yield. According to Manitoba Agricultural Services Corporation

(MASC, 2015), a delay in planting up to the first week of June could led to a 20 - 30% reduction in yield depending on crop type. These challenges have necessitated Canadian producers, especially canola farmers, to seek sustainable water management strategies.

Agricultural drainage is an important water management technique that provides a conducive growing environment for proper crop development and growth by lowering the water table and removing excess water from the soil profile. The benefits of agricultural drainage are well documented in several textbooks and journals, some of which are increased crop yield and income, timely farm operations, early planting, improved soil health, soil properties etc. (Madramootoo et al. 1993, Fraser and Fleming 2001, Hudson and Roberts 1982, Huffman et al. 2011. Van Zandvoort et al. 2017). On the contrary, drainage has led to the loss of wetlands, mainly in North America, Europe, and Asia (Davidson 2014) and increased nutrient loadings leading to eutrophication and anoxic conditions (King et al. 2015).

It is essential to evaluate different water management systems for agronomic, economic, and environmental benefits. This is because the performance of water management systems varies from one location to another due to a variety of factors that are not fully understood (Allerhand et al. 2013). Field experiments and modelling are complementary tools that assist water managers, producers, and decision-makers in effective water management. Technological advancements in data collection and computing abilities have made hydrologic models attractive for many researchers. Models have been used to provide an understanding of the agro-hydrological process and guide decision-making processes such as water management, irrigation timing and scheduling, drainage design and management, nutrients transport, and climate change assessments (Phogat et al. 2014, Karandish and Šimůnek 2016, Gunn et al. 2018, Mante et al. 2018, Pease et al. 2017). It should be noted that model results are affected by uncertainties originating from model inputs, model assumptions, structure, and parameter uncertainty (Seiller & Anctil 2016). ET (evapotranspiration) is one of the most critical inputs to hydrologic models. In most hydrologic models, the reference evapotranspiration ( $ET_{o}$ ) quantifies the atmospheric water demand under ideal conditions (Allen et al. 1998). The FAO Penman-Monteith (FAO PM) equation has been identified as the standard model for determining ET<sub>o</sub> (Allen et al. 1998). However, the weather variables required to compute the FAO PM equation are not readily available. Hence, researchers have developed simple empirical ET equations that require fewer weather inputs but such methods are location specific. Thus, it is recommended to evaluate the suitability of an empirical ET<sub>o</sub> model before use in new environments (Tabari et al. 2013).

# **1.2 Objectives of the research**

The overall objective of this research was to determine the impact of different water management techniques, namely, controlled Drainage (CD), free drainage (FD), and no Drainage (ND) on yield and oil quality of Canola grown in southern Manitoba. Specific objectives include to:

- (i) assess the effects of different water management systems on canola yield and canola oil quality parameters (oil content, protein content, glucosinolates, and fatty acid composition)
- (ii) apply the DRAINMOD model to simulate water table depth under canola production and validate the model for the field.
- (iii) assess how climate change may affect the hydrology and canola yield of an agricultural field and
- (iv) assess the performance of the FAO PM equation and fourteen empirical ET models in southern Manitoba.

# **1.3 Justification of the research**

Since the last century, canola and oilseed production and areas under cultivation have been rising in Canada and other countries. There has been an exponential growth in exports, trade between countries on Canola and its products. Production in Canada has been projected to reach 25 million tonnes by 2025 (Morrison et al. 2016).

Although studies have shown that water management affects canola yield and quality, the response may vary considerably among production areas that differ in growing conditions, soil properties, varieties, drainage design, and other management practices. These variations justify the need for water management strategies that are location specific. Even though subsurface drainage has been practiced for centuries, especially in humid and semi-arid regions, it is relatively new in southern Manitoba. In recent times, producers in the area have begun embracing the agricultural tile drainage installation. Despite being the highest economic crop in the province, no study has quantitatively described the impact of water management on canola production in Manitoba. This research is a follow-up to previous studies on water management in potatoes (Satchithanantham 2013) and corn (Cordeiro 2014). The findings from this study are essential for canola growers regarding the best water management strategy that

will increase yield with minimal environmental impacts, especially in the face of an uncertain and changing climate. It is also important to build the local database for efficient and effective water resources management for growing Canola and decision making. The development of alternative ET models using weather data from the region would address the challenge of incomplete historical data needed for long-term modelling studies.

#### **1.4 Scope of the research**

The research evaluates one of the major problems limiting maximum production in the Canadian prairies and how water management can affect canola yield and quality in southern Manitoba. At the start of the growing season, excess water due to snowmelt may delay planting and field operations. Rainfall is uneven during the growing season, resulting in prolonged soil water deficit. The study attempts to contribute to knowledge by evaluating agronomic impacts of drainage on canola yield and quality in southern Manitoba, understanding water table dynamics through experimental and modelling exercises, predicting climate change impacts on crop yield under conventional and controlled drainage, and developing empirical constants for ET models.

This thesis contains seven chapters. Chapter one gives the general introduction, including the background of the study, objectives of the study, justification of the study, and scope of the research. Chapter two gives the literature review on the canola plant, growth and development, water management strategies, and hydrologic modelling. Chapter three evaluates three different water managements on canola yield and oil quality. Chapter four presents DRAINMOD simulations of the water table, including determining the optimum drain spacing and economic analysis. Chapter five assesses climate change impacts on crop yield under two water management systems (conventional and controlled drainage) using the parameterized model. Chapter six determined an assessment of the FAO PM equation under limiting conditions and calibration of empirical ET constant for southern Manitoba. Chapter seven presents the conclusion of this thesis, including the main findings and contribution to knowledge and recommendations for future studies.

## **Chapter Two**

#### **Literature Review**

# 2.1 Canola

Canola was developed by breeding a type of rapeseed to lower the erucic acid content by plant breeders at the University of Manitoba. The name canola was coined from **Can**adian oil, low acid. It is a dicotyledonous plant characterized by its yellow petals arranged in a crosslike pattern producing round seeds in small pods. Canola plants belongs to the genus *Brassica*, in the family of Brassicaceae, which is comprised of about 338 genera and 3700 species (Bailey et al., 2006). In Europe, Asia, and other regions, canola is also referred to as oil seed rape. For rapeseed to be called canola, it must meet the specifications set by the Canola Council of Canada (CCC). As defined by CCC (2021), Canola is '*seeds of the genus Brassica (Brassica napus, Brassica rapa or Brassica juncea) from which the oil shall contain less than 2% erucic acid in its fatty acid profile, and the solid component shall contain less than 30 micromoles of alone, or any mixture of, 3-butenyl glucosinolate, 4-pentenyl glucosinolate, 2-hydroxy-3 butenyl glucosinolate, and 2-hydroxy- 4-pentenyl glucosinolate per gram of air-dry, oil-free solid'.* 

# 2.1.1 History - Origin, Statistics, Production, Uses

Historically, rapeseed leaves and seeds, were used as potherbs, vegetables, spices, and condiments because of their hot flavour (Daun, 2011). It was also consumed as vegetable oil and lighting oil in Asia and Europe for almost 4000 years (Barthet, 2016). Rapeseed was first grown in Canada during the Second World War to augment supplies used to service steam and marine engines. After the war, a large quantity of rapeseeds continued to be grown and produced in Canada. Meanwhile, other oil plants such as soybean and sunflower did not perform agronomically well compared to rapeseed. The climatic condition in the prairies favoured rapeseed production since it required fewer heat units to grow compared to soybean and sunflower (Barthet, 2016). However, rapeseed had erucic acid and glucosinolate contents in significant quantities, which caused heart lesions in animals (Daun, 2011). In a research effort to eliminate these unwanted compounds and improve their taste and quality, canola was born. The development of rapeseed into canola is arguably the most important in rapeseed development. Through rigorous traditional plant breeding and field trials, the *B. napus* cultivar "Tower" was registered by Dr. Baldur Stefansson in 1974. It became the first rapeseed variety low in erucic acid and glucosinolate. In 1977, Dr. Downey and his research group registered

the "Candle" cultivar as the first *B. campestris* low in erucic acid and total glucosinolate. In 1978, the name canola was coined by the Rapeseed Association of Canada.

Global canola production and harvested areas have increased beginning in the last century. Production increased from about 1000 metric tons in 1943 to 2 million tons in 1977 (Barthet, 2016), then from 29.7 million tons in 1994 to 68.8 million tons in 2016. In the same vein, the cultivated area has increased from 22.72 million Ha in 1994 to 33.7 million Ha in 2016 (FAO 2018). This increase has been attributed to a number of factors, some of which are the use of herbicide-tolerant varieties, development of hybrids with better agronomic qualities, high economic returns, biodiesel potentials, and government support for green energy (Morrison et al. 2013, Harker et al. 2012). Canola is the second largest oil crop after soybeans, with a global production of 69.6 million metric tonnes in 2020 (USDA, 2022).

Canada is the largest producer of canola globally, with an annual production of 22.9 million metric tons in 2017 grown on 23.0 million acres (CCC, 2021). Canada produced about 24% of the world's Canola and is also the largest exporter, accounting for 64% of export trade (Cliff Jamieson 2021). The Canadian prairies account for more than 98% of the total canola production in Canada, while British Columbia, Ontario and Quebec account for the remainder. According to Statistics Canada (CCC 2021), in 2018, the Canadian prairies produced about 22,302,300 metric tons. Manitoba contributed about 14.9% (3,379,100 million tons) of the total Canadian amount and is the third-largest producer behind Alberta (6,679,200 million tons) and Saskatchewan (12,244,000 million tons). Canola is Canada's most valuable crop contributing about \$29.9 billion annually to the economy, creating more than 207,000 jobs and paying \$12 billion in wages (LMC International 2020). It is estimated that about 43,000 Canadian farmers grow Canola (CCC 2021). Canola oil is regarded as one of the world's healthiest oils due to its low saturated fat and high unsaturated fat contents, even though there are conflicting, inconsistent and inconclusive results on the health benefits (Lauretti and Pratico 2017, Lin et al. 2013).

Canola is mainly cultivated for its oil. Proximate analysis of canola oil indicates 48.4% oil and 22.5% protein (Barthet, 2016). After the oil has been extracted from Canola, the by-product, canola meal, is an excellent food supplement. Industrially, canola oil can be used to produce lubricants, oils, fuel, soaps, paints, bioplastics, cosmetics or inks (CCC, 2021). Studies have shown that Canola is a stockpile of bioenergy (Fore et al. 2011). When mixed with wheat, it can be used to produce ethanol. The canola seed can also be used to produce biodiesel. Canola

also has the potential for environmental reclamation and heavy metal decontamination (Ali et al. 2017).

#### 2.1.2 Lifecycle

Canola completes its lifecycle (planting to maturity) between 80 to 120 days depending on various factors, including cultivar, environmental variables, nutrients, management etc. According to CCC (2021), Canola has eight distinct growth stages, identified using the decimal classification (BBCH) method (Lancashire et al. 1991). They include the germination (growth stage 0), leaf development (growth stage 1), stem elongation (growth stage 3), inflorescence or budding (growth stage 5), flowering (growth stage 6), seeds and pod development (growth stage 7), ripening (growth stage 8), and senescence (growth stage 9).

# 2.1.3 Canola planting and management

Canola is planted by seeding. Under favourable growing conditions, germination occurs after 2-4 days of planting. Canola grows across a wide range of climates, soils, and environments in North America, Asia, Europe, Australia and Africa. Canola performs well on drained soils, with pH between 5.5-8.3, and moderately tolerant to salinity (Francois 1994, CCC, 2021). A shallow planting depth of 1 cm with good soil moisture was recommended by Harker et al. (2012). The benefit includes a shortened length of time to emergence, flowering and maturity. Canola has also shown to be an excellent rotation crop with small grain crops or tuber crops such as potatoes, or soybeans (CCC, 2021). The introduction of hybrid cultivars that are herbicide and disease tolerant provides an additional benefit of weeds, insects, and diseases control (Cathcart et al. 2006, Morrison et al. 2016).

Management practices, including planting date, planting density, tillage, and water conservation practices, have been reported to significantly affect canola performance (Kirkland et al. 2000, Angadi et al. 2004, Clayton et al. 2004, Hang et al. 2009, McKenzie et al. 2011, Abdullah 2014, Barthet 2016). In cold regions, planting is hindered by excess soil water due to snowmelt and low soil temperature. In Manitoba, Excess Moisture Insurance (EMI) protects against excessively wet conditions for producers who cannot plant before June 20. According to MASC (2020), \$623,106,473 was paid to farmers as claims due to excess soil water in the 2019/2020 growing season. Planting date is an important management decision producers could use to maximize environmental variables (temperature and moisture) and reduce heat and water stress during sensitive growth stages. Even though early planting has been

recommended (between middle to late April) (McKenzie et al. 2011, Clayton et al. 2004, Kirkland and Johnson 2000, Angadi et al. 2004), the optimal date is driven by historical frost risk, which varies with location. Benefits of early planting include efficient use of soil moisture, ensuring early flowering, avoiding heat and water stress, reducing the risk of frost damage in the fall and higher yield and oil quality (Angadi et al. 2004, Kirkland and Johnson 2000). In Australia, Kirkegaard et al. (2016) also observed that the traditional recommended planting date (usually mid to late May) decreased water use efficiency (WUE), increased frost risks, and reduced yield. Faraji et al. (2009) showed that early planting resulted in higher yield, more flowering days, longer days to maturity, higher above-ground dry matter, higher WUE and high harvest index. Their results showed that late planting decreased yield from 3780 to 582 kg/ha in 2005/2006 and 3543 to 162 kg/ha in 2006/2007. However, in Brazil, Confortin et al. (2019) reported that planting in mid-autumn (May 20) produced the highest seed and oil yield. Higher planting densities are hypothesized to increase yield and oil yield (Yousaf et al. 2002, Zhang et al. 2012, French et al. 2016). In a comparative study of different planting densities by Zhang et al. (2012), seed and oil yield increased by 10-20% and 12-18%, respectively. Likewise, in a 16-site study across western Canada, from 2010 to 2012, Gan et al. (2016) reported that, depending on the environment, higher planting density and more extended postflowering period increased yield. In the same way, O'Donovan (1994) found out that a high planting density of 200 m<sup>2</sup>/plant suppresses weeds. In Iran, Koocheki et al. (2014) observed that increasing plant density not only increased yield but also reduced nitrogen loss. Another important management factor is row spacing, as it affects access to nutrients and water, competition with weeds, and use of sunlight and energy. Ismail (2016) studied the optimum combination of irrigation, row spacing and intra-row spacing for growing canola. Full irrigation with a row spacing of 20 cm and an inter-row spacing of 10 cm optimized canola oil and seed production. This agrees with the report of Hu et al. (2015). They observed that 30 cm row spacing increased soil water content, canola yield and WUE in Saskatchewan. However, Potter et al. (2007) found no significant effect of row spacing on oil yield in Australia.

#### **2.1.4 Nutrients Requirements**

The major nutrients for crop growth and development are macro elements which include Nitrogen, Phosphorus, Magnesium, Sulphur, and Calcium. Fertilizer application rates depend on the available nutrients in the soil, as excessive loss of fertilizers is the major cause of water pollution, eutrophication, and algal bloom (Rabalais et al. 2007). Generally, canola plants require more N fertilizer than other oilseed plants (Singh et al. 2006). According to

Assefa et al. (2018), the nutrient requirement for canola is given as 62–12–45– 28 kg of plant N–P–K–S Mg/yield. Gan et al. (2007) reported that the nitrogen requirement for maximum yield varies between 106 to 162 kg N/ha depending on the cultivar. The effect of increased N rates on canola yield depends on weather, soil, moisture and variety (Ozer, 2003). Results from different studies suggest that high nNitrogen rates have resulted in increased canola yield by increasing the number of branches and flowers per plant, increasing leaf area index, extending the life duration of leaves, and increasing total crop absorption (Ozer 2003, CCC 2021, Wright et al. 1988). Ebrahimian et al. (2017) found out that N and Zn can be used to offset the negative effects of salinity on Canola. A mild application rate of 15 kg/ha of Sulphur and a high application rate of 120 kg/ ha of Nitrogen increased the WUE of Canola (Ngezimana and Agenbag 2015). Other microelements for the successful growth of Canola include boron, molybdenum, copper, zinc, and manganese (CCC, 2021).

# 2.1.5 Factors affecting Canola Yield

A lot of factors affect canola yield. While some researchers have studied only one factor (Ozer, 2003, Naghavi et al. 2015, BirunAra et al. 2011, Pavlista et al. 2016, Herget et al. 2016), others have investigated two or more factors (Younes, et al. 2011, Ismail, 2016, Kamkar et al. 2011, Koocheki et al. 2014). After reviewing more than 100 scientific publications on factors affecting canola yield, Assefa et al. (2018) ranked weather (environment) as the most important, followed by management, genetic and crop agronomic performance.

Environmental stresses, mainly temperature (heat) and water stress, significantly affect canola yield and oil quality. There is a general consensus in the literature that heat and drought stress have a very strong negative effect on canola emergence, growth, yield, and oil quality (Dreccer et al. 2018, Gan et al. 2004, Vigil et al. 1997, Elferjani and Soolanayakanahally 2018, Meng et al. 2017). Laboratory experiments and field studies have been used to evaluate the individual and combined effects of both stressors. The extent of their impacts is also dependent on the stage of development. The reproductive stage (flowering and pod development) is the most sensitive stage (Gan et al. 2004). A study by Tesfamariam et al. (2010) showed that water stress during the flowering stage reduced seed yield, oil yield and water use by 64.5, 14.6 and 53.2%, respectively, compared to the no-stress situation. Aksouh et al. (2006) demonstrated in a field experiment in Australia that exposing Canola to a short period of high temperature during the flowering stage reduced crop yield by 52% more than mild temperature stress over an extended period. This also agrees with the result of Faraji et al. (2009) in Iran. A study by

Hammic et al. (2017) showed that temperature and water stress had a more significant impact on canola yield than the nutrient condition of the soil. Of the two stresses, heat stress has the more significant detrimental effect on crop yield since Canola is a cool-season crop; therefore, it is highly susceptible to high temperatures (Angadi et al. 2000, Gan et al. 2004, Elferjani and Soolanayakanahally 2018). Elferjani and Soolanayakanahally, (2018) reported an 89% reduction in yield due to the combined impacts of drought and heat stress, while drought and heat stress alone resulted in 83 and 31% reductions, respectively. Gan et al. (2004) reported that high temperature decreased main stem pods, seeds per pod and seed weight by 75, 25 and 25%, respectively.

Canola plants under stress resulted in high protein content and low canola oil qualities such as increased concentration of glucosinolates, reduced growth, yield, and oil quality (Tesfamariam et al. 2010, Tohidi-Moghadam et al. 2011, Hammac et al. 2017, Rashid et al. 2018, Pokharel et al. 2020, Lohani et al. 2021). In a study by Champolivier and Merrien (1996), they observed a 48% reduction in yield because of water stress during the flowering stage. They further reported that oil yield decreased between 6 and 14% when subjected to water stress at different stages except for the vegetative stage, although protein content increased up to 14%. Tohidi-Moghadam et al. (2011) observed a 15% increase in glucosinolate content under water stress.

#### 2.1.6 Water requirement of Canola

The water requirement of Canola depends on several factors, including cultivar, climate, soil type, target yield, and management practices. Under full irrigation, canola water use was 702 mm in South Africa (Tesfamariam et al. 2010). It ranged from 449 to 462 mm in Turkey (Dogan et al. 2011), 582 mm in Nebraska (Herget et al. 2016), and 400 to 480 mm in Alberta (Alberta Agriculture and Rural Development 2011). Canola is usually grown under rain-fed conditions in southern Manitoba and many other regions, although canola yield increases with water availability. Faraji et al. (2009) reported that irrigation could cushion the effects of temperature stress, especially during the flowering stage, without yield loss.

Irrigation is reported to have increased canola yield, phenology, and yield components, including the number of leaves, branches, flowers and pods, crop height, stem length, stem diameter, canopy growth, 1000-seed weights, oil yield, and biomass (Dogan et al. 2011, Bilibio et al. 2011, Ghobadi et al. 2006, Faraji et al. 2009). With irrigation amounts of 10, 20, and 30 cm in semi-arid Nebraska, canola yield increased by 519 kg/ha, 1224 kg/ha, and 1686 kg/ha, respectively, compared with the rain-fed system (Herget et al. 2016). Faraji et al. (2009)

compared supplemental irrigation and rain-fed irrigation and reported that canola yield increased by 22.5 and 12.4% in the 2005 and 2006 growing seasons. Herget et al. (2016) reported that canola yield was five to 6 times under full irrigation relative to no irrigation. For areas with limited water supply, especially in an arid and semi-arid climate, deficit irrigation could minimize yield loss (BirunAra et al. 2011, Naghavi et al. 2015, Pavlista et al. 2016, Herget et al. 2016). Mousavi et al. (2010) implemented a different form of deficit irrigation called partial root zone (PRD). They reported no significant difference in yield, pod length, number of pods, branches and seed, 1000-seed weight and plant dry matter between PRD at 50% and full irrigation.

On the other hand, too much water in the soil is detrimental to plants with low tolerance to saturated conditions such as canola. The impacts of waterlogging on crops are well known and documented in literatures (Drew 1997, Jackson 2004, and Greenway and Gibbs 2003). Under saturated conditions, soil pores are saturated with water. Consequently, the exchange of gases is stalled, oxygen gas is depleted, CO<sub>2</sub> concentration increases, and photosynthesis (food production) is stopped because stomata cells are closed (Jackson 2004, Bedard-Haughn, 2009, Huang 2000). Waterlogging causes hypoxia, reducing root respiration and growth and reducing nutrient uptake (Barrett-Lennard 2003, CCC, 2021). It also leads to altered soil chemistry (Evans and Fausey 1999; Jackson and Drew 1984), exposure to soil-borne pathogens (Stolzy and Sojka 1984), and if not attended to, the crop dies. Waterlogging also affects soil structure, increases the risk of salinization, increases water table and saline seepage, limits proper root development, leads to roots clustering around a certain depth, increases the emission of greenhouse gases, and slows down nitrogen fixation (Bedard-Haughn, 2009). The extent of damage is also dependent on temperature, length of ponding and stage of development (Franklin et al. 2005, Ploschuk et al. 2018, Zhou and Lin 1995). However, canola cultivars respond differently under waterlogged situations (Ashraf & Mehmood 1990, Zou et al. 2014). Waterlogging hinders canola growth and development, which ultimately leads to decreased yield (Ploschuk et al. 2018, Zou et al. 2014). Zhou and Lin (1995) reported that water logging during the seedling and bud appearance stage led to a 21.3% decline in pods per plant and a 12.5% fall in seeds per pod. They further reported that flooding at pod formation and flowering stage did not significantly impact yield. A lysimetric study of waterlogging effects on Canola by Cannell and Belford (1980) showed that plant height, seed yield, oil yield and straw declined by 17, 14, 17 and 23%, respectively. Wollmer et al. (2018) conducted a greenhouse experiment to investigate waterlogging effects on yield and oil. Results showed that waterlogging during the vegetative development stage has more impact on yield and yield components than at the

flowering stage. Specifically, they reported that waterlogged conditions decreased yield by 25 and 15% during the vegetative and flowering stages, respectively but had no effect on oil quality. Zhou et al. (1997) observed that applying nutrients and growth regulators can offset the impacts of waterlogging on Canola.

# 2.2 Water management - Drainage

Waterlogging occurs because of too much water in the soil. The Canadian climate is prone to such, especially at the beginning of the growing season, because of the snowfall in winter. In the Canadian prairies, a summer storm of high intensity and short duration (Dumanski et al. 2015) could cause the water table to rise quickly within the root zone. Factors aiding this condition include poor internal drainage, low hydraulic conductivity, compaction, the soil type predominant in the region, and a shallow impermeable layer restricting infiltration (Madramootoo et al. 2007 Bedard-Haughn, 2009). The practical solution to this challenge is drainage. Drainage is an old technology used by ancient civilizations to remove excess water during times of flooding. It is estimated that 33% of the total agricultural land globally is under drainage (Smedema and Ochs 1997). It was introduced into North America by early settlers from Europe (Madramootoo et al. 2007) and can be found in almost every part of the world with drainage-related issues. Drainage systems and practices have evolved from open field ditches and canals (surface) to buried drainpipes (subsurface). It has also evolved from heavy clay and concrete tiles to corrugated plastic tubing installed with drainage machinery. The development of corrugated plastic tubing in the 1970s and the laser control drain plow, first developed in Canada, revolutionized the drainage industry (Irwin 1989, Madramootoo et al. 2007). Although drainage is usually practiced in regions with a humid climate, it is also advantageous for the semi-arid Canadian Prairies climate (Ayars et al. 2006, Mante et al. 2018, Dou et al. 2021).

Drainage removes excess water from the plants' root zone to provide a conducive environment necessary for root development and crop growth. Sources of excess water include rainfall, snowmelt, high water table and over-irrigation. Like most regions with poorly drained soils, drainage is essentially needed before a growing season so that the field is trafficable, and crops are planted at the right time. Studies by Satchithanantham (2014) and Cordeiro (2013) have shown that the water table can get as high as 0.4 m in southern Manitoba. This portends severe consequences for the timeliness of farm operations, crop production and total yield. According to Manitoba agricultural services corporation (MASC, 2015), a delay in planting up to the first week of June could lead to a 20 to 30% reduction in yield depending on crop type.

There are many advantages of agricultural land drainage. Benefits accruing from drainage installation on a wide variety of soils, crops, and climates abound in many publications and textbooks (Fraser and Fleming 2001, Hudson and Roberts 1982, Van Zandvoort et al. 2017, Madramootoo et al. 1993). Drainage ensures that agricultural land can be accessed in the shortest possible time, especially after heavy rainfall, without structural damage or compaction on the soil (Mante et al. 2018). Since the heat capacity of water is far greater than the heat capacity of the soil, drainage provides conducive warm soil temperature necessary for plant germination in the early spring. Soil temperature directly impacts seed germination, and growing degree days (GDD), an index used to predict the developmental stages of crops. Most crops, including canola, require an ideal growing soil temperature of  $\geq$  10 °C for germination (CCC, 2012).

# 2.3 Types of Drainage

There are two main methods of land drainage: surface and subsurface drainage. As the name suggests, surface drainage removes excess water that collects on the surface. Ponded water on the soil surface could be removed by surface drains, open ditches, land forming, land levelling, or combinations of the above. Surface drainage is usually practiced in areas with high rainfall intensities and soils with low infiltration capacity (Huffman et al. 2011). Surface drainage reduces crop loss and controls surface runoff and soil erosion (Brown and Ward, 1997). Although it does not remove water within the soil profile (NRCS 2001), it is simple, common (Skaggs et al. 1994) and a cost-effective drainage method (Manik et al. 2019) with a high benefit-cost ratio and high net returns (Colwell 1978). A comparative study of surface, subsurface and controlled drainage subirrigation (CDSI) by Grigg et al. (2002) in a sugarcane farm in Louisiana showed that surface drainage performed better than subsurface and CDSI by reducing nitrate losses and increased crop yield. Wanchuk and Apedaile's (1988) study showed that surface drainage is more profitable than subsurface drainage. However, Madramootoo et al. (2001) argued that surface drainage performs poorly compared to subsurface methods in terms of water removal as they also restrict the use of farm machinery. Also, there is evidence in the literature that surface drainage has led to increased nutrient enrichment of water bodies because of increased runoff and peak outflow relative to subsurface drainage (Schindler et al.,

2012, Rabalais et al., 2007, Blann et al. 2009, Smith et al. 2015, King et al. 2015, Stamm et al. 1998, Sharpley & Menzel, 1987).

On the other hand, subsurface drainage removes excess water below the soil profile. It is also used for salinity control and salt balance within the profile (NRCS, 2001). It uses drainage ditches and buried perforated pipes to lower the water table to the desired depth. It is applicable in areas with a high naturally occurring water table, poor internal drainage, and water-logged soils (Madramootoo et al. 2001, Huffman et al. 2011). Subsurface drainage increased agricultural activities in North America and other regions of the world (Gramlich et al. 2018) as artificially drained lands are reported to be the most productive (Skaggs et al. 1994). As the water table rises due to heavy rainfall or over-irrigation, gravitational water is drained out, and the table is kept within or below the tile drains. Through this process, the water table is quickly lowered to a depth conducive to root development and crop development, and so, field operation is unhindered or can be started.

### 2.4 Water table management techniques

The practice of manipulating the water table depth to remove excess water and salts from the soil profile is called Water table management (WTM). The advantages of WTM have been highlighted by many researchers (Sunohara et al. 2014, Schott et al. 2017, Jafari-Talukolaee et al. 2016, Jafari-Talukolaee et al. 2018, Vanderleest et al. 2016). WTM is implemented in three forms: free drainage (FD), controlled drainage (CD) and controlled drainage- sub-irrigation (CD-SI).

# 2.4.1 Free Drainage (FD)

FD, also called conventional drainage, functions by lowering the water table by gravity flow towards the drains by using shallow drains. This represents the earliest form of subsurface drainage, where the primary objective was to remove excess water from the root zone and increase crop yield. Schott et al. (2017) compared no drainage, shallow drainage, conventional drainage, and controlled drainage in Iowa for 5 years. Results showed that shallow drainage performed best by reducing flow and nitrate loss by 60 and 61%, respectively. In Nova Scotia, Smith et al. (2019) reported that shallow drainage reduced nitrate loading better than controlled drainage and conventional drainage by decreasing nitrate load by 54.9-73.1%. However, allowing free flow of drainage water to water bodies has led to severe environmental challenges ranging from nutrient enrichment, water pollution, eutrophication, and algal bloom (Gentry et al. 2007, Rabalais et al. 2007, Randall and Goss 2008). Discharging drainage water to water

bodies bypasses the natural denitrification and mineralization processes in soil. This contributes to increased nutrients, sediment, and pesticide loadings in water bodies. Notable cases include the Gulf of Mexico, China Sea, Lake Winnipeg, and the Great Lakes (Rabalais et al., 2007, Hawley et al., 2006, Schindler et al., 2012, Chen et al., 2007). However, depending on age and drain type, existing free drainage systems can be retrofitted with control structures (Strock et al. 2010)

# 2.4.2 Controlled Drainage

Controlled drainage, CD, also known as drainage water management (DWM), is the practice of placing control structures on subsurface drains at the outlet to regulate drainage water into water bodies. CD is regarded as one of the best management practices (BMPs). Numerous studies have reported the effectiveness of CD under different climates, soil types, and crops to include an increase in yield and income, reduction of drainage flow volume, decrease in nitrate and phosphorus concentration, disease control, salinity management, and improved soil structure (Evans et. al. 1989, Evans et al. 1995, Tan et al. 1999, Fausey 2005, Drury et al. 2009, Darzi et al. 2007, Mood et al. 2009, Crabbé et al. 2012, Fayrap and Koc 2012, Awale et al. 2015, Gunn et al. 2015, Mehring et al. 2015, Youssef et al. 2018, Smith et al. 2019). The reduction of nitrates is attributed to the denitrification process and enhanced mineralization that removes nitrates before reaching the ground or surface water (Elmi et al. 2000, Poole et al. 2013, Gramlich et al. 2018).

Management practices like tillage also affects the performance of CD. In a comparative study of Controlled Drainage-No Tillage and free drainage in a clay loam soil in Ontario, Tan et al. (1999) reported that nitrate loss decreased from 27.3 kg N/ha to 19.8 kg N ha while drainage volume decreased by 22%. A review of the literature showed that the performance of CD varies with soil, prevailing weather conditions, crops, and management. While it is successful in one location, it is insignificant or sometimes detrimental in another. In relation to other water table management, it is expensive and requires expert management to avoid crop failure and shallow root development (Huffman et al. 2011, Madramootoo et al. 2001). Despite its economic, agronomic, and environmental benefits, the CD has witnessed low patronage and adoption by farmers. An economic analysis by Chette et al. (2012) showed that CD has a payback of 3-4 years, yet farmers are still very uncertain (Madramatoo et al. 2007). Other factors contributing to low adoption include lack of extension services, increased labour, and cost (Dring et al. 2016, Huffman et al. 2011). Tan et al. (1998) conducted a 2-year field study on clay loam in Ontario. Their results showed that CD did not increase soybean. In addition, the CD has the potential for increased overland flow, salinization, and greenhouse gas (GHG)

emissions. Hornbuckle et al. (2005) reported that CD has the potential for increased salinization. CD also provides an alternative route to nitrogen losses either through gaseous or groundwater seepage (Sunohara et al. 2014) and has the potential to increase GHG emissions (Nangia et al. 2013). Kumar et al. (2014) observed GHG releases like CO<sub>2</sub>, N<sub>2</sub>O, and CH<sub>4</sub> in a poorly drained soil in Ohio, although at low concentration. Hydrologic models have been coupled with GCMs to assess future impacts of controlled drainage with climate change. Hanke (2018), via modelling, reported worse performance of CD under climate change. SWAT simulation results showed increased surface runoff leading to increased nutrient losses in Ontario. However, Van Zandvoort et al. (2017) reported no significant difference in GHG flux in CD and FD. Their results agree with Nangia et al. (2013) except for CO<sub>2</sub> emissions.

# 2.4.3 Controlled Drainage-Subirrigation (CDSI)

The principle of operation of controlled drainage allows irrigation to be practiced alongside it through capillary action. This is called Controlled Drainage - SubIrrigation (CDSI). It is a system that enables dual functions of irrigation and drainage. Under excess water conditions, the drainage mode is activated, and the subirrigation mode is used during water deficit. It effectively enhances crop growth and nutrient uptake, increases yield, and reduces nutrient loss in the field (Drury et al. 2009). Experiments have shown that CDSI improves crop yield and water quality, conserves water, and significantly reduces nitrate and phosphorus loss (Ng et al. 2001, Tan et al. 1998, Bonaiti and Borin 2010, Tan and Zhang, 2011, Drury et al. 2009). Tan et al. (1999) showed that CDSI effectively increases crop yield and agricultural water quality. In a field study in Ontario, the tomato was grown on sandy loam soil under two water table management: CDSI and Free drainage (FD). The experimental results showed that nitrate concentration and nitrate loss decreased by 38 and 37%, respectively, while tomato and corn yield increased by 64% and 11%.

Similarly, CDSI reduced drainage volume, nitrate concentration and Nitrate by 45, 6.3 and 50% relative to free drainage (Tan et al. 1998). Also, tomato yield was increased by 11%. Bonaito and Borin (2010) studied the combined effect of irrigation and drainage under four treatments from 1997 to 2003 in Italy. Results showed that treatments under CDSI conserved more water, had the lowest nitrate loss, and reduced drainage volume. Similarly, Tolomio and Borin (2018) combined open and subsurface drains with free and controlled drainage. Their results indicated that comparing CD with FD, discharge, nitrogen and phosphorus loss decreased by 81, 92 and 65%, respectively. CDSI is also an effective method for reducing and controlling phosphorus loss in the field (Tan and Zhang et al. 2011). They conducted a 5-year

study to study the efficiency of CDSI and FD in mitigating P loss. They reported that CDSI lowered Total Phosphorus and Particulate Phosphorus by 12 and 15%. Drury et al. (2009) compared three water management strategies in a corn-soybean field in Ontario with clay loam soil and two fertilization rates. Results showed that nitrate concentration and loss were less than the conventional drainage at both nitrogen fertilization rates, but the yield was also larger. The performance of subirrigation is weather dependent. Under very wet conditions, it has recorded poor performance. A 2-year field study in southwestern Québec by Madramootoo et al. (2001) recorded a 25% loss in corn yield in the first year. They recommended high management effort for subirrigated fields. In a four-year study in Ohio, Fausey (2005) found no significant difference in corn yield due to water table management. Satchithanantham (2013) found no significant impact on potato yield due to CDSI in southern Manitoba.

# 2.5 Requirements of Water Table Management (WTM)

For a successful implementation of WTM according to NRCS, (2001) the proposed site must meet the following conditions: flat topography, naturally occurring high water table, a shallow impermeable layer, negligible deep seepage loss, a suitable outlet, and an adequate water supply with good water quality. Past studies in southern Manitoba have reported that an impermeable layer mainly of clay lies below the water table at 6m below the soil surface (Cordeiro 2014, Satchithanantham, 2013). Design considerations include soil, topography, drainage intensity, water availability and water quality. The rate at which the drainage system removes excess water by lowering the water table height is dependent on drainage intensity (DI). It is defined as the drainage rate (cm/day) at the water table being lowered at some depth below the soil surface (Skaggs et al. 2006). DI is strongly affected by two main design factors: drain spacing and drain depth. Drain depth and spacing dictate the amount of flow from a subsurface drainage system. This directly affects crop yield, water table depth, soil moisture content, and most especially, water quality. There is a consensus in the literature, supported by field studies and modelling, that drains installed at a deeper depth and closer spacing increase drainage rate.

In contrast, shallow and or widely spaced drains result in lesser drainage rate (Skaggs et al. 2005, Skaggs et al. 2006, Cooke et al. 2002, Schott et al. 2017). Skaggs et al. (2005) plotted published findings of nitrate losses with drainage rates in Indiana and North Carolina. It was evident that a strong positive relationship exists between drainage rate and nitrate losses, although the magnitude varies. This variation was attributed to site conditions such as management, soil type and climate. By running a 50-year simulation in DRAINMOD, Skaggs

et al. (2005) reported that for soils with low hydraulic conductivity (K) values, drains need to be closely spaced to avoid excess water stress. Simulation results showed that relative yield starts decreasing at spacing more significant than 15 m, while for soils with higher K values, an 80% relative yield was obtained at a spacing of 35 m. Of the two design variables, drain spacing seems to have a lesser impact on drainage flow and nutrient transport (Skaggs et al. 2005, Tan and Zhang 2014, Madani and Brenton, 1995, Gremling et al. 2018). Tan and Zhang (2014) found that drain depth played a more significant role in P transport than spacing. They concluded by recommending that drain depth could be an effective method of minimizing P loss.

# 2.6 Integrated Water Management

A holistic approach is important for effective water management (UNESCO 1987, Kunz and Moran, 2016). It is a method that integrates parts into a whole system. The hydrologic cycle is made of different components, including precipitation, evaporation, runoff etc. Addressing the challenges or constraints of effective water management requires a fundamental understanding of the hydrologic cycle. Hydrologic models integrate these different components into a system to better understand the dynamics and complexities involved in natural systems. Agriculture is mainly dependent on factors that are variable and uncertain. Therefore, decisionmaking and management have necessitated the use of agricultural system modelling.

Agricultural system modelling allows simulations of the impacts of different management practices without actually implementing them. Modelling has become a powerful tool to provide researchers and scientists with both qualitative and quantitative answers. Models are effective tools used for investigating the impacts of agricultural practices and evaluating multi objectives such as agronomic, economic and environmental (Masasi et al. 2020). In recent times, the use of models in agricultural management has gained popularity. This has been attributed to technological advancements in computer and computer speed and developments in electronics and sensors for data collection (Beven 2012. There are numerous models developed within the last three decades, and some of them have minimal differences and similarities. Models vary in structure, processes and complexities (Ranatunga et al. 2008). Models have been broadly classified into empirical, mechanistic, deterministic, stochastic or distributed models (Ranatunga et al. 2008, Chow et al. 1998).

Different models are used to demonstrate other processes in agriculture and thus are used for various purposes. Numerous models exist and have been used in simulating crop growth and yield (AQUACROP), irrigation management and scheduling (SIMDUAL Kc), water management (DRAINMOD), surface water hydrology (SWAT), groundwater hydrology (MODFLOW), soil water dynamics and nutrient transport (HYDRUS), salinity (SALTMED), soil-water-plant continuum (SWAP) and water quality (RZWQM2). These are standard models used in agricultural water management, and most studies have found them reliable and accurate. Thus, they can be used as a management tool and assist in making decisions.

#### 2.7 Model Assessment

Model performances are assessed using a qualitative and statistical test. Qualitative assessment involves a graphical plot of observed versus simulated on a 1:1 scale with zero intercepts. Standard statistical tests used in hydrologic model assessment include the coefficient of determination (R<sup>2</sup>), NSE (Nash Sutcliffe Efficiency), root mean square error (RMSE), mean percent error (MPE) and mean absolute error (MAE). Details of the statistical indices are contained in subsequent chapters. Generally, NSE and R<sup>2</sup> values close to 1 and error indices (RMSE, MAE, MBE) close to 0 indicate a perfect model (Moriasi et al. 2015, Jackson et al. 2019).

# **Chapter Three**

# Canola yield and quality under tile drainage in the Canadian Prairies Abstract

For areas with seasonally shallow water tables and poorly drained soils, subsurface drainage systems are ideal for removing excess water from the root zone and improving soil workability, trafficability, and timeliness of field operations. With increased interest in tile drainage in southern Manitoba, the objective of this study was to evaluate the impacts of drainage on canola yield and canola oil qualities over three growing seasons (2019-2021) in Winkler, Manitoba. The study was carried out on replicated field plots with three different drainage treatments: controlled drainage (CD), free drainage (FD), and no drainage (ND). Subsurface drain tiles were installed at 8 m spacing for CD and 15 m spacing for FD, at a depth of 0.9 m. Compared to FD plots, the CD plots had significantly higher yields in 2019 with good rainfall. With deficient rainfall in the following years (2020 and 2021), the impact of drainage, especially CD, diminished, with no significant differences between the treatments. This could be related to the narrower drain spacing, rapidly removing water during short periods of high-intensity rainfall. The results showed no significant difference between the treatments but significant differences in canola oil quality parameters between the years, suggesting that environmental variables (mainly temperature and precipitation) may have masked drainage impacts. Keywords: Subsurface drainage, Water management, canola, Southern Manitoba.

#### **3.1 Introduction**

Canada, ranked fifth as a global exporter of many agricultural products, is a key player in global agricultural production (FCC, 2020). Agricultural production is concentrated in the Canadian Prairies (Saskatchewan, Alberta, and Manitoba). Canada is the largest producer and exporter of canola, accounting for 24 and 64% of the total global production and export trade, respectively (Cliff Jamieson 2021). The Canadian Prairies accounts for a significant amount of the total canola production in Canada. Canola is a type of rapeseed with significantly lower erucic acid and glucosinolates contents, making it fit for human consumption and animal feed. Canola is ranked as the second most-produced oilseed after soybean, with 69.6 million metric tonnes in 2020 (USDA, 2022). Canola production in Canada has increased steadily over the past two decades. The total production in 2018 (20.7 million tonnes) is more than 1.6 times the production in 2008 (12.6 million tonnes) (CCC 2022). This tremendous increase is due to an

increase in cultivated area, improved hybrid cultivars, biofuel potentials, management methods, and high economic returns (Morrison et al. 2016). In Manitoba, canola generated the highest economic return, contributing \$1.6 billion in 2021 (Manitoba Agriculture, 2022). Hydrologic extremes, including waterlogging, drought, and high temperature, are the major threats to canola production in the Canadian Prairies. In Manitoba, this accounts for a combined historical crop loss of 71% (MASC, 2020). Projected future climate scenarios for the region predict wetter springs and dryer summers (Sauchyn and Kulshreshtha, 2008). The accumulated snow in the winter and early rainfall in the spring could have severe consequences on soil workability, trafficability, timeliness of farm operations, and consequently total crop yield. According to Manitoba Agriculture, a delay of planting up to the first week of June could lead to a 20 - 30% reduction in yield depending on crop type. These challenges have necessitated Canadian producers, especially canola farmers, to introduce sustainable water management practices.

Agricultural drainage is an important water management technique that provides a conducive growing environment necessary for root development and crop growth by lowering the water table and removing excess water within the root zone. Although drainage is mainly practiced in humid climates, it is also of great importance in semi-arid climates such as the Canadian Prairies in controlling salinization (Dou et al. 2021). The benefits of agricultural drainage are well documented in the literature, two of which are an increase in crop yield and a decrease in nutrient and drainage outflows (Fraser and Fleming 2001, NRCS 2001, Huffman et al. 2011). On the other hand, drainage has led to the loss of wetlands, mainly in North America, Europe, and Asia (Davidson 2014). Subsurface drainage has also led to increased nutrient loadings and eutrophication of most water bodies (King et al. 2015). Notable cases include the Gulf of Mexico, China Sea, Lake Winnipeg, and the Great Lakes (Hawley et al., 2006, Rabalais et al., 2007, Schindler et al., 2012, Chen et al., 2013). To reduce nutrient loadings, controlled drainage (CD), was developed, which is the practice of placing control structures on subsurface drains at the outlet. Early studies on CD can be traced to the United States in the 1970s. Due to its enormous benefits, it has been recommended and adopted in various countries, including Canada, Italy, Sweden, Australia, China, and Iran (Wesström et al. 2014, Sunohara et al. 2016, Tolomio and Borin 2018, Jouni et al. 2018, Dou et al. 2021).

Numerous studies have reported varying results on the agronomic and environmental impacts of different water management systems. Using CD, for example, which is regarded as one of the best management practices (BMPs), a number of studies have reported positive benefits, including increased yield and income, decrease in drainage flow, drop in nitrate and phosphorus export, disease control, salinity control, and improved soil structure (Tan et al. 1999, Fausey 2005, Darzi et al. 2007, Drury et al. 2009, Skaggs et al. 2012, Cordeiro et al. 2014, Satchithanantham et al. 2014, Awale et al. 2015, Gunn et al. 2015, Mehring et al. 2015, Sunohara et al. 2016, Smith et al. 2019, Wang et al. 2020, Mourtzinis et al. 2021, Dou et al. 2021, Helmers et al. 2022). On the contrary, negative results have also been reported for CD, including lower yield, increase in surface runoff, alternative routes to nutrients losses, increased potential for salinization and increased GHG emissions (Tan et al. 1998, Hornbuckle et al. 2005, Nangia et al. 2013, Kumar et al. 2014, Hanke 2018).

The above literature suggests that the performance of a water management system may be location-specific, which can be linked to a variety of factors that are not fully understood (Allerhand et al. 2013). Therefore, an understanding of the underlying factors affecting drainage systems is critical. It is also important to continuously evaluate different drainage systems, especially with global changes in climate variables, climate variability, and uncertainty, even as the Canadian Prairies are vulnerable to climate change impacts (Qian et al. 2012). This study was also necessitated by the increasing number of agricultural drainage system installations in the region. Therefore, there is the need to develop local datasets of crops commonly grown for long-term simulation modelling studies and develop alternative management practices, including free drainage (FD), controlled drainage (CD), and no drainage (ND-control), on canola yield and canola oil quality, including oil content, protein content, glucosinolate content and fatty acid composition, in Winkler, southern Manitoba. The prevalence of flat topography and seasonal shallow water tables in southern Manitoba provides an opportunity to implement the three different water table management treatments.

# 3.2 Materials and method

## 3.2.1 Study Area

Field data were collected on research plots at Hespler Farms, located on latitude 49.12°N; longitude 97.93°W, in southern Manitoba (Fig. 3.1). The region has a semi-arid climate, with average annual precipitation of 563.3 mm. Historical weather analysis obtained from the closest weather station (Schanzenfeld), located 10 km from the study site, revealed that rainfall and snowfall account for 80 and 20% of the precipitation. Also, the historical analysis showed that rain and snowfall have peak values in June (100.1 mm) and December (22.9 mm),
respectively. The area has an average annual temperature of 4.0 °C, with the lowest monthly temperature of -14.6 °C in January and the highest temperature of 20.0 °C in July (Environment Canada, 2019). The soil in the study area belongs to the Gleyed carbonated Rego black subgroup of the Reinland soil found in Morden-Winkler (Smith et al.1973). The soil is classified as imperfect to moderately well-drained (Smith et al.1973). The soil belongs to the sandy loam textural class with 69% sand, 20% silt, and 11% clay. The average bulk density, field capacity, porosity, and drainable porosity are 1.38 g/cm<sup>3</sup>, 0.31 m<sup>3</sup>/m<sup>3</sup>, 47.8%, and 14.1%, respectively (Cordeiro, 2014). These average values were obtained over the 1.2 m profile (Cordeiro, 2014). It also has a distinct soil colour change from black soil in the top 30 cm to light-coloured (brown) soil at depths > 60 cm.

#### **3.2.2 Experimental Layout**

The whole field is approximately 5.2 ha, with canola and soybean in a 3-year rotation (2019-2021). Canola was planted in the eastern section in 2019 and 2021 and the western section in 2020. A buffer strip of about 4 m separated both crops. Each section measuring 84 m \* 300 m is divided into three, with each representing a replicate (Figure 3.1). Each replicate contained the three treatments: Controlled drainage (CD), Free drainage (FD), and No drainage (ND) as the control. The ND has an area of 0.2 ha (40 m\* 50 m), CD 0.2 ha (40 m\* 50 m), and FD 0.23 ha (44 m\* 50 m). For drained plots, (FD and CD) drain tiles were installed 90 cm below the soil surface, parallel to the planting row in the North-South direction. The drain tiles are corrugated plastic pipes with 0.1 m diameter and 50 m long connected to the submain, about 0.4 m in diameter. On CD plots, drain tiles were spaced 8 m apart. Drainage control structures (Agridrain Corp, Adair Iowa), with adjustable stop logs, were placed at the outlet of the submain to conserve water and limit drainage flow. On FD plots, drain tiles were spaced at 15 m, connected to the submain, which discharged to the outlet separately. All the plots received equal amounts of water as canola is usually cultivated as a rainfed crop.



Figure 3.1: Field layout and assignment of treatments

# 3.2.3 Agronomic practices - Tillage, planting, chemigation, and harvesting

Table 3.1 shows the agronomic practices, including tillage equipment, planting, harvest, fertilizer application, and chemigation during the study period. Plant rows were arranged parallel to the drains (North-South direction). Canola (c.v. 'Invigor') was seeded at 5cm depth, at a 5 lb/ha rate, on an inter-row spacing of 25 cm and planting spacing of 10-12 cm. Fertilizers were applied based on soil tests, usually done after harvest in the preceding year. The 2019 fertilizer application was done based on a soil test in September 2018. All treatment plots received the same fertilizer amount each year.

Harvesting was done on 533.4 m<sup>2</sup> area (Fig. 2, that is, the length of the combine swath (10.7 m) and width of the plot (50 m)) on each plot using a combine harvester (John Deer s680). The combine passed through centred on the location of the piezometers for measuring the water table elevation. The yield per plot (Mg/ha) was calculated as the quantity of seed weighed on the combine as it moved down the strip of the plots.

	Tillage	Planting	Harvest					Chemical	Application	Application
Year		date	date	Ν	Р	S	Κ	(product)	rate (L/ac)	date
	John Deere 730-									
2019	disc drill	May 2nd	Sep-15	115	20	10	0	(liberty)	1.5	Jun-06
		5	1					< J/	1.5	June 14
								Fungicide (Proline)	1 126	Jun-26
								Desiccant	1.120	Juli 20
								(glyphosate)	1.0	Aug-14
	Heavy							Herbicide		0
2020	Harrow	May-18	Sep-09	147	0	4	31	(liberty)	1.5	June 11
		•						Herbicide		
								(RT540)		Aug-20
								Fungicide		
								(Rovral)	0.84	July 2
								Fungicide		
								(Rovral)	0.42	July 9
	No- tillage. The seed was							herbicides- liberty No fungicide was applied		
	nlanted							because of the		
	on							extreme drv		
2021	stubble	May-04	Sept. 2	127	31	15	29	weather	1.2	June 4
		~	Ŧ						1.5	Jun-15

Table 3.1: Agronomic practices during the study period

# **3.2.4 Instrumentation and Data Collection**

# 3.2.4.1 Weather

An on-site weather station located about 150 m from the study area collected daily maximum and minimum temperature, precipitation, and evapotranspiration. Additional weather data, including relative humidity, wind speed and solar radiation, were collected on an hourly time step from the Winkler weather station, managed by Environment Canada, located about 12.53 km from the study area. The data from both sources provides complete weather information of the study area.

#### 3.2.4.2 Water table depth

Within each replicate, three wells were installed to measure groundwater depth. A total of nine wells were installed on the canola field. Watertable depth was measured using water level sensors (Solinst Levelogger Junior 3001, Solinst, Canada, Ltd., Georgetown, Ontario, Canada) suspended inside a piezometric well. Also, a barometric pressure sensor (Solinst Barologger) was used to correct the readings of the levelogger for atmospheric pressure. A 0.1 m diameter hole was dug to about 2 m using the hand auger. The piezometers are made of schedule 40 steel pipe, with an internal diameter of 0.0413 m and pointed edges. A Kevlar rope was used to suspend the sensor, tied to a ring (0.038 m diameter) hung on a bolt located 0.038 m from the top edge of the piezometer. The sides have slits covered with a geomembrane to allow easy movement of water and prevent soil and clay particles from entering the piezometer. The side shave slits covered with a geomembrane to allow easy movement of water and prevent soil and clay particles from entering the piezometer. The piezometers were inserted in the auger holes, and an offset of about 0.3 m was left above the soil surface. Bentonite sand mixture was packed into the annular space after installation to prevent preferential flow along the length of the piezometer. The piezometers had caps on the top to prevent rainwater entering from the top. Sensors were installed and set on a 3-hr logging interval. On drained plots, the wells were located mid-spacing between two drain tiles.

# 3.2.4.3 Soil nutrient

Soil samples were collected and analyzed on all the treatment plots for soil nitrate-N and soil test phosphorus-P (STP). A soil auger was used to collect the soil samples at 30 cm, 60 cm, and 90 cm depth during different developmental stages. The samples were sent to the Agvise laboratory for analysis. The soil nitrate-N and phosphate-P were extracted using the cadmium reduction nitrate method and the Olsen phosphorus method, respectively (Gelderman and Beegle 2015).

# **3.2.5 Canola Quality Analysis**

Oil, protein, and glucosinolate content were analyzed using Near-Infrared Spectroscopy (NIRS), which works on the principle of reflectance. The NIR spectroscopic analysis was done using a NIR scanning monochromator (XDS Rapid content <sup>TM</sup> Analyzer). About 5 - 7 g (n = 30) of canola seeds free of debris, were collected in the standard ring cup (diameter = 0.014 m, height = 0.01 m). Spectra data were recorded as the logarithm of reciprocal reflectance in the wavelength range of 400 - 2500 nm at 0.5 intervals. The scanning process per sample is approximately 1 min. The data was analyzed using the WinISI II software.

The fatty acid composition was determined by gas chromatography (GC) of fatty acid methyl esters. Three subsamples from each plot were used for this analysis. The GC procedure is given below. First, about 300 mg of canola seeds were crushed in a test cylinder and placed in a 13 × 100 mm test tube, 3 ml of heptane was added and allowed to stand overnight to extract oil, after which supernatant was decanted into a 13×100 mm test tube the next day. This was followed by adding 500 ul of 0.5 N sodium methoxide reagent. The mixture was shaken for 30 minutes. This separates the fatty acid from the mixture. After this, 100 ul of acidified water (0.3% acetic acid) was added, mixed gently and was allowed to clear by putting it in the fridge for 102 hours. About 500 ul of the mixture was put into a 2 ml autosampler vial for analysis. GC was performed on a 3900 Varian model fitted with CP-8400 autosampler and flame ionization detector. Six major fatty acids (Palmitic (C16:0), Stearic (C18:0), Oleic (C18:1), Linoleic (C18:2), Linolenic(C18:3), and Erucic (C22:1)) were analyzed and expressed as percentages of the total fatty acids. Peak areas were measured using the Varian star workstation software system. Canola quality tests were done at the Plant Science Department, Oil Quality Lab, University of Manitoba.

#### **3.2.6 Statistical analysis**

Field data, including canola yield and nutrients (nitrates and phosphates), were analyzed separately for each year and the 3-year average by conducting an ANOVA test using the JMP software (Version 16, SAS Institute, Inc, Cary, N.C). ANOVA was used to check the differences in treatment means for each measured field data. Soil nitrates and phosphates tests were analyzed for each sampling date. The mean values of the field data were separated using the using Tukey's test at a 5% (p<0.05) significance level.

# 3.3 Results and discussions

#### 3.3.1 Weather variables

The observed growing season rainfall (May to September) during the study period (2019-2021) was compared with the 30-year (1981-2010) monthly average. Figure 2 summarizes the average monthly weather during the study period and the long-term periods. Apart from the 2019 growing season with a total rainfall of 374.2 mm, close to the long-term average (376.5 mm), the 2020 and 2021 growing seasons were 35.6 and 38.9% lesser than the long-term average. The total growing season rainfall impact may have been masked by the monthly

variation observed within the season. Monthly analysis showed that the study periods deviated significantly from the monthly historical data. For example, in September 2019, the site received more than three times (151 mm) the long-term value (42.2 mm). Apart from July 2020, with rainfall 14.9% greater than the long-term average, all the other months were substantially lower than the long-term values. In 2021, May, June, July, and September received 49.6, 55.6, 70.3, and 47.9% less than the long-term values. The average temperature for the 2019 growing season (16.4 °C) is almost the same as the long-term average (16.6 °C), slightly higher (17.1 °C) in the 2020 season and significantly warmer in 2021 with an average temperature of 18.0 °C. The monthly temperatures during the study periods were greater than the long-term average, except in May. The figure also showed an increasing temperature trend across all the months except in September, where 2019 was slightly higher than 2020. Also, peak average temperature values were recorded in July, corresponding to periods of highest water use and least values in May.



Figure 3.2: Comparison of monthly rainfall and average temperature during the study period and 30-year average.

#### 3.3.2 Canola yield

Figure 3.3 shows the average annual yield per treatment. The result shows large differences in crop yield between the years, suggesting that yield may have been influenced by weather variables. The average yields during the study period were 3.2 Mg/ha in 2019, 2.9 Mg/ha in 2020, and 1.3 Mg/ha in 2021. Relative to the 2019 growing season, there was a consistent decrease in yield across the treatments and years. This could be linked to unfavorable weather variables and drainage design (drain spacing).



Figure 3.3: Average yield per treatment throughout the study period (2019-2021)

Means followed by different letters are significantly different based on Tukey's means separation at 5% significance level.

In 2019, the yield per plot differed among treatments (p = 0.0143). The average yields per treatment were 3.27, 3.51, and 3.02 Mg/ha for ND, CD, and FD, respectively. For each replicate, CD plots were consistently higher than other treatments. Statistical analyses revealed that there was a significant difference between CD and FD. Although there was no significant difference between CD and FD. Although there was no significant difference in yield between CD and FD. The excess soil water generated as a result of high intense rainfall was quickly removed through the drain tiles since the CD plots are closely spaced, thereby providing a quicker return to good growing conditions for the crop. Another reason is the cumulative benefits of CD. Studies have shown that CD has long-term benefits that may be hidden in short study periods (Thorp et al. 2008, Cooke and Verma, 2012).

In 2020, the average yields per treatment were 3.12, 2.52, and 2.97 Mg/ha for ND, CD, and FD, respectively. There was no significant difference among the treatments (p>0.05). CD had the lowest yield because of high variability within the replicates. Another reason for low average yields in CD could be linked to the drain spacing. Since the 2020 and 2021 growing seasons were dry, the closely spaced drains in CD plots could have intercepted and removed the infiltrating rainwater, making it unavailable to the plant roots. Also, controlled drainage plots may have removed the excess rainfall received late in the 2019 growing season resulting in less soil water for the 2020 growing season. This together with significantly low rainfall in spring could have adversely affected yield. Closely spaced drains tend to draw water rapidly from the profile. Unlike the ND plots, where the groundwater table is recharged by the rainwater, and through capillary action, plants roots can meet their water requirements. Moreover, since the soil has "imperfect" internal drainage (Smith et al. 1973) with heterogeneous texture and structure, a capillary barrier could easily form the layers. This creates a dry layer in the soil profile despite the presence of groundwater. The dry layer is formed due to the mismatch between high ET demand and upward flux from groundwater. Kross et al. (2015) noted that the inherent and intrinsic soil properties are crucial to understanding how crop yields are affected by water management practices.

In 2021, the average yields per treatment were 1.14, 1.52, and 1.07 Mg/ha for CD, FD, and ND, respectively, with no significant difference between the treatments due to high variability within the replicates. The results showed a further decrease in average yield by 61.6 and 56.2% relative to the 2019 and 2020 growing seasons, respectively. The inconsistency of the combine harvester under low yields could have led to the underestimation (personal communication with the farmer), even though canola production in Canada fell by 35.4 % (12.6 million tonnes), the lowest level since 2007 (Statistics Canada, 2022). This was due to the extreme drought and heat stress in the Canadian Prairies. The 2021 growing season started with deficit soil moisture content due to the low snow accumulation in the preceding winter, low spring runoff, and spring rainfall. The drought and heat continued into the developmental and reproductive growth stages (budding, flowering, and pod development). In June and July, the rainfall was 55.5% and 70.4% lower when compared with the long-term monthly average. Also, in 2021, there were 20 days with Tmax> 30 °C in June, seven days with Tmax> 30 °C in July, and one day with Tmax> 30 °C in August, respectively. Also, there were two days each with Tmax > 35 °C in June, July, and August. Using 29.5 °C as the threshold temperature, the heat stress index, which is defined as the summation of the difference between the maximum temperature and the threshold temperature (Morrison and Stewart 2002), the total heat stress index calculated from June to September in 2021 (126.9) was almost thrice (45.3) and twice (63.8) the 2019 and 2020 growing seasons, respectively.

In the field, both high temperatures and drought coexist. They reduce crop yield by altering the biochemical, molecular, and physiological processes responsible for growth and development, such as photosynthesis, respiration, flowering, pollination, seed filling (Prasad and Staggenborg 2008). While water stress would reduce the ability to obtain the nutrients needed for the development of reproductive organs, photosynthesis, and carbon assimilation, heat stress disrupts reproductive processes such as embryo sac differentiation, pollen gametogenesis, and embryo development, ovule fertilization, endosperm development, etc. resulting in the reduction of the number of flowers, number of pods, seeds per pods, and number of pods per plant which ultimately results in low yield. (Morrison et al. 2002, Gan et al. 2004, Hammac et al. 2017, Elferjani and Soolanayakanahally, 2018).

Generally, the prevailing weather conditions could affect the impacts of drainage on the crop. In this study, the beneficial effects of drainage, especially CD, appeared to decrease with increasing drought conditions when the control structures are not set correctly to hold back the water. This agrees with other studies that state that the effects of CD are minimal, hidden, or insignificant under drought conditions due to the inability to store soil water within the profile (Kross et al. 2015). Skaggs et al. (2012) explained that CD benefits crop yield by retaining water that would have been lost via drainage and making that water available to the plant at later times. The yield results in this study agree with previous studies in the same area (Cordeiro 2014, Satchithanantham 2013 and other related studies (Helmers et al. 2012, Schott et al. 2017, Acharya et al. 2019). In 2012 with low rainfall, Cordeiro et al. (2012) and Satchithanantham et al. (2012) found no significant difference between the treatments. Acharya et al. (2019) studied the impacts of drainage and other management practices (tillage and crop rotation) on corn and soybean yield from 2014-2017 in North Dakota. Drainage treatments studied included ND, FD, and CD. They reported that rainfall throughout their study period was less than the 30 - yearaverage, varying between 17 to 46%. Their results showed no significant difference in soybean yield between the treatments implying that drainage had an insignificant impact on soybean yield.

Conversely, corn yield was not affected by drainage for years with significantly low rainfall (2016 and 2017). In a 5-year (2011-2015) study in Iowa, Schott et al. (2017) reported no significant difference between the drainage treatments during 2011, 2012, and 2013 growing

seasons. Also, Helmers et al. (2012) reported no significant difference in corn yields between the treatments during 2007, 2008, and 2010 growing seasons and no significant difference in soybean yields during the 2007 and 2008 growing seasons. These studies have demonstrated that drainage impacts on crop yield are dependent on the prevailing weather (rainfall). On the other hand, Poole et al. (2013) reported an average increase of 11 and 10% for corn and soybean yields, respectively, when compared with the conventional drainage but no significant impact on wheat yield.

Overall, the results suggest that the impacts of drainage diminished with increasing dry conditions. In 2019 with good growing season rainfall, CD plots were significantly different from FD and consistently had higher yields than FD and ND. As drying persisted in the following years, the benefits of CD were not apparent, even though there were no significant differences between the treatments. Given that the growing season has been predicted to be dry in the summer and that drying will persist in the future, under rainfed conditions, the results in this study suggests that CD should be operated in a way to limit the outflows from drain tile during periods of rainfall to improve soil storage. During the springtime and heavy rainfall periods, the drain outflow could be pumped into water storage structures that could be reused during prolonged dry periods. This would provide both environmental and agronomic benefits. Careful management of the drainage control structures to conserve water is needed.

#### **3.3.3 Nutrient analysis**

### 3.3.3.1 Soil Nitrate-N content

Figures 3.4 show the soil nitrate-N content within the root zone for each treatment at different growth stages during the 2020 and 2021 growing seasons. The ND plots had higher nitrate concentrations relative to drained plots (FD and ND) for all the different growth stages. Even though ND was higher, the differences were not statistically significant (p>0.05). This may be due to the high variability of the data (see standard error value in Figure 3.4).

In this study, soil Nitrate-N content seems to vary with rainfall events. Samples collected after rainfall events had lower concentrations when compared with those collected after prolonged dry periods. Results also showed that the averaged nitrate content was higher than those reported in previous studies (Abbas 2015). This indicates low nitrate release and uses due to low soil moisture, resulting in low yields. Favourable temperature and sufficient soil moisture are needed for nutrient release (Abbas 2015). Given that nitrogen fertilizer has poor efficiency

because less than half of applied N-fertilizer is absorbed by the plants (Wiesler 1998), under drought conditions, canola produces less yield leaving more residual N in the soil, which accumulates via soil mineralization and nitrogen fertilization (Randall et al. 2005, Maaz et al. 2016).

Depth analysis showed increasing nitrate-N content as depth increases. The statistical test showed that the 90 cm nitrate-N content was significantly higher than other depths, especially during the vegetative and flowering growth stages. This may be because of the plant's root depth. Another factor is the nutrient-rich groundwater found in the study area. A previous study measured nitrate content in the groundwater and reported high values varying from 55 to 90 ppm (Haider 2005). Where no significant difference exists (maturity and after harvest), nitrate-N at the 90 cm depth was still greater.

The 2021 soil nitrate-N analysis was like the preceding year (Fig. 3.4). This may be because both years were dry. The results also showed no significant differences between the treatments even though ND plots had higher nitrate content than drained plots. However, depth analysis showed that soil nitrate was concentrated on the topmost layer and decreased with depth during vegetative and flowering growth stages. This corresponded to June and July, with significantly low rainfall. The third and fourth sampling periods saw a reversal of nitrate concentration between the 90 and 30 cm depths, with an increasing trend. This corresponded to August and September, which received significant rainfall amounts. This may be due to the following factors: rainfall washing down the nitrate-N content, capillary flux, or crop use at the later stages.



Figure 3.4: Soil Nitrate-N content within the rootzone during 2020 and 2021 growing

#### **3.3.3.2** Soil Test Phosphate (STP)

Figure 3.5 shows the STP content within the root zone for each treatment at different growth stages during the 2020 and 2021 growing seasons. In 2020, only the 30 cm depth was analyzed. Results show no consistent STP trends and no statistical differences between the treatments due to the high variability in the sample. However, STP was highly concentrated at the top layer throughout the study period. Phosphate (P) is not as mobile as other nutrients as it is adsorbed to soil particles. This agrees with other studies (Vadas et al. 2005, King et al. 2015, Gramlich et al. 2018, Pease et al. 2018, Liu et al. 2021). Wilson et al. (2016) observed that STP at the top depth (0-15 cm) was significantly higher than 15-60 cm depth under different fertilizer inputs, management, and histories in eight fields in southwestern Manitoba. They reported a mean average of 10.8 and 2.8 mg/kg for 0-15 cm and 15-60 cm.

Meanwhile, too much fertilizer application can cause the downward transport of P in the soil profile and P export in drainage water. As shown in Fig. 3.5, STP decreased with increasing depth. Due to low rainfall and high temperature, low soil moisture might also have limited dissolved P, limited P release, and reduced nutrient uptake, resulting in reduced yield. Pease et al. (2018) reported that weather conditions influence nutrient dynamics.

The STP is an indicator of dissolved P in snowmelt runoff and export from drain tiles (Vadas et al. 2005, King et al. 2015, Gramlich et al. 2018, Grenon et al. 2021). Several studies have reported a strong positive correlation between STP and dissolved P in drain tiles (Pease et al. 2018) and runoff (Wilson et al. 2019, Liu et al. 2021). According to the Manitoba soil fertility guide (2007), the level of STP in the study area is classified as very high STP (>15 ppm), medium STP (8-11 pm) and low STP (<3 ppm) for the top, mid-profile, and deep profile, respectively. The high concentration of STP at the study site poses an increased risk of P enrichment and water pollution as research has shown that sites with STP higher than the recommended values would have high soluble P concentrations in drainage water and runoff (Pease et al. 2018).



Figure 3.5: Soil Test Phosphate within the rootzone during 2020 and 2021 growing seasons

# 3.3.4 Canola quality assessments

#### 3.3.4.1 Oil content, Protein content, and Glucosinolate contents

Canola quality assessments, including oil, protein, and glucosinolate contents, are presented in Figure 3.6. The figure shows the average annual values across the treatments. Statistical results showed no significant statistical differences between ND, CD, and FD. Also, there were no clear observable trends between the treatments during the study periods. However, there was a statistical difference between the years for some of the parameters. This suggests that environmental variables may have masked drainage impacts. This is in line with studies that reported that the environment has the most significant influence on canola quality (Omidi et al. 2010, Hammac et al. 2017).

Oil content is positively correlated to yield (Khalatbari et al. 2021) since it constitutes about 40-45% of the dry weight (Barthet, 2020). Across the treatments, average oil content was 45.9% in 2019, 46.0% in 2020, and 43.3% in 2021. The average protein content was 24.6, 24.2, and 27.8% in 2019, 2020 and 2021. Also, the average glucosinolate was 5.7, 19.6, and 13.9  $\mu$ mol/g in 2019, 2020 and 2021, respectively. Despite the unusual weather observed in 2021, the glucosinolate content did not exceed the threshold set by the Canada Council of Canola. The values in this study were consistently higher than western Canada's 5-year average for oil, protein and glucosinolate content (Barthet, 2020).



Figure 3.6: Average annual oil, protein, and glucosinolate contents per treatment during the study period (2019-2021)

The results showed there was a significant decrease in oil content and increased protein content in 2021 relative to 2019 and 2020. The 2021 growing season was characterized by heat and drought stress. This agrees with other research that reported oil and protein content are inversely related (Rathke et al. 2005, Aksouh-Harradj et al. 2006, Hossain et al. 2019). Heat stress decreases oil content by downgrading the expression of genes associated with photosynthesis and lipid metabolism. In contrast, protein content is increased because of upgrading the expression of genes related to protein biosynthesis (Zhou et al. 2018). Also, the increase in protein content under heat stress could be related to the competition for carbon during the biosynthesis process (Rathke et al. 2005) and increased nitrogen bioavailability, resulting in more amino acid assimilation (Singer et al. 2016). However, the oil content recorded in this study was higher when compared with similar studies (Pavlista et al. 2016, Elferjani and Soolanayakanahally 2018, Khalatbari et al. 2021, Chaganti et al. 2021), despite the weather conditions. This may be due to cultivar differences and other local factors such as soil and nutrients.

#### 3.3.4.2 Fatty Acid (FA) constituents

Figure 3.7 shows the annual average of the FA components per treatment. As expected, of the total fatty acid profile, the oleic acid (C18:1) was highest, comprising 63.8% of the total fatty acid content, followed by Linoleic acid (C18:2), 19.1%, linolenic acid (C18:3), 8.8%, Palmitic (C16:0), 3.7%, Stearic acid (C18:0), 3.7%, and erucic acid (C22:1), 0.1%. Statistical results showed no clear observable trends for all the FA components and no significant differences between the treatments. However, there was a significant decreasing trend for Stearic and Oleic acid over the study period while Linoleic and Erucic acid increased. Also, there was a higher proportion of Linoleic, Erucic acids in 2021; Linolenic and Palmitic acids and 2020; while Stearic and Oleic acids had higher contents in 2019. This could be related to the weather of the growing periods, which was classified as normal, hot-dry, and extremely hot and dry for 2019, 2020, and 2021 respectively. The results of this study agree with numerous studies that state that the FA composition is mainly affected by heat and drought stress. The trend observed in Oleic, Linolenic acid agrees with Zhou et al. (2018), while the trend in Stearic acid agrees with Moghadam et al. (2011) and that of Palmitic agree with Pokharel et al. (2020).



Figure 3.7: Average annual proportions of fatty acids (Palmitic acid (C16:0), Stearic acid (C18:0), erucic acid (C22:1), Linolenic acid (C18:3), Linoleic acid (C18:2), Oleic acid (C18:1) per treatment during the study period (2019-2021)

There are inconsistent trends of canola fatty acid profile in the literature due to environmental stresses. This is because of the strong interaction between cultivar and environment (Hammac et al. 2017, Pokharel et al. 2020). However, a lot of studies report that heat and drought stress increased relative proportions of saturated fatty acids (palmitic and stearic acid), oleic acid (monosaturated fatty acid) and decreased polyunsaturated fatty acids (linoleic and linolenic) (Pritchard et al. 2000, Pavlista et al. 2011, Aksouh et al. 2006, Pokharel et al. 2020). On the contrary, while some studies reported decreased and increased proportions of oleic and linoleic acid, respectively (Elferjani and Soolanayakanahally 2018, Zhou et al. 2018), Pavlista et al. (2016) reported a slight impact on fatty acid composition, with no significant effect on Oleic and linoleic acid.

# **3.4 Conclusion**

The impacts of water management on canola yield and qualities were evaluated under three different treatments: controlled drainage (CD), free drainage (FD), and no drainage (ND) for three growing seasons (2019-2021). Canola yield, oil quality tests (oil, protein, and glucosinolate contents), fatty acid profile tests, nitrate and phosphate contents were also measured for each treatment. The result showed that weather variables likely influenced drainage impacts on yield. In 2019 with relatively reasonable rainfall amount and average normal temperature, results showed that CD plots consistently had the highest yield with a significant statistical difference compared to FD. However, as the growing season rainfall decreased in the following years, the impact of drainage, especially CD, diminished, resulting in no significant differences between the treatments. Soil nutrient analysis showed that ND had higher soil nitrate content across the treatments, even though it was not significantly different compared to CD and FD, respectively.

Canola oil quality results, including oil, protein, glucosinolate, and fatty acid profile, indicated that environmental variables might have masked drainage impacts. There were no significant differences between the treatments but varied significantly over the years. With the predicted rise in temperature and prolonged water deficits in the future in the Canadian Prairies, the results in this study suggest re-examination of the field drainage design and precise water management strategies. A significant amount of water from snowmelt is lost during springtime, which could otherwise be captured, stored, and re-used during sensitive growth periods.

## **Chapter Four**

# DRAINMOD simulation of drain spacing impact on canola yield in heavy clay soils in the Canadian Prairies

#### Abstract

Excess moisture within the root zone due to the shallow water table is a leading cause of crop loss in Manitoba. In this study, the ability of the DRAINMOD model to predict water table depth (WTD) in clayey soil was evaluated using measured field data from the 2019 and 2020 canola growing seasons in Arborg, Manitoba, Canada. Statistical analysis and graphical plots showed close agreement between the measured and simulated WTD with an overall coefficient of determination ( $\mathbb{R}^2$ ), root mean square error (RMSE), mean average error (MAE), and mean bias error (MBE) of 0.93, 9.84 cm, 7.06 cm, and -0.13 cm, respectively. Since the model simulation was deemed satisfactory, the model was run with 30-year historical climate data to assess the impacts of different drain spacings on canola yield. Simulation results showed that the average surface runoff increased while average drainage and relative canola yield decreased as the drain spacing increased. The simulation results suggest that the long-term average yield would be maximized by close drain spacing  $\leq 15$  m. Economic analysis showed that 10 m drain spacing would maximize the return on investment. The need for long-term simulations to develop appropriate site-specific water management strategies is demonstrated.

Keywords: DRAINMOD, model, drain spacing, canola yield, Canadian Prairies

# 4.1 Introduction

The water table (WT) is an important component of the agro-hydrologic cycle, especially in an area with a shallow water table, such as in southern Manitoba. The WT directly or indirectly affects other hydrologic variables such as drainage, runoff, flood, and droughts (Amatya et al. 2019). It affects biochemical processes such as nutrient cycling, organic matter mineralization, and nitrification (Amatya et al. 2019). Under dry conditions, WT can satisfy crop water demands through capillary action. It also affects field operations such as tillage, planting, and harvesting, etc. A high-water table could result in saturated soil conditions. Excess soil water is the leading cause of crop loss in Manitoba (MASC, 2020). Therefore, understanding water table dynamics is key to sustainable water management in the region, especially in the face of increased climate variabilities and extremes.

However, water table dynamics are complex and often difficult to predict because they depend on soil, crop, climate, topography, and management practices. Because of this, hydrologic models have been developed to address the present and future needs of producers while ensuring environmental sustainability (Masasi et al. 2020). To address the challenges of limited field data and uncertainties in measured data, models are first parameterized in other to gain confidence in the model's output. Models can accommodate climatic variability using historical weather data. Models run with long-term weather data have been used to evaluate drainage design (Cordeiro and Sri Ranjan 2015; Sands et al. 2015), operation strategies of water table management systems (Ale et al. 2009), fertilizer rates and fertilizer application (Hashemi et al. 2020; Wilson et al. 2020) in order to suggest better alternatives. With increased climatic variability and uncertainty, there is a need to identify a cost-effective drainage system that maximizes crop yield and reduces nutrient loss.

Drainage design variables, e.g., drain depth and spacing, are the most critical factors that can be used to control water table depth and assess the performance of a drainage system (Kladivko et al. 2004). When installing a drainage system in a new area, most drainage contractors develop their own rules or adopt what is practiced elsewhere (Sands et al. 2015). Modelling and field studies have shown that drainage design variables are location-specific due to differences in soil, weather, and management practices. For example, Singh et al. (2006) recommended drain depth and spacing of 1.05 m and 25 m, respectively, to reduce subsurface drainage and maximize crop yield in Iowa with silty clay loam. In southern Minnesota with the same soil texture, Sands et al. (2015) reported an optimal drain depth and spacing of 1.35 and 46 m, respectively. For environmental benefits, Hashemi et al. (2020) recommended a drain depth and 50 m, respectively, for silty clay soil in Iran.

There are numerous hydrologic models, but only a few include processes that can simulate the hydrology of cold regions. Model simulation in a cold environment is challenging, limiting the use of models in such areas (Dayyani et al. 2010; Morrison et al. 2014). In cold climates, snowmelt and soil freeze-thaw are important processes that affect the hydrologic cycle as it controls infiltration, groundwater recharge, crop water uptake, and available plant water (Baulch et al. 2019).

DRAINMOD is one of the standard hydrologic models used for water management in poorly drained soils (Skaggs 1978, Skaggs et al. 2012). The DRAINMOD model has been applied globally under different climates, water management systems, soil, and crop types (Singh et al.

2006; Wang et al. 2006; Yang 2008; Luo et al. 2010; Sands et al. 2015; Malota and Senzanje 2015; Mohammadighavam and Klove 2016; Gaj and Madramootoo 2017; Yousseff et al. 2018; Askar et al. 2020; Saadat et al. 2020). In Canada, DRAINMOD has mainly been applied in humid and Atlantic regions (Shukla et al. 1994; Singh et al. 1994; Yang et al. 2007; Dayyani et al. 2010; Morrison et al. 2014; Golmohammadi et al. 2016) with limited application in the Canadian prairies with a semiarid climate, even though the region contributes significantly to the Canadian agriculture. The model was first evaluated in southern Manitoba with sandy loam soil under corn and potato production (Cordeiro and Sri Ranjan, 2015; Satchithanantham and Sri Ranjan 2015). It should be noted that even under a similar climate, model performance varies with location (Orth et al. 2015). Most DRAINMOD applications under canola production were carried out in Iran (Hassanpour et al. 2011, Davoodi et al. 2019, Hashemi et al. 2020). There has not been any study of DRAINMOD model application under canola production in the Canadian Prairies. This study fills the knowledge gap. Canola is an important crop with significant economic value, contributing an average of \$29.9 billion per year to the Canadian economy since 2016 (LMC International 2020). Canada is the world's largest producer of canola, exporting about 20 million tonnes annually (Statistics Canada, 2018). The Canadian prairies (Saskatchewan, Alberta, and Manitoba) account for more than 98% of Canadian production. Manitoba produced more than 3.3 million tons in 2018 (Statistics Canada, 2018).

Extensive model testing and evaluation under different climate, soil, crop, and management practices are still needed, especially in cold environments (Yang et al. 2007; Saadat et al. 2020). Therefore, the overall objective of this study is to evaluate the DRAINMOD model. The specific objectives are (i) to calibrate and parameterize the DRAINMOD model, (ii) to simulate water table depth under canola production in the Interlake region of Manitoba, Arborg (iii) to determine the water balance components of the agricultural field using a 30-year historical climate data (1986-2015), (iv) to evaluate the impact of different drain spacings (8, 15, 30, 50, 70 and 90 m) on canola yield, drainage, and runoff, and (v) to determine the drain spacing that would maximize the return on investment. This study emphasizes the need for long-term studies under different water management strategies to maximize yield, increase profitability, and ensure environmental sustainability.

## 4.2 Materials and methods

#### 4.2.1 Study Area

A field drainage experiment was designed to evaluate drainage impacts on crop yield at the PESAI (Prairies East Sustainable Agriculture Initiative) site (Arborg, Manitoba, Canada, to assess drainage impacts on crop yield. The PESAI site (latitude 50.904, longitude -97.273, elevation 229 m) is an agricultural research farm in the rural municipality of Bifrost-Riverton, near Arborg, MB. The region has a semiarid climate, with cold-dry winters and warm summers (Bueckert et al. 2013). The experimental area was divided into four sections (Fig. 4.1).

The first three sections measure 600 m by 60 m and contain the drainage treatment replicated three times. Each of the plots is separated by a 5 m buffer zone and is of equal size with an area of 1.2 ha. Nine out of the twelve plots have drain tiles installed at a design drainage depth of 0.9 m with three different spacing (4.57, 9.14, and 13.72 m). This site serves as a demonstration site for showcasing tile drainage's impact on crop performance. Therefore, three different spacings were used with three other crops in rotation. However, this study used the data from the plots with the 13.72 m spaced drain tiles planted with canola. The remaining three plots are the control treatments (no drain).

The laterals are installed at an average slope of 0.15%, varying from 0.9 m to 1.1 m, in the North-South direction, and connected to 200 mm headers which discharge into the outlets at the south edge of the field. The drain tiles are made up of corrugated pipes, 200 m long and 100 mm diameter. The area has a nearly flat topography with a slope of 0.01%. The soil belongs to the Fyala series, which consists of poorly drained Rego Humic Gleyol soil that develops weakly to moderately calcareous, lacustrine clay deposits (Podolsky, 1982). The soil is predominantly heavy clay with low permeability, deep water table depth and peat content. The soil is prone to shrinkage and swelling due to its high clay content. Thus, soil cracks are common in the field.



Figure 4.1: Layout of the study area (Not to scale)

During the study period, tillage was done after harvest in the fall. Canola (*Brassica napus* cultivar 'Invigor' L233P) was seeded using a 3.7 m wide Salford air seeder at a rate of 7.85 Kg/ha on May 29<sup>th</sup> and 21<sup>st</sup> in 2019 and 2020, respectively. The inter-row and plant spacings are 0.19 m and 0.10 m, respectively. Fertilizer application was based on the residual soil nutrient status. Soil tests for Nitrogen and Phosphorus were done annually in September, preceding the growing season at the Farm Edge laboratory using standard procedures. In 2019, 89.7-123.3 kg/ha of Nitrogen and 44.8 kg/ha of Phosphorus were applied. In 2020, while all the plots received 20 kg/ha of Phosphorus, the tile-drained and undrained plots received 112.1 and 56 kg/ha of Nitrogen, respectively. Weeds were controlled by spraying Liberty at 3.34 L/ha. Similarly, Decis was sprayed on June 25 at a rate of 0.124 L/ha to control insects, while Reglone, a desiccant, was sprayed at 2.06 L/ha on Aug 25. For both years, canola attained full maturity between 16-17<sup>th</sup> August and was harvested on September 9<sup>th</sup> and 2<sup>nd</sup> in 2019 and 2020, respectively. Crop yield was determined after each year's harvest by weighing using the Wintersteiger plot combine. Canola yield was calculated based on the weight of seeds from two approximately 20 m long strips and 1.25 m wide, from each replicate.

## 4.2.2 Field instrumentation and data collection

Water table depth was measured using water level sensors (Solinst Levelogger Junior 3001, Solinst Canada, Ltd., Georgetown, Ontario, Canada) suspended inside the piezometric wells.

The sensor has inbuilt data logging capabilities, set on a 3-hour logging interval. A barometric pressure sensor (Solinst Barologger) was also installed at the outlet of replicate 1 to correct the readings of the levelogger for atmospheric pressure. The wells are made of schedule 40 steel pipe, with an internal diameter of 0.0413 m. A Kevlar rope was used to suspend the sensor, tied to a ring (0.038 mm diameter) hung on a bolt made very close (0.038 m) from the top. The wells have slits on the sides and are covered with socks to allow easy movement of water and prevent soil and clay particles from clogging. The pipes were inserted in the auger holes. Bentonite was used to avoid preferential flow, and wells also had a cap at the top to prevent direct entry of rainwater. The average mid-spacing water table depth for the tiles installed at 13.72 m spacing was used for model simulation in this study. As read from the contractor design, the drain tile closest to the observation well was approximately 1.06 m, despite having a design depth of 0.9 m. In 2019, the levelogger sensor was installed on the 12th of July due to late delivery and installation of the water level sensor, but in 2020, monitoring started on May 25th. The monitoring period in 2019 was shorter than the 2020 growing season. In order to accommodate more data points, the monitoring period was extended to the end of September.

#### **4.2.3 DRAINMOD MODEL**

DRAINMOD is a process-based, field-scale, hydrologic model developed to evaluate the impacts of different water management systems in poorly drained agricultural fields (Skaggs 1978). DRAINMOD performs water balance on a soil profile, midway between two tile drains from the soil surface to an impermeable layer (Skaggs 1978). It also calculates the water balance at the soil surface. The model predicts surface runoff, sub-surface drainage, evaporation, infiltration, crop yield, and water table depth on a daily, monthly, and yearly time scale. A detailed description of the model is reported by Skaggs (1978). Since its inception in the 1970s, more processes have been added to simulate hydrologic processes accurately. Examples include the inclusion of soil nitrogen and salinity (Kandil et al. 1995; Brevé et al. 1997), freeze-thaw processes (Luo et al., 2000), C-N dynamics (Youssef et al. 2005), coupling with crop growth models (Negm et al. 2014), and phosphorus fate and transport (Askar et al. 2020). The DRAINMOD model computes the relative yield of crops using the stress-day method (Evans and Skaggs, 1991), which considers yield reduction due to excess water, drought, planting delay, and salinity stress. In this study, salinity was not considered because the measured soil electrical conductivity (ECe) ranged between 0 - 4 dS/m

(<u>https://agrimaps.gov.mb.ca/agrimaps/)</u>, is less than the threshold (5-10 dS/m) that causes yield reduction in canola (Francois 1994, CCC, 2021).

Therefore, the relative yield is computed as the product of YR<sub>w</sub>, YR<sub>d</sub>, and YR<sub>p</sub>.

$$Y_R = YR_w \times YR_d \times YR_p \tag{4.1}$$

where  $Y_R$  is the overall relative yield,  $YR_w$  is the relative yield due to water stress only,  $YR_d$  relative yield due to drought only,  $YR_p$  and relative yield due to planting delay only. The linear functions showing yield reduction arising from wet stress, drought, and planting delay are given by Evans and Skaggs (1991).

## 4.2.3.1 Model inputs

The inputs required to run the DRAINMOD model are classified into the soil, drainage design, crop, temperature, and trafficability properties. One of the advantages of the DRAINMOD model is that it requires few measured data (Jin and Sands 2003). It has in-built subroutines for determining complex model inputs that are usually time-consuming and difficult to obtain.

### 4.2.3.1.1 Weather data

An on-site weather station, located about 100 m from the study plots, and managed by Manitoba Agriculture, provided the data used in this study. Weather data is used to determine the potential evapotranspiration (ET). The model offers the option for users to input already computed ET or use Thornmwaite's (1948) ET equation. In this study, the latter was used. Hourly precipitation, maximum and minimum temperature, and geographic location are the required inputs to determine ET.

#### 4.2.3.1.2 Soil

The required soil inputs include soil-water characteristics, drainage volume, WTD, upward flux relationship, and infiltration parameters. The soil information for the study area was obtained from Manitoba Agrimap and fed into the ROSETTA software to derive the initial soil hydraulic properties of the study area. As read from the digitized Manitoba Agrimap and Podolsky (1982), average saturation, field capacity, permanent wilting point, clay, silt, and sand fractions are 45.4%, 37.6%, 21.6%, 90%, 9%, and 1 %, respectively with pH of 7.0-7.5. The average bulk density for heavy clay is 1.07 g/cm<sup>3</sup> (Manitoba Agriculture, 2021). The ROSETTA software uses a pedotransfer function to determine the soil water retention and hydraulic

properties from soil texture and bulk density (Schaap et al. 2001). ROSETTA-derived soil hydraulic parameters are as good as experimentally derived soil parameters (Salazor et al. 2008). Based on field excavation, the soil profile was divided into two layers (0-40 cm) and (40-250 cm) to represent the top and deep layers, respectively. Table 4.1 shows the final calibrated soil hydraulic parameters.

Layer	$\theta_r$	$\theta_{s}$	α*	L	n (-) *	K <sub>sat</sub> (cm/day)*	Ko
(cm)	(m <sup>3</sup> /m <sup>3</sup> )	$(m^{3}/m^{3})$	(cm <sup>-1</sup> )				(cm/d)
0-40	0.01094	0.516	0.021	-0.6831	1.13	24	20
40-250	0.01942	0.540	0.019	-0.8319	1.12	23.5	20

Table 4.1: Soil hydraulic parameters of the study area

 $\theta_r$  residual water content,  $\theta_s$  saturated water content,  $\alpha^*$  curve parameter, n curve shape parameter, K<sub>sat</sub> is saturated hydraulic conductivity, K<sub>o</sub> matching point at saturation

\* Calibrated parameter

# 4.2.3.1.3 Drainage design

The drainage system inputs were obtained directly from the field, as shown in Table 4.2. Drain spacing was 13.72 m, drain depth was 1.06 m, the effective radius was 0.51 cm (actual radius was 5 cm), and the depth to the impervious layer was assumed 6 m following Cordeiro and Sri Ranjan (2015) and Satchithanantham and Sri Ranjan (2015). DRAINMOD has an internal subroutine that determines the equivalent depth from the drain to the impermeable layer and Kirkham's coefficient based on drain spacing, effective radius, and distance between the drain and the impervious layer.

Table 4.2: Model inputs in DRAINMOD

Model inputs	Value	
Drainage system		
Drain spacing (m)	13.72	
Drain depth (m)	1.06	
Effective drain radius (cm)	0.51	
Drainage coefficient (cm/day)	1	
Distance from surface to impermeable layer (m)	6	
Maximum surface storage (m)	0.5	
Kirkham's depth for flow to drains	0.25	

Crop parameters

Desired planting date (day)	138		
Last day of the year to plant without yield loss (day)	140		
Length of the growing season (day)	90		
Days required to prepare seedbed and plant (day)	2		
Root zone lower water content	0.1*		
Limiting water table depth (cm)	30		
Period to count wet and dry days	May 1 - Aug	gust 31	
Yield reduction parameters			
Linear relationship relating excess water stress to relative yields	Intercept 100% Slope: -0.3*		
Linear relationship relating drought stress to relative yields	Intercept 10	00% Slope: -	
Crop susceptibility factors (DAP)	$1.22^{*}$		
	0(0-6), 0.3 (7-44), 0.15(45-		
	59), 0.1 (60-89), 90 (0) *		
Trafficability	Period I	Period II	
Start counting work days (day)	April 1	September 15	
End counting work days	June 30	November 15	
Minimum air volume required to work the land (cm)	1.5		
Minimum rain to stop delay work (cm)	1		
Minimum time after rainfall before work can restart (day)	2		
Soil temperature			
Soil thermal conductivity (W/m/°C)	a = 0.5, b = 1	1.5	
Average air temperature which precipitation is snow (°C)	0		
Average air temperature above which snow starts to melt	2		
(°C)			
Snowmelt coefficient (mm/day/°C)	3*		
Critical ice content above which infiltration stops (cm <sup>3</sup> /cm <sup>3</sup> )	0.2		

\* Calibrated

#### 4.2.3.1.4 Crop data

The crop input data include desired planting date, harvest date, effective rooting depth, and stress crop parameters. The optimum planting date was set as May 25, according to Manitoba Agriculture. Recent canola species have a shorter growing period as they are due for harvest between 90-100 days after planting (DAP). Although the maximum rooting depth for the canola crop can reach 1 m (Cutforth et al. 2013), the effective rooting depth is where more than 70% of water uptake occurs (Hashemi et al. 2020). Therefore, the effective rooting depth ranged from 3 to 30 cm, as shown in Table 4.3. The model uses this depth to define the zone from which water can be removed in response to ET under soil limiting conditions. As contained in Table 4.2, the crop susceptibility factors were obtained from Hassanpour et al. (2011) and interpolated between the dates.

Month	Day	Depth (cm)		
1	1	3		
4	25	3		
5	14	30		
5	27	30		
6	1	30		
6	20	30		
7	24	30		
8	20	30		
9	24	3		
10	25	3		
12	31	3		

Table 4.3: Effective root depth of canola used in the simulation

#### 4.2.3.1.5 Trafficability

The trafficability input parameter was set for spring and fall conditions to check for trafficable conditions necessary for planting and harvest. In DRAINMOD, planting is delayed when all the conditions set by the user are not met: the minimum required air volume to work the land, the minimum rain to restart work, and the delay after rain to continue work. The values for these conditions are listed in Table 4.2.

#### 4.2.4 Model performance

The performance of the DRAINMOD model was assessed using both graphical and statistical indices. A visual plot showing a close match between simulated and measured water tables is deemed acceptable. Also, statistical indices commonly used in hydrologic model performance were used. They include R<sup>2</sup>, RMSE, MBE, and MAE.

# **Coefficient of determination** (**R**<sup>2</sup>)

The coefficient of determination  $(\mathbb{R}^2)$  is used to show a correlation between measured and predicted values. It ranges from 0, indicating no correlation, to 1, meaning perfect agreement between observed and predicted. It is calculated as:

$$R^{2} = \left(\frac{\sum_{i=1}^{n} (o_{i} - \overline{o})(P_{i} - \overline{P})}{\sqrt{\sum_{i=1}^{n} (o_{i} - \overline{o})^{2}} \sqrt{\sum_{i=1}^{n} (P_{i} - \overline{P})^{2}}}\right)^{2}$$
(4.2)

## **Root Mean Square Error (RMSE)**

RMSE is used to quantify the mean differences between observed and predicted values. It is given as:

$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (O_i - P_i)^2}{n}}$$
(4.3)

# Mean Bias Error (MBE)

MBE is used to quantify model over or underestimation. Positive and negative values indicate overestimation and underestimation, respectively.

$$MBE = \frac{\sum_{i}^{n} (P_i - O_i)}{n} \tag{4.4}$$

## Mean Absolute Error (MAE)

MAE is a measure of error variability. The average error is better measured using the MAE than the RMSE (Willmot and Matsuura 2005).

$$MAE = \frac{\sum_{i}^{n} |(P_{i} - O_{i})|}{n}$$
(4.5)

where,  $O_i$  is measured values,  $P_i$  is the predicted values,  $\overline{O}_i$  is the mean of observed values, and n is the total number of observed data points. An optimal model is obtained when  $\mathbb{R}^2$  is equal to 1, and RMSE, MAE and MBE are equal to 0 (Mkhabela and Bullock 2012, Hashemi et al. 2020).

#### 4.2.5 Assessment of drain spacing on predicted relative canola yield

After model calibration, the model can be extrapolated over various scenarios, management practices and weather conditions (Boumann et al. 1996). In this study, historical weather data (1986-2015), including maximum and minimum temperature, and precipitation were obtained from the archives of Environment Canada for the Arborg weather station and used to run DRAINMOD. Daily precipitation was uniformly distributed over 4 hours, using the inbuilt weather utility in DRAINMOD. Missing temperature and rainfall data were gap-filled using the closest station (Gimli).

According to Cordeiro and Sri Ranjan (2015), drainage contractors in Manitoba have generally adopted a tile drain installation depth of 1.06 m and a spacing of 15 m. They attributed this choice to reduced energy requirements during installation. In this study, drain depth was kept constant while the drain spacing was changed during the simulation. Six drain spacings were selected (8, 15, 30, 50, 70, and 90 m). The DRAINMOD model was first run with long-term climate data to determine the water balance components. This was followed by running the model on different drain spacing to evaluate the impacts on water balance variables and relative canola yield. A 2-year warm-up period (1984 and 1985) was used to initiate and stabilize the model.

# 4.2.5.1 Economic analysis

The rate of return on investment under different spacing was determined by subtracting the average production cost (\$/ha) and drainage cost (\$/ha) from the production income (\$/ha). The average annual canola production income was calculated as the product of the relative yield, potential canola yield, and unit canola price. In this study, the potential canola yield, unit canola price, and annual canola production cost (excluding land cost) were assumed as 2796.4 kg/ha (45 bu/ac), \$0.45/kg (\$11.25/bu), and \$862.64/ha (\$349/ac) according to MARD (2021). The equation for annual drainage cost was obtained from Ghane et al. (2021). The initial cost of the drainage system was \$3.76/m, assuming that it cost \$2469.1/ha (\$1000/ac) for 15.2 m (50 ft) drain spacing in Manitoba. The interest rate on investment per year, depreciation per year based on a 50-year expected lifetime, and maintenance rate per year were assumed as 6%, 2%, and 0.25%, respectively, according to Crabbé et al. (2012).

#### 4.3 Results and Discussions

#### **4.3.1** Weather variables

During the 2019 and 2020 growing seasons, the total precipitation received was 326 and 237 mm, respectively, 26.6 and 42.6% lower than the 30-year average of 499 mm. Monthly rainfall analysis showed that the site exhibited wide variation from the long-term average values (Figure 4.2). For example, the long-term average was significantly higher than the 2019 and 2020 values in May and August. The 2020 rainfall was higher than the 2019 and long-term rainfall in June, while in September 2019, 2019 was more than twice and four times the 30-year average and 2019 values, respectively. Compared to the long-term average, both years are considered dry, with 2019 being the drier since the September rain had occurred around the harvest period.



Figure 4.2: Comparison of monthly rainfall and average temperature during the study period and 30-year average

The average temperature during the study period was very close to the long-term average values. The highest average temperature was recorded in July, corresponding to the period of highest crop water use. The average temperatures for 2019, 2020, and long-term are 0.86, 2.29, and 1.56 °C, respectively.

# 4.3.2 Model parameterization

The model was calibrated by manually changing parameters affecting the hydrology of agricultural fields following Singh et al. (2006). The initial values were obtained from the literature (Cordeiro and Sri Ranjan 2015 and Satchithanantham and Sri Ranjan 2015) and

adjusted through trial and error until predicted matched measured values. The calibration parameters include soil hydraulic parameters, ET correction factors, and snow coefficient (Table 4.1 and 4.2). Given the high variability of soil properties, Ksat was used as a calibration parameter. After several trial-and-error attempts, the final calibration values were 0.9 and 0.1 cm/hr for layers 1 and 2, respectively, which are close to the reported values in the region (Podolsky, 1982). Another soil hydraulic parameter is the alpha, which modifies the relationship between water table depth vs volume drained, and water table depth vs upward flux was also calibrated. The calibrated monthly ET correction factors are 1.0, 1.2, 1.2, 1.2, 1.2, 0.5, 0.9, 1.0, 1.1, 1.2, 1.2, 1.2, and 1.2 for January through December. To account for freeze-thaw and snowmelt, the soil temperature input data is required. The initial temperature parameters were obtained from the literature with similar weather characteristics (Cordeiro and Sri Ranjan 2015 and Satchithanantham and Sri Ranjan 2015) and later adjusted during calibration. The calibrated snowmelt coefficients and critical ice content above which infiltration stops were set to 3 mm/day/°C and 0.2 cm<sup>3</sup>/cm<sup>3</sup>

# 4.3.3 Water table dynamics

Figure 4.3 shows the measured and simulated water table depth (WTD) in 2019 and 2020. Water table fluctuation was mainly in response to rainfall, ET, and drainage. The graphical comparison between measured and simulated WTD shows that the model followed the same trend as the measured, even though they differed in magnitude. The dry weather of the 2019 growing season caused the water table to be around 2m at the start of the monitoring period, as the model overpredicted WTD by 15%. We observed that the 37 mm of rainfall on July 12th only caused a slight rise in simulated and observed water tables due to the extremely dry soil profile. As presented in Figure 4.2, the total rainfall in May, June, July, and August were 56.6, 60.0, 4.4, and 6.2%, respectively, lower than the 30-year long-term values. Extreme dry and drought conditions were also reported in North Dakota, which is also part of the great plains of North America (Acharya et al. 2019). However, as the simulation progressed, we observed a closer match between simulated and measured and WTD rise in response to rainfall in September. The average simulated and measured WTD are 196.17 cm and 189.68 cm, respectively. The statistical result showed acceptable model performance with R<sup>2</sup>, RMSE, MBE, and MAE of 0.553, 8.745 cm, -6.625 cm, and 7.301 cm, respectively. The negative MBE is an indication that the model underpredicted WTD.

Similarly, in 2020, the plot also shows that DRAINMOD could capture the WTD changes during the monitoring period. With minor observable differences, the model followed the rise and fall of the measured water table. For example, the 32 mm rainfall on June 25th caused the water table to rise to 63.5 cm, although the model lagged by one day and under predicted by 15% (53.6 cm). The trend of the WTD was in close agreement till the end of the growing season in September. At the start of the simulation, observed and simulated followed the same trend even though DRAINMOD was slightly higher than the measured WTD. Towards the end of the growing season, there was little, or no response of both simulated and observed WTD to the late rainfall received in August. This lack of response may be due to the prolonged dryness between the rainfall events. Given the low hydraulic conductivity of clay soil, the rain may have only wet the soil profile or used by the crop through ET. The graphical plot (Fig. 3 & 4) showed that the model satisfactorily simulated WTD throughout the growing season. The average simulated and measured WTD are 135.67 and 139.36 cm, respectively. As indicated in Table 4, statistical results showed satisfactory model performance with R<sup>2</sup>, RMSE, MBE, and MAE of 0.896, 9.712 cm, 3.698 cm, and 6.259 cm, respectively. The positive MBE is an indication that the model overpredicted WTD

The model performance in 2020 was better than the 2019 growing season. This may be due to the larger datasets used in 2020. The rise and fall of the water table in heavy textured soil may be affected by the presence of an impermeable layer that causes water to flow horizontally from mid-way in the drains (Wahba and Christen 2006). Also, the ET estimation method may affect the difference between simulated and observed WTD since ET is a significant pathway for water loss during the growing season. Overall, the model performance during the monitoring periods was excellent.



Figure 4.3: Comparison of measured and predicted WTD in response to rainfall during the 2019 and 2020 growing season

		RMSE	MBE	MAE
	$\mathbb{R}^2$	(cm)	(cm)	(cm)
2019	0.553	8.772	-6.625	7.301
2020	0.896	9.712	3.698	-6.259
Overall	0.927	9.838	0.135	7.056

Table 4.4: Summary of statistics indices on DRAINMOD performance for simulating WTD



Figure 4.4: Regression of measured and simulated WTD over the study period

# 4.3.4 DRAINMOD Hydrologic long-term assessment

The main outputs of the DRAINMOD model include ET, sub-surface drainage, surface runoff, water loss (sub-surface drainage + surface runoff), and water table depth. The average total ET was 297 mm on the annual scale, with the highest ET of 355.2 mm in 2015 and the lowest value of 220.6 mm in 1992. Similarly, the average water loss was 222.7 mm, with a maximum value of 399.6 mm in 2008 and a minimum value of 51.9 mm in 2003. Average water table depth was shallowest in 1993 (64.53 cm), deepest in 1989 (213 cm), with an average value of 102.1 cm.

Water balance analysis (Table 4.5) shows that ET, water loss (drainage + runoff) accounted for 57.2 and 42.9% of the total precipitation. About 94.1% of ET occurs during the growing season (May - September), and 56.62% is water loss, where drainage and runoff account for 34.1 and 22.4%, respectively. Similarly, in the non-growing season (October - April), drainage and runoff represent 19.97 and 23.4%, respectively, of the water loss. Runoff peaked in April, which corresponds to the period of high snowmelt in the region. However, runoff constitutes about 48.9% in the growing season. Similarly, peak subsurface drainage occurred in May, and drainage represents approximately 63.06% in the growing season.
Hydrologic	Annual	Growing	Non-growing
Variables	Average	Season (GS)	season NGS)
Rainfall (mm)	519.4	356.1	163.3
Infiltration (mm)	417.5	307.2	110.2
Evapotranspiration (mm)	297.0	279.6	17.4
Runoff (mm)	101.9	49.9	52.1
Drainage (mm)	120.5	75.9	44.5
Water loss (mm)	222.8	126.1	96.6

Table 4.5: Summary of annual average water balance

Similar results were also observed under different drain spacing. Also, under increasing drain spacing, across the months, drainage rates decreased while surface runoff increased. We also observed that drainage and runoff were highest at the beginning of the growing season (April and May) and gradually decreased into the winter. Results also showed no runoff from December through February and minimal drain discharges from December to March. This corresponds to the winter period with frozen ground. This study agrees with similar long-term simulation studies with similar climates (Jin and Sands 2003, Satchithanantham and Sri Ranjan 2015). Also, the results agree and disagree to some extent with the field study conducted by Kokulan et al. (2019) in the same province. They reported that tile drainage constituted about 10-25% of the annual runoff. The difference may be due to the prevailing weather of the study periods.

The long-term water balance analysis allows us to manage water by altering the variables controlling hydrology. ET is the largest source of water loss during the growing season, followed by drainage and runoff from the above analysis. Therefore, water management strategies such as controlled drainage would conserve water and potentially reduce nutrient losses from drainage tiles.

Figure 4.6 presents the simulated average monthly water balance components for the different spacing. As drain spacing increases, the predicted average annual runoff increases with a marginal increase in ET, while infiltration and subsurface decrease. This has a practical implication on the predicted relative yield because as drain spacing increases, less water is removed from the profile, trafficable conditions are not met, planting is delayed, which would result in reduced yield.



Figure 4.5: Monthly average drainage and runoff under different drain spacing



Figure 4.6: Impact of drain spacing on the average annual water balance components

## 4.3.5 Impact of drain spacing on canola yield

Figure 4.8 presents the average predicted relative canola yield under different drain spacings. Only years with yield results across all six spacings are given. The years 1993, 2000, 2005, 2007, and 2008 were excluded because zero yields were obtained at drain spacings > 50 m. Out of the 25 years, 16 experienced yield reduction due to excess water, six due to combined effects of excess water and planting delay, while the remaining were due to all the stressors. Wet stress caused the highest yield reduction followed by planting delay and drought stress, as shown in Fig. 8.



Figure 4.7: Predicted relative yield from 1986 to 2015 under different drain spacing

Simulation results showed varying weather-yield results, indicating underlying factors influencing yield. Simulation results suggest that years with precipitation greater than the long-

term average had relative yield reduction due to excess water. Examples include 2010 (749 mm; 48.2% above average), 2012 (656 mm; 29.8% above average), 1996 (558.4 mm, 10.5% above average) etc. However, it should be noted that high precipitation amount does not directly translate to yield reduction due to wet stress. The timing of the rainfall is a significant determining factor. We observed that the highest reduction was associated with years with high rainfall amounts in May and June. This corresponds to the period with the highest susceptibility factor (0.28), as canola yield is most susceptible during the vegetative stage (Hassanpour et al., 2011). Wet stress was also observed in years with precipitation less than the average long-term average. These years have the highest predicted relative yield because they experienced no wet stress at the early developmental stage. Examples include 1987 (348.8 mm; 31.6% below average), 1988 (368 mm; 27% below average), 2002 (19.2% below average) etc. The simulation results also showed that the predicted relative yield was affected by factors such as the precipitation amount received during the late part of the previous year and rainfall duration. The antecedent water table depth and snowmelt and early spring rain in the incoming year would likely affect yield through wet stress and planting delay stress. We observed that years with total precipitation amounts > 50 mm in April had yield reduction. Therefore, with low drainage amounts for widely spaced drains, yield reduction is likely to happen mainly because of excess water, a shift in planting date, and non-workable conditions.





Overall, as drain spacing increased, SEW-30 (sum of excess water within the 30 cm rootzone) and planting delay increased while the predicted relative yield and workdays decreased (Table

4.6). The average predicted relative canola yields were 95.5, 88.4, 75.4, 64.2, 60.6, and 57.4% for 8, 15, 30, 50, 70, and 90 m drain spacing.

Drain	SEW-30		Planting	Predicted relative
spacing (m)	cm-days	Workdays	delay (years)*	Yield
8	60.2	101.5	1 (1)	95.5
15	174.3	89.3	2 (11)	88.4
30	356.9	73.1	4 (15)	75.4
50	450.3	64.6	9 (23.9)	64.2
70	503.6	61.3	9 (25.6)	60.6
90	529.8	60.7	10 (26.3)	57.4

Table 4.6: Relationship between predicted relative yield and objective functions under different spacing

\* Planting delay in years is the number of years out of the 25 simulated years with planting delay, while the value in bracket represents average planting delay (days)

Based on figure 4.9, to achieve at least 80% of its potential yield regardless of the precipitation amount, drains should be closely spaced, at most 15 m apart for the heavy clay soil in Arborg, Manitoba. The results of this study agree with some field and simulation studies (Acharya et al. 2019, Hashemi et al. 2020, Singh and Nelson 2021). Scherer et al. (2015) recommended close spacing for finer soils such as clay. In a field study in North Dakota, Acharya et al. (2019) reported that narrow spacing resulted in the highest crop yield and residual soil nitrate in silty clay loam soil. Hashemi et al. (2020) reported that drain spacing greater than 40 m with heavy clay soil resulted in reduced drainage, increased surface runoff, and potential for runoff. However, close drain spacing has economic and environmental concerns (Cordeiro and Sri Ranjan 2015, Asker et al. 2020).

Close drain spacings would result in increased costs arising from installation, labour, and materials. To this end, an economic analysis was conducted. From Figure 4.9, the 10 m drain spacing is the optimal spacing that would maximize the return on investment



Figure 4.9: Effect of drain spacing on relative canola yield and average annual return on investment.

## 4.4 Conclusion

Subsurface drainage is essential for agricultural production, especially in poorly drained soils. In Manitoba, excess water from early spring and snowmelt is the leading cause of crop loss. Thus, producers in the area have embraced subsurface drainage, leading to a rapid increase of agricultural fields with drainage. However, drainage design parameters are location-specific because of the unique soil, climate, and management practices. Most drainage contractors install these drainage systems by adopting what is practiced somewhere else without considering these unique characteristics. To assess the performance of these systems using field studies is insufficient to evaluate the drainage systems under varying climate and management conditions. This has necessitated the use of hydrologic models together with field data to run the hydrologic model. In this study, the DRAINMOD model was first parameterized by simulating water table depth over two growing seasons under canola production. Both statistical and graphical results showed that the predicted water table depth was in close agreement with measured values. Thereafter, a 30-year simulation was conducted to determine the hydrology and evaluate the impacts of drainage spacing on canola yield. Water balance results showed that ET, Runoff, and Drainage peaked in July, April, and May, respectively. This provides a basis for water management for the region. The simulation results showed that drain spacing affects the predicted relative canola yield. Also, results show that maximum yields were achieved under closed spacing. Even though close spacings could increase cost, the economic analysis showed that the 10 m drain spacing would maximize the return on investment. This study underscores the need for site-specific drainage design variables to maximize yield, increase profitability and ensure environmental sustainability.

## **Chapter Five**

# Assessment of the potential impacts of climate change on the hydrology and canola yield using the DRAINMOD model

## Abstract

Climate change is a major concern for agricultural production regions like the Canadian Prairies. The Canadian Prairies accounts for most of Canada's grain and oil crop. Therefore, understanding the hydrologic and crop yield response to climate change is important for adaptation and mitigation. Downscaled climate model projections from CANESM2 for historical (1981-2010), midcentury (2041-2070), and late-century (2071-2100) periods under three representative concentration pathways (RCP2.6, RCP4.5, and RCP 8.5) were used as climate inputs to drive a parameterized DRAINMOD model. The model was parameterized using measured water table depth from two canola growing seasons (2019 and 2020) in Arborg, Manitoba, Canada. The projected changes in the climatic variables showed that even though the mean annual precipitation is projected to increase, rainfall is expected to decrease, and temperature increase during the growing season. DRAINMOD simulation results suggest that during the midcentury, subsurface drainage outflow is expected to decrease by 11, 17, and 15% for RCP2.6, RCP4.5, and RCP8.5, respectively. During the late century, similar flow reductions were observed under the same climate scenarios. On the other hand, ET and surface runoff are expected to increase under all climate scenarios except for runoff during the late century under RCP 8.5. Furthermore, results showed that the relative canola yield would decrease due to drought stress caused by temperature rise and reduced rainfall during the growing season. Simulation results showed that yield reduction due to excess water would be lower in the future while it would be higher due to drought stress. Even though controlled drainage can decrease the drainage outflow from subsurface drainage in the future, comparable canola yield did not improve due to drought stress during the growing season. The presence of heavy clay with low hydraulic conductivity could limit water movement, thereby minimizing the impact of CD on crop yield. Tile drainage could be used to drain the water, to facilitate field operations before planting, and this water could be stored for subsequent use to overcome the anticipated drought stress during the growing season. The results presented in this study represent a wide range of options for water managers and producers to develop strategies that would mitigate projected heat and drought stress in the future.

Keywords: DRAINMOD modelling, canola yield, climate change, subsurface drainage

## **5.1 Introduction**

Climate change is arguably the most significant environmental challenge in the twenty-first century, as it affects almost all the facets of life, including health, agriculture, infrastructure, social, economic etc. (Rosenzweig and Parry 1994, Lal 2005, Kang et al. 2009, Romero-Lankao et al. 2014, IPCC 2013). Observations worldwide provide undeniable evidence that the global climate has changed and will continue to change (Romero-Lankao et al. 2014, Qian et al. 2016, Qiu et al. 2020). In recent times, extreme hydrologic events (droughts, flooding, wildfires etc.) have increased, and new temperature records are set for many locations, including Canada. Changes in climatic variables are a significant threat to food security, livestock management and production and water resources (Rosenzweig and Parry 1994, Lal 2005). Greenhouse gases (GHG) are the leading cause of climate change (IPCC 2013). Studies have shown that increasing GHG concentration will further lead to increased warming and long-term changes in climatic variables (IPCC 2013, Almazroul et al. 2021). Temperature and precipitation are the major indices used in climate change studies (Almazroul et al. 2021). Changes in these climatic variables can alter the water cycle (Hanke 2018, Jalota et al. 2018, Modi et al. 2021).

Water resources and climate are closely interlinked. Therefore, projected changes in climate variables such as temperature, precipitation, and CO<sub>2</sub> concentration have far-reaching consequences on crop growth, productivity, and the hydrologic variables. Even though climate change is global, the impacts and severity vary by location (Abd-Elaty et al., 2019). The location of Canada in the northern hemisphere has predisposed it to more serious negative impacts even though there are also potential benefits (Qian et al. 2019). The Canadian Prairies are vulnerable to climate change events such as droughts, prolonged dryness, and intense soil moisture deficits (Bonsal et al. 2019). This may be because crop production, especially canola, depends solely on rainfall (Qian et al. 2019). Canada is the world's largest producer and exporter of canola (CCC, 2021). The Canadian prairies account for more than 98% of the total production (CCC, 2021). A recent economic report shows that canola contributes \$29.9 billion annually to the economy, 207,000 jobs and \$12 billion in wages (LMC International 2020). Excess water from snowmelt or prolonged dryness hinders optimum crop production in the Canadian Prairies. Understanding the projected changes in climatic variables, mainly temperature and precipitation, is vital for developing effective water management strategies and sustained crop production.

Field experimentation and hydrologic modelling are complementary tools that provide guidelines and effective water management strategies to cope with predicted climate change impacts (Qian et al. 2019, Kurki-Fox et al. 2019, Qiu et al. 2020, Sojka et al. 2020, Adhikari et al. 2020). Future climate data are usually obtained from global circulation models (GCMs), advanced tools for reliable future climate data. However, GCMs cannot be directly run with hydrologic models because of their coarse resolution (>250km). Scientists have developed different downscaling or bias correction methods to overcome this challenge (Wilby and Wigley 1997), after which they are used as inputs in a hydrologic model. Commonly used models for climate change assessment include DRAINMOD (Awad et al. 2021), SWAT (Wagena and Easton 2018), RZWQM2 (Jiang et al. 2020), HYDRUS (Haghnazari et al. 2020), etc.

There are numerous studies on future climate change impacts on water resources (Barnett et al. 2004, Schilling et al. 2020). In Canada, a lot of these impact studies have focused on crop yield, biomass, greenhouse gases emissions, and permafrost degradation (Zhang et al. 2008, Wang et al. 2012, Smith et al. 2013, He et al. 2018, Qian et al. 2018, Qian et al. 2019, Jiang et al. 2020). The predicted impacts vary from one location to another due to the uncertainties associated with the GCMs, models, and other local factors such as soil types, climate, crop, and management (Qian et al. 2016). For example, Qian et al. (2019) reported that canola yield would increase in the future using three different crop models even as temperature increases. On the other hand, Qian et al. (2018) reported a 21-42% decrease in yield in the Canadian Prairies using the CSM-CROPGRO-Canola model. This justifies the need to assess climate change impacts using robust agricultural models. Smith et al. (2013) applied the DNDC model to simulate wheat yield and nitrous oxide emissions in western Canada. They reported that climate change would increase crop yield and GHG emissions. In Iowa and Quebec, Wang et al. (2015) and Jiang et al. (2020) simulation results using the RZWQM2 model indicated that climate change would increase drainage outflow, nitrate loss and soybean yield, while corn yield would decrease. Using the integrated farm system model, Cordeiro et al. (2019) reported increased forage and biomass production due to climate change in Newfoundland. Using the SWAT model under projected climate change scenarios, Wagena and Easton (2018) reported that tile drainage would be environmentally beneficial by decreasing surface runoff, sediment export and dissolved and total phosphorus. Climate change studies do not precisely predict climate change impacts. Instead, they provide possible potential impacts from which decisions, management, adaptation, and mitigation can be developed (Gunn et al. 2018, Qiu et al. 2020, Pease et al. 2017, Wagena and Easton 2018, Sojka et al. 2020, Adhikari et al. 2020).

The DRAINMOD model is a comprehensive water management system with successful applications in simulating water table depth, drainage flow, nitrates concentration, runoff, and crop yield (Cordeiro and Sri Ranjan 2015, Satchithanantham and Sri Ranjan 2015, Ndulue and Sri Ranjan 2022). Most climate change studies using the DRAINMOD model were conducted in the U.S. (Singh et al. 2009, Pease et al. 2017, Kurki-Fox et al. 2019, Vepraskas et al. 2020, Adhikari et al. 2020), humid regions in Canada (Dayyani et al. 2012, Golmohammadi et al. 2016, Golmohammadi 2021), China (Awad et al. 2021), Poland (Sojka et al. 2020). Although some studies have examined climate change impacts on crop yield and nitrous emission in the Canadian Prairies (Wang et al. 2012, Smith et al. 2013, Qian et al. 2018, Qian et al. 2019), there are limited studies focused on climate change impacts on water management using DRAINMOD.

Understanding projected climate change impacts in the Canadian Prairies and assessing different water management strategies cannot be overemphasized. Therefore, the overall objective of this study is to use a parameterized DRAINMOD model to determine how climate change may affect the hydrology of an agricultural field and how that impacts canola yield. The specific objectives are to (i) characterize future climate in an agricultural field in Arborg, Manitoba, in the twenty-first century under different climate scenarios (ii). determine the potential impact of climate change on the hydrologic variables (subsurface drainage, surface runoff, evapotranspiration, water table depth), and crop yield (iii) suggest management practices to adapt to and mitigate climate change.

## 5.2 Methodology

## 5.2.1 Site description

Field data was collected from the PESAI (Prairies East Sustainable Agriculture Initiative) experimental research site in Arborg, Manitoba (Latitude 50.904, Longitude -97.273, Elevation 229 m). The site is in the Interlake region of Manitoba, which is bounded on the south, east and west by Winnipeg, Lake Winnipeg, and Lake Manitoba, respectively. The region is in the Icelandic River drainage basin with a boreal plain ecozone (Kean 1998). The climate is a dry subhumid, cool continental climate. The non-growing season usually starts around October and lasts until March, with temperatures remaining below freezing point. Snowmelt and soil

thawing begins in late March and early April when soil temperature rises (> 0 °C). The longterm mean temperature ranges from -18.3 °C in January to 18.6 °C in July (Environment Canada 2021). The long-term annual precipitation is 499.4 mm, with rainfall and snowfall accounting for about 81 and 19%, respectively (Environment Canada 2021). The long-term data show that the average monthly rainfall begins to rise in April, peaks in June, and gradually decreases in September when snowfall begins. The average snowfall is highest in the winter, reaching its highest value between November and January. During the 2019 and 2020 study periods, the observed rainfall during the growing season was 26.6 and 42.6% less than the long-term average. Still, the observed temperature was like the long-term average (Ndulue and Sri Ranjan 2022). Depending on the crop, planting usually starts in May, and the growing season extends to September. Since the dominant soil type is heavy clay, topography, and climatic condition, waterlogging and drainage are major water management issues hindering trafficability and crop production. The predominant soil type is heavy clay with low permeability, high water table depth and peat content. The soil in the area belongs to the Fyala series, which consists of poorly drained Rego Humic Gleyol soil that developed on weakly to moderately calcareous, lacustrine clay deposits (Podolsky, 1982).

## **5.2.2 Experimental layout**

The complete detail of the experimental layout is reported in (Ndulue and Sri Ranjan 2022). A summary is presented herein. Three drainage treatments replicated thrice were applied on the experimental site: three drained plots with different drain spacings of 4.57 m (15 ft), 9.14 m (30 ft), 13.72 m (45 ft), and control plots with no drainage. All the drained plots have drain tiles installed between 0.9 and 1.1m below the soil surface, at a slope of 0.01%, in the North-South direction of the field. The tiles are connected to a collector drain that discharges to a drainage control structure at the edge of the field. Piezometers were installed on the centre of each plot to measure water table depth using water level sensors every 3 hours (Solinst Levelogger Junior 3001, Solinst, Canada, Ltd., Georgetown, Ontario, Canada).

The site is a 5-ha field comprising wheat, canola, and soybean in rotation, with land preparation and tillage usually done in the fall after harvest. Throughout the study period, planting and harvesting followed the same sequence: wheat first, followed by canola, and then soybean. Fertilizers, including N and P, were added based on the initial soil test. Herbicides, fungicides and desiccants were applied following standard management procedures for growing the crops in the area.

## 5.2.3 Climate projections

Daily climate model projections were downloaded from the pacific climate impact consortium (PCIC) website (https://data.pacificclimate.org/portal/downscaled\_gcms/map/). The PCIC database provides statistically downscaled climate data across Canada from 1950 to 2100 via an interactive web interface. It uses the BCCAQ method for downscaling coarse GCMs from low resolution to fine-scale resolution. The BCCAQ is a hybrid method that combines two downscaling methods: BCCA and QDM. The BCCA uses a spatial aggregation from a linear combination of historical analogues for daily scale fields. In contrast, QDM uses quantile mapping to preserve relative changes in GCMs quantiles to avoid inflationary effects. The robustness and performance of BCCAQ have been verified by (Werner and Cannon 2015). The BCCAQ method performed better than other downscaling methods; thus, it was recommended for hydrologic modelling impact studies (Werner and Cannon 2015). In this study, the updated method, BCCAQV2, was used to obtain climate data for the baseline (1981-2010), midcentury (2041-2070) and late century (2071-2100) under three emission scenarios called Representative Concentration Pathways (RCP 2.6, RCP 4.5, and RCP 8.5). The RCPs represent the net radiative forcing  $(W/m^2)$  expected by the end of the twenty-first century. Therefore RCP 2.6, RCP 4.5, and RCP 8.5 represent net radiative forcing of 2.6 W/m<sup>2</sup>, 4.5 W/m<sup>2</sup>, and 8.6  $W/m^2$ , respectively, are low, medium, and high baseline scenarios (Van Vuuren et al. 2011).

The downscaled GCM used in this study, CANESM2, is one of the GCMs hosted by the Coupled Model Intercomparison Project Phase (CMIP5) (Taylor et al. 2012). It has a spatial resolution 1/12° and has been used for many climate change studies in North America (Cordeiro et al. 2019). It was selected based on the ordering of Giorgi and Francisco (2000) as recommended by (<u>https://pacificclimate.org/data/statistically-downscaled-climate-scenarios</u>). CANESM2 outputs, maximum temperature, minimum temperature, and precipitation were used as climate inputs to run the DRAINMOD model. Daily precipitation data was disaggregated on hourly time steps using the inbuilt disaggregation tool in the model.

## **5.2.4 DRAINMOD MODEL**

DRAINMOD is a versatile, process-based, field-scale, hydrologic model developed to evaluate the impacts of different water management systems in poorly drained agricultural fields (Skaggs 1978). The model simulates hydrologic variables, including surface runoff, subsurface drainage, evaporation, infiltration, water table depth, and crop yield on a daily, monthly, and yearly time scale. The model has undergone tremendous improvements over the years to include many processes such as soil nitrogen and salinity dynamics, freeze-thaw, Phosphorus fate and transport etc.

## 5.2.4.1 Model setup, inputs, and parameterization

DRAINMOD was parameterized using a 2-year water table depth measured at the study site. Soil-derived properties, including soil water characteristics curve, WTD vs upward flux etc., were derived from the soil properties using the ROSETTA model. Crop properties, including rooting depth over the growing season, were obtained from the literature. Drainage design parameters, including drain depth and spacing, were based on site design. Calibration was done by manually changing soil hydraulic parameters, E.T. correction factors, and snow coefficient. Final calibrated values are reported in (Ndulue and Sri Ranjan 2022). The model performed satisfactorily based on the graphical and statistical analysis of observed and simulated water table depth. The simulated daily water table depths closely matched the observed with an R<sup>2</sup> ranging from 0.55 to 0.93 (Table 5.1). Also, error indices, including MAE, and MBE, were within the recommended acceptable range (Skaggs et al. 2012).

		RMSE	MBE	MAE
	$\mathbb{R}^2$	(cm)	(cm)	(cm)
2019	0.553	8.772	-6.625	7.301
2020	0.896	9.712	3.698	-6.259
Overall	0.927	9.838	0.135	7.056

Table 5.1: Summary of statistical indices used to assess the DRAINMOD model (Adapted from Ndulue and Sri Ranjan 2022)

The above table showed that the DRAINMOD model performed well in simulating the ground water hydrology of the study area. Consequently, the parameterized model was used to evaluate how climate change could affect field hydrology and canola yield under future climate scenarios. It is assumed that the parameterized model outputs are representative of the study area given the excellent model performance.

## 5.2.5 DRAINMOD simulations for climate change assessment

Each climate variable from the baseline and future period were compared on annual, monthly and seasonal periods to determine the relative changes in the variable. The parametrized DRAINMOD model parameters were kept unchanged to assess climate change impacts. At the same time, it was driven by baseline and future climate scenarios under two water management scenarios: free drainage and controlled drainage. Eighteen scenarios were generated (3 RCPs, three-time periods, and two water management strategies). In DRAINMOD, the difference between free drainage and controlled drainage is the consideration of the weir settings. The controlled drainage is simulated by considering the outlet conditions (gate elevation, storage capacity in the outlet ditch, drainage flux etc.). This study sets the weir to 0.6 m during the growing season (May 15-August 30). The weir height is set to the drain depth during planting and harvest to allow trafficable conditions. After harvest and nongrowing periods, the weir is raised to 0.3 m below the ground surface to reduce drainage outflow. The result from each scenario was compared with the baseline and expressed as a percent increase or decrease.

Also, the t-test was used to analyze the mean ET, runoff, and subsurface drainage between free drainage and controlled drainage. ANOVA test with Tukey procedure at a 5% (p<0.05) significance level was used to assess the relative yield difference between the baseline, midcentury, and end of the century periods. All statistical analyses were done in JMP software (Version 16, SAS Institute, Inc, Cary, N.C).

### 5.3 Results and Discussions

## 5.3.1 Projected changes in climatic variables

Table 5.2 summarizes the projected changes in maximum temperature, minimum temperature, and precipitation in the 21<sup>st</sup> century (2041-2100) relative to the baseline climate (1981-2010). Figure 5.1 shows the long-term seasonal climatic changes for baseline and predicted periods. The mean annual results show a general increase in precipitation and temperature in the region under all future scenarios. Temperature is projected to be warmer in the future under all climate scenarios. During the midcentury, the average annual temperature is predicted to rise by 2.2, 3.2 and 6.9 °C under RCP 2.6, RCP 4.5, and RCP 8.5 scenarios, respectively, relative to the baseline. By the end of the century, the average mean temperature is projected to increase by 2.5, 4.2, and 6.9 °C for the same scenarios. Monthly analysis showed a significant increase in average temperature, especially during spring (March and April) and Winter (November and December). Also, the temperature rise intensified during the late century and high scenario (RCP 8.5). Generally, for each RCPs, the projected increase in temperature intensified toward the end of the 21<sup>st</sup> century.

Periods	T <sub>max</sub>	$T_{min}$	$T_{avg}$	PCP (%)
R2.6-2041-2070	2.20	2.23	2.21	9.98
R2.6-2071-2100	2.41	2.52	2.47	8.19
R4.5-2041-2070	3.16	3.32	3.24	5.63
R4.5-2071-2100	4.08	4.23	4.16	8.89
R8.5-2041-2070	3.63	3.90	3.77	9.34
R8.5-2071-2100	6.63	7.11	6.87	9.34

Table 5.2: Projected changes in the annual climatic variables

\*Tmax, Tmin, Tavg, PCP are maximum, minimum, and mean temperature and precipitation, respectively

Similarly, there was also a general increase in annual precipitation between the different scenarios and periods. During the midcentury, the average annual precipitation increased by 9.8, 5.63, and 9.3% relative to the baseline period under RCP2.6, RCP4.5, and RCP8.5, respectively. Yearly average precipitation would increase by the end of the century (2071-2100) by 8.2, 8.9, and 9.34% for the same scenarios, respectively. Rainfall inter-seasonal variation may have been masked using the annual values. Monthly and seasonal analysis showed that the highest increase in precipitation is expected during spring, followed by fall and winter. On the other hand, the results indicated that precipitation is predicted to decrease in the summer, ranging from 0.2 to 13.3% for all the climate scenarios except during the midcentury under RCP2.6, with a 6.3% increase. However, precipitation is projected to increase more in the nongrowing season ranging from 12.4 to 29.5% for all climate scenarios, than in the growing season, ranging from 0.54-8.7%, except during the late century under RCP 4.5 and RCP 8.5.



Figure 5.1: Seasonal distribution of minimum temperature ( $T_{min}$ ), maximum temperature ( $T_{max}$ ), average temperature ( $T_{avg}$ ), and precipitation (PCP) for baseline, midcentury (2041-2070), and end of the century (2071-2100) periods under three climate scenarios (RCP 2.6, RCP 4.5, and RCP8.5). The standard error represents uncertainty in the data

Overall, the projected changes in precipitation and temperature indicate a potential warmer climate and drought conditions in the growing season. The results of this study are comparable to other research in the Canadian Prairies (Vincent et al. 2018, Bush and Lemmen 2019, Zhang et al. 2019, Chipanshi et al. 2021). For example, the annual mean temperature for the study area is projected to rise between 4-8.2 °C and precipitation 7-10% between 2021-2100 under a high scenario (http://climatedata.ca).

## 5.3.2 Projected changes in the hydrology under free drainage and controlled drainage

#### 5.3.2.1 Subsurface drainage

DRAINMOD simulations (Table 5.3) showed that the average annual subsurface drainage is predicted to decrease in the future compared to the baseline period, even though the decrease is not statistically different from the free drainage system. The mean annual subsurface drainage during the midcentury is predicted to decrease by 11.1, 17.2, and 15.2% under RCP2.6, RCP4.5, and RCP8.5, respectively. Subsurface drainage will decrease by 19.2, 13.1, and 11% for the same scenarios by the end of the century.

During the baseline period, more than 51% of subsurface drainage was accounted for during the growing season, while the remainder occurred during the nongrowing season. However, the reverse was observed for all climate scenarios in the future. Seasonal analysis showed that the increase in subsurface drainage during the nongrowing seasons would come from the winter period during the late century under RCP2.6, RCP4.5, and RCP8.5, spring period midcentury for RCP4.5 and RCP 8.5 and late century RCP 4.5. Across all the future climate scenarios and baseline periods, subsurface drainage was highest during the spring and lowest during winter except during the late century under RCP8.5. This is due to temperatures falling below the freezing point during the winter. Randall and Goss (2011) reported that freezing reduces subsurface drainage during winter, especially in seasonally frozen soils. Changes and temperature rises could be the main factor driving subsurface drainage during the spring. This could lead to waterlogging and delay in planting as farm equipment will not be able to access and work in the field due to decreased soil trafficability. The results of this study agree with Pease et al. (2017) but disagree with Hanke (2018), Dayanni et al. (2012) and Singh et al. (2009). They reported that predicted temperature rise resulted in increased subsurface drainage during the winter. The results may be due to differences in climate projections and models used in the study area.

DRAINMOD simulations under controlled drainage showed a significant difference between future climate scenarios and the baseline period. Also, there were further reductions in subsurface drainage in the future compared with the free drainage systems. Results showed that subsurface drainage differed significantly from the baseline period, ranging from 21 to 29.6%. The effect of controlled drainage was more evident during May-September. The reduction in subsurface drainage during the growing season is more significant than during the nongrowing season, unlike the free drainage system. As an important water management strategy, this confers a considerable advantage over free drainage by reducing flow and nutrient loss through the drains under future climatic conditions. Cordeiro et al (2014) and Satchithanantham et al. (2014) reported a significant reduction in nutrient export from fields under controlled drainage compared to free drainage during the spring snowmelt period. However, the need for access to the field for land preparation during the spring may not be conducive to operating the drainage control structures to hold the water back in the field. One way to overcome this limitation is to pump the water out of the drainage control structures into off-farm water storage reservoirs. This stored water could subsequently be used for irrigation to alleviate the drought stress during the growing season under projected climate scenarios.

# 5.3.2.2 Runoff

Simulation results show that the average annual runoff is predicted to rise under free drainage for all climate scenarios except during the late century for RCP8.5 (Table 3). The increase ranged from 1.2% during the late century for RCP 4.5 to 24.1% during the late century for RCP 2.6. Runoff losses during the nongrowing season were more than the growing season for both the baseline and future periods, with the highest occurring during spring and constituting about 75-84%. Spring is the transition period between the nongrowing and growing season and is usually marked by high snowmelt and soil thawing processes. With a projected increase in rainfall, more runoff loss is expected. Pagán et al. (2016) stated that climate change is worsened in water-stressed regions where runoff is dominated by snowmelt. The increase in rainfall could result in a decrease in infiltration and an increase in runoff. Since rainfall in the region is characterized by high intensity and short duration (Satchithanantham and Sri Ranjan 2015), more runoff is expected. Also, the increase in rainfall during winter and spring alongside a rise in temperature across the seasons, especially winter and spring, would result in less snow accumulation and more runoff.

Nevertheless, surface runoff is predicted to decrease during the growing season and increase during the nongrowing season compared with the baseline period. This is due to the decreased rainfall in the summer period. Also, runoff during the winter is expected to increase due to the projected temperature rise relative to the baseline period.

As expected, runoff loss was more under controlled drainage (CD) than free drainage (FD), ranging from 8 to 24%. Generally, CD decreases subsurface drainage and increases soil water storage, leading to high water table and runoff potentials. One of the trade-offs associated with controlled drainage is the increase in surface runoff. This could be a potential pathway for nutrient loss in the subsurface drained field (King et al. 2015). According to Hanke (2018), more than ten times the total phosphorus loss would occur via surface runoff with controlled drainage in a simulated study in Ontario. They concluded that controlled drainage would result in increased water quality problems now and in the future.

	Free Drainage (FD)						Controlled Drainage (CD)								
Hydrologic		Annual							Annual						
parameter	Periods	average	GS	NGS	Winter	Spring	Summer	Fall	average	GS	NGS	Winter	Spring	Summer	Fall
Drainage	Baseline														
(mm)	(1981-2010)	99	51	48	6	58	20	15	71	34	37	3	45	5	18
	RCP2.6-2041-2070	88	41	47	5	50	18	15	56	23	33	3	35	4	14
	RCP2.6-2071-2100	80	36	44	7	48	14	11	50	22	28	3	34	2	11
	RCP4.5-2041-2070	82	39	43	5	62	11	4	55	24	31	4	47	1	3
	RCP4.5-2071-2100	86	34	52	6	61	11	8	52	18	34	4	44	2	6
	RCP8.5-2041-2070	84	39	45	6	60	11	7	52	22	30	4	45	2	5
	RCP8.5-2071-2100	88	26	62	14	56	5	13	57	17	40	9	38	1	11
	Baseline														
Runoff (mm)	(1981-2010)	83	31	52	0	68	11	4	106	39	67	0	87	13	6
	RCP2.6-2041-2070	98	25	73	4	82	9	3	122	36	86	4	100	13	5
	RCP2.6-2071-2100	103	25	78	7	82	7	7	123	32	91	7	101	8	7
	RCP4.5-2041-2070	88	25	63	2	77	6	3	108	34	74	3	94	7	4
	RCP4.5-2071-2100	84	19	65	6	68	8	2	105	26	79	7	87	9	2
	RCP8.5-2041-2070	98	26	72	3	83	9	3	119	35	84	3	103	10	3
	RCP8.5-2071-2100	72	15	57	10	54	2	6	90	22	68	11	71	2	6
Evapotranspiration	Baseline														
(mm)	(1981-2010)	306	287	19	0	27	240	39	310	290	20	0	27	241	42
	RCP2.6-2041-2070	348	322	26	0	39	269	40	358	332	26	0	39	273	46
	RCP2.6-2071-2100	342	318	24	0	37	268	37	354	331	23	0	37	275	42
	RCP4.5-2041-2070	345	320	25	0	38	275	32	355	330	25	0	38	279	38
	RCP4.5-2071-2100	362	327	35	0	50	272	40	375	339	36	0	50	279	46
									•						

Table 5.3: Summary of DRAINMOD hydrologic simulation

	RCP8.5-2041-2070	351	318	33	0	43	264	44	363	331	32	0	43	271	49
	RCP8.5-2071-2100	377	328	49	1	67	272	37	388	339	49	1	68	277	42
Water table depth	Baseline														
(cm)	(1981-2010)	107	106	107	116	76	113	122	93	94	92	100	61	103	106
	RCP2.6-2041-2070	111	120	104	112	76	124	130	100	108	94	102	62	115	120
	RCP2.6-2071-2100	117	126	110	116	80	133	138	105	115	99	105	64	124	129
	RCP4.5-2041-2070	125	130	121	130	71	138	160	118	122	116	126	62	132	154
	RCP4.5-2071-2100	120	130	113	123	69	137	150	109	119	102	113	53	129	142
	RCP8.5-2041-2070	121	132	113	118	73	143	149	111	122	104	109	59	136	142
	RCP8.5-2071-2100	118	141	103	108	68	152	146	111	133	95	102	53	146	145

FD: Free drainage, CD: Controlled drainage; GS: growing season (May-September); NGS: Nongrowing season (October-April); Winter (December-February), Spring

(March-May), Summer (June-August); Fall (September-November)

#### 5.3.2.3 Evapotranspiration (ET)

The ET is an important component and a significant driver of the energy balance and water cycle in agricultural fields. In this study, ET accounts for more than 90% of the rainfall lost to the atmosphere during the growing season. ET is highest in the summer (May-September), followed by fall and spring, with negligible amounts during the winter. Results showed that the mean annual ETs from the climate scenarios were significantly different (p<0.0001) from the baseline period, with increases ranging from 42.3 to 49.7%.

Similarly, water loss by ET is higher when compared with the free drainage system, even though the difference was not significant. This is expected because CD generally decreases subsurface drainage while increasing ET and surface runoff, as confirmed in simulation and field studies (Skaggs et al. 2010). The simulation results show that controlled drainage reduced subsurface drainage and outflow under the different climate scenarios by 21-29% and 3-15%, respectively, due to a 1.8-15% increase in surface runoff. The projected rise in temperature is the major cause of water loss, mainly through soil evaporation and plant transpiration under both free drainage and controlled drainage. Previous studies in the same area have shown that temperature has a strong positive correlation with ET (Martel et al. 2018, Ndulue and Sri Ranjan 2021). As temperature rises, the radiative energy required to evaporate water increases, resulting in prolonged drought conditions.

### **5.3.2.4** Water table depth (WTD)

The mean WTD during the baseline period is within the drain installation depth. Simulation results indicate that water depth is predicted to be slightly deeper ranging from 110 to 124.8 cm, and deeper during the growing season, from 120 to 140.6 cm during the 21<sup>st</sup> century. The springtime happens to have the shallowest WTD for all the future climate scenarios and the baseline period. This is due to snowmelt happening around this period. A rise in temperature during the springtime, as predicted, could have also contributed to shallow water depth relative to the baseline period. During the summer period, all climate scenarios had water tables deeper than the baseline in the future. The result agrees with Kurki-Fox et al. (2019), who found out that climate change will cause the water table to fall. Under CD, the mean WTD was shallower in relation to the free

drainage. With the low hydraulic conductivity of the heavy clay in the study area, capillary flux may be minimal even though the water table is within the drain depth. The WTD changes under CD during the season were similar but differed in magnitude (Table 3) to that observed under free drainage.

## 5.3.3 Projected changes in relative yield of canola

Relative to the baseline period, the canola yield showed a decreasing trend with time for all the RCPs. The mean annual relative yields during the midcentury and late century were significantly lower than the baseline under RCP8.5 for both FD and CD (Table 5.4). The DRAINMOD model computes crop yield by considering stressors due to excess water, drought, and planting delays. Analysis of the stressors shows that relative yield due to planting delay gave mixed results under different climate scenarios and water management systems. However, relative yield due to wet stress will decrease in the future, while relative yield due to dry stress will increase. This is due to the marginal increase in average annual precipitation, reduced rainfall during the growing season (summer), and an overwhelming increase in temperature, especially during the growing season (June-August). Heat stress would lead to more water loss and drought conditions. The negative impacts of heat and drought on canola yield have been confirmed by numerous field studies (Angadi et al. 2000, Gan et al. 2004, Elferjani and Soolanayakanahally, 2018, Hammac et al. 2018, Wu et al. 2018). Angadi et al. (2000) and Gan et al. (2004) reported a maximum yield reduction of 53 and 77% under heat and drought stress conditions during the flowering stage, respectively. Elferiani and Soolanayakanahally, (2018) reported 89% reduction in seed yield due to drought and temperature stress. These studies agree with the predicted effects of high temperature and drought on canola yield using DRAINMOD.

Simulation results showed that during the midcentury under RCP 2.6, RCP 4.5 and RCP 8.5, relative yield due to wet stress was similar in magnitude under free drainage and controlled drainage. It was highest during the late century for all RCPs. Also, all climate scenarios showed that yield decrease was mainly caused by drought stress under free drainage and controlled drainage, and its impacts increased with increasing RCPs. This is confirmed in the DRY DAYS values, which indicate the number of days with deficit soil water conditions. During the late century for RCP 8.5, the number of dry days was more than ten times compared to the baseline

period. Relative yield under controlled drainage was lower than free drainage for all the climate scenarios. This may be due to the number of WORKDAYS and SEW-30 (sum of excess water within the 30 cm rootzone) values (Table 5.4). The simulation results agree with other studies that reported low or no significant crop yield under water management systems that depend on capillary flow, such as controlled drainage and subirrigation. This is due to the low hydraulic conductivity of the soil, which declines rapidly as soil moisture decreases, resulting in reduced capillary flow from the water table to the plant roots (Skaggs 1978). Gunn et al. (2018) reported that the lower yield in field plots with clay was attributed to the lower soil hydraulic conductivity.



Figure 5.2: Relative yield due to different stressors (planting delay, wet stress and dry stress) as a function of time (baseline, mid-century, and late-century periods) under free drainage and controlled drainage under three climate scenarios (RCP2.6, RCP4.5, and RCP8.5)

Climate scenario &	SEW-30	Dry days	Workdows	Overall Palative Viald (%)*
Water Management	(cm-day)	DIY days	workdays	Overall Relative Tield (%)
FD-BASELINE	119.2	4.5	78.7	84.6 a
FD-RCP-2.6-2041-2070	113.4	13.0	91.4	82.6 a b
FD-RCP-2.6-2071-2100	77.9	19.1	95.6	79.7 a b
CD-BASELINE	130.7	4.6	68.3	79.2 a b
CD-RCP-2.6-2041-2070	84.8	11.8	80.1	78.1 a b
CD-RCP-2.6-2071-2100	85.3	17.3	84.7	76.8 a b c
FD-RCP-4.5-2041-2070	63.2	22.3	93.0	75.0 a b c d
FD-RCP-4.5-2071-2100	52.3	27.1	95.2	72.2 a b c d
CD-RCP-4.5-2041-2070	64.3	21.1	89.6	71.6 a b c d
CD-RCP-4.5-2071-2100	56.5	25.6	84.3	69.6 b c d e
FD-RCP-8.5-2041-2070	96.6	32.2	94.9	62.6 c d e f
CD-RCP-8.5-2041-2070	108.0	30.9	86.3	62.1 d e f
FD-RCP-8.5-2071-2100	53.6	45.1	99.7	56.7 e f
CD-RCP-8.5-2071-2100	49.2	44.1	88.1	54.5 f

Table 5.4: Canola yield, SEW<sub>30</sub>, Dry days, and workdays under different climate scenarios and water management system

\*Different letters are significantly different based on Tukey's means separation at 5% significance level.

From the above simulation results, water managers and producers can develop strategies to deal with the predicted impacts of climate change. One of such strategies is to capture, store, and reuse the large quantity of drainage water from snowmelt usually loss before planting. This could reduce vulnerability to heat and drought stress and improve canola productivity. Studies have identified irrigation as an adaptive strategy to climate change (Zou et al. 2012, Kresovic et al. 2014, Chartzoulakis and Bertaki 2015, Kashyap and Agarwal 2020, Rosa et al. 2020, Thiery et al. 2020). Water saving techniques such as deficit irrigation or alternate wetting and drying can also be practiced. This would not only save water but also reduce GHG emission (Setyanto et al. 2018). Another strategy is to engage in management practices that would not only increase ground water storage but also retain residual soil water content such as mulching, cover crop, crop residue, and zero tillage. Other adaptive and mitigation practices to heat and drought stress include modification

of planting dates, use of drought-resistant cultivars, improved breeding techniques (Wu et al. 2018, Raza et al. 2019), and reduction of activities that encourage GHG release.

## **5.4 Conclusion**

Climate change is predicted to alter the hydrologic cycle and crop yield globally. The impacts and severity may vary from location to location. Understanding how these projected changes can affect hydrology and crop yield is important for developing climate change adaptation strategies using water management, especially in the Canadian Prairies because it is Canada's food basket. The DRAINMOD model was parameterized using the 2019 and 2020 water table data measured at the PESAI research site. Bias-corrected CANESM2 GCM participating in the CMIP5 under RCP 2.6, 4.5, and 8.5 were used to drive the DRAINMOD model. The projected changes in the climatic variables indicate that even though the mean annual precipitation is projected to increase, rainfall is expected to decrease during the growing season, and the climate becomes warmer.

DRAINMOD simulation results suggest that in the future, subsurface drainage is expected to decrease due to lower growing season precipitation. In contrast, ET and surface runoff are expected to increase under all climate scenarios except under RCP 8.5 for runoff during the late century. Also, simulation results indicate that the average canola yield is predicted to decrease by 2.5-32.9% due to drought stress caused by the rise in temperature and reduced growing season rainfall. The capture and storage of off-season drainage water in reservoirs for subsequent irrigation could be one strategy to mitigate increased growing season dry stress under all climate scenarios as an adaptation method to overcome the impact of climate change, as reported in this research. Both field and modelling studies have shown that drainage water recycling (DWR) has not only increased crop yield but improved water quality (Frankenberger et al. 2017, Kaur et al. 2021, Moursi et al. 2022). Even though DWR could reduce farm size and increase cost, Frankenberger at al. (2017) argued the long-term benefit in yield would offset the cost.

#### **Chapter Six**

# Performance of the FAO Penman-Monteith Equation under limiting conditions and fourteen Reference Evapotranspiration Models in Southern Manitoba

## Abstract

Evapotranspiration is a key component of water and energy balance. An accurate estimate of reference evapotranspiration (ET<sub>o</sub>) is important for determining the water demand of field crops, water management, and hydrological modelling. The FAO-PM  $ET_{o}$  equation is the standard equation for estimating  $ET_0$ . Its use is limited by the requirement for too many observed inputs that are not readily available in most weather stations. Empirical models requiring readily available inputs have been developed as an alternative. However, their performance is location specific. Therefore, this study assesses the performance of the FAO-PM ET<sub>o</sub> computed with limited data and fourteen empirical  $ET_0$  models. Meteorological data (2012-2019) was analyzed under the semi-arid climate conditions in southern Manitoba. Model performance was assessed using statistical indices, including R<sup>2</sup>, RMSE, NSE, MPE, and MAE. Results showed that ET<sub>o</sub> estimates under missing wind speed, relative humidity, or solar radiation were acceptable, although model performance decreased with increased missing data yielding average to poor performance. Based on R<sup>2</sup>, RMSE, and NSE values, among the 14 models compared, the best performing models are Valiantzas-1, Valiantzas-3, Irmak, Valiantzas-2, and Priestly-Taylor model. New empirical coefficients were developed, requiring one or two climatic inputs to improve empirical models. Results showed that the calibrated models performed better than the original equation. The sensitivity analysis showed that ET<sub>o</sub> is most sensitive to maximum temperature (T<sub>max</sub>) and solar radiation (R<sub>s</sub>), followed by vapour pressure deficit (VPD), wind speed (U<sub>2</sub>), and minimum temperature (T<sub>min</sub>). Therefore, it is recommended to ensure accurate measurements of temperature and solar radiation for accurate ET<sub>o</sub> estimates. This study provides alternative ET<sub>o</sub> models with lesser inputs, reasonable and accurate ET<sub>o</sub> estimates for southern Manitoba and other areas with similar climate characteristics.

Keywords: Reference evapotranspiration, ET<sub>o</sub> Models, Prairies, southern Manitoba.

## **6.1 Introduction**

Water resources management is a global issue exacerbated by climate change, population explosion, and poor water management (Duran-Encalada et al. 2017). Efficient and sustainable

water management in agriculture accurately determines the irrigation water requirements of crops. Efficient water management implies applying water only, when necessary, at the right quantity and time (Sharma and Irmak, 2012). An accurate estimate of evapotranspiration ensures optimal water use. Evapotranspiration (ET) is a combined evaporation and transpiration term, which describes water loss through the soil surface and stomatal openings in leaves. These processes cooccur, which makes it difficult to separate them (Allen et al. 1998). It is challenging to estimate because of the complex plant-soil interactions (Singh and Xu, 1997). After precipitation, ET is the most important component of the hydrological cycle (Alexandris et al. 2008). The determination of the crop evapotranspiration ( $ET_c$ ) is usually preceded by calculating reference evapotranspiration (López-Urrea et al. 2006). The term "reference evapotranspiration ( $ET_o$ ), as defined by Allen et al. (1998), refers to the evapotranspiration rate from a well-watered hypothetical grass surface of 0.12 m in crop height, an albedo of 0.23 and surface resistance of 70 m/s." The ETo is essential for determining irrigation water requirement of crops (Allen et al. 1998), irrigation scheduling, planning, and management (Sentelhas et al. 2010), ecological and climate change studies (Nistor et al. 2017), and hydrological modelling (Schneider et al. 2007).

There are two methods for determining ETo: direct and indirect. Direct methods are based on the principle of water balance and water vapour transfer. They include the use of lysimeter and eddy covariance methods (Shuttleworth, 2008). These methods are accurate and give better results than any other method, but they are usually expensive, time-consuming and requires skill and experience (Allen et al. 1998). The indirect method involves using meteorological data from a weather station or pan observation. They are usually empirical and or physically based equations relating  $ET_0$  to weather parameters. The FAO Penman-Monteith equation (FAO-PM) is the recommended standard equation for estimating  $ET_0$  because it correlates strongly with lysimeter data and provides consistent  $ET_0$  values in a wide range of locations and climates (Allen et al. 1998). Thus, it is widely adopted as the benchmark for evaluating other empirical  $ET_0$  models.

Despite its accuracy, robustness and global acceptance, the use of the FAO-PM method is limited because it requires many data inputs absent in most weather stations. For most regions, including Canada, most stations measure only temperature and precipitation (Maule et al. 2006). Under data-scarce conditions, Allen et al. (1998) provided alternative methods for estimating missing solar radiation, relative humidity, and wind speed. Most studies, including Jabloun and Sahli (2008), Djaman et al. (2018), and Koudahe et al. (2018), have reported good performance of the FAO-PM

equation under limiting conditions. Similar results were also reported by Sentelhas et al. (2010) and C'ordova et al. (2015) except during missing solar radiation.

Empirical ET<sub>o</sub> models with fewer climatic inputs have been developed as an alternative to the FAO-PM model. However, their performance is location specific. In southern Greece, with a semiarid Mediterranean climate, Xystrakis & Matzarakis (2011) evaluated thirteen ET<sub>0</sub> models. Their results showed that the Hanson and Turc models could serve as an alternative to the FAO-PM equation. Under the Sahelian climate in Senegal, Djaman et al. (2015) showed that Valiantzas, Trabert, Romanenko, Schendel, and Mahringer equations performed well compared to the FAO-PM equation. Tabari et al. (2013) evaluated the performance of thirty-one ET<sub>o</sub> models under a humid climate in Iran. Their results showed that two developed radiation-based equations, Blaney-Criddle, modified Hargreaves (M4), and the Snyder pan evaporation-based equation, performed well compared to the FAO-PM equation. Likewise, the Turc equation ranked first above four other ET<sub>o</sub> equations under a humid climate (Trajkovic and Kolavic 2009). Gao et al. (2017) reported that under different regions and climates, the Priestly Taylor, Hargreaves Samani, and Makkink equations were the best for arid, semi-arid, and humid climates respectively. Under humid continental climate, Sentelhas et al. (2010) found that simple ET<sub>o</sub> models, after calibration, could serve as an alternative to the FAO-PM equation. In the Canadian Prairies, with semi-arid and subhumid climates, Maulé et al. (2006) developed simple ET<sub>o</sub> and compared it with the traditional Linacre and Baier-Robertson ET<sub>o</sub> models. They reported that the Maule models gave accurate and precise ET<sub>o</sub> estimates. Similarly, Martel et al. (2018) compared the performance of seven ET<sub>o</sub> models against the FAO-PM equation in the Canadian Prairies. They reported that the Maule model ranked first and performed best while the Blaney-Criddle ranked worse.

With so many empirical  $ET_0$  models, there is no rule of thumb regarding the  $ET_0$  equations under various climate because of the mixed results obtained even under the same climatic conditions (Paparrizos et al. 2014) and also, lack of objective criteria in model selection (Singh and Xu 1997). Therefore, it becomes necessary to assess the performance of empirical  $ET_0$  models before their application in a location different from that for which it was initially developed. Although considerable studies have been carried out on empirical  $ET_0$  model comparison across the globe, including Canada, the global changes in climate (Azhar and Perera 2011, USGRP 2018), the lack of long-term data, and the diverse nature of the Canadian Prairies justify the need to continuously evaluate  $ET_0$  models especially  $ET_0$  models not yet evaluated and under limiting conditions. The Canadian Prairies is vast, supporting various grain and oil crops such as wheat and canola in large quantities. For example, the Prairies contribute about 50% of Canada's grain (Bueckert and Clarke 2013) and also is the highest producer of canola in Canada (Statistics Canada, 2018). From a detailed literature search, there is no information on the performance of recently developed models like the Valiantzas  $ET_0$  models in Canada. Thus, the objectives of this study were to (i) assess the performance of FAO-PM under missing data conditions. (ii) analyze the performance of fourteen empirical ETo models relative to the FAO-PM equation, (iii) calibrate and develop new empirical constants, and (iv) determine the most sensitive climate variables influencing  $ET_0$  in southern Manitoba.

#### 6.2 Materials and Methods

## 6.2.1 Weather data

Weather data was obtained from the Winkler weather station in southern Manitoba (Fig. 6.1). The station is located at latitude 49.122°N; longitude 97.932°W, elevation 279 m, and managed by Manitoba Agriculture. The weather station is located in the midst of rural agricultural lands as shown in Fig. 1. Weather data collected included solar radiation (MJ/m<sup>2</sup>/day), minimum and maximum air temperature measured at 2 m (°C), relative humidity (%) and wind speed (m/s) at 10 m for a 7-year period (2012-2019). Routine checks and equipment maintenance are usually followed to ensure data quality and integrity (Ojo, 2020, personal communication). Manitoba has a semi-arid climate, with annual precipitation of 450 to 520 mm, an average annual temperature of 2.2 °C, an average minimum temperature of -16 °C in January and an average maximum temperature of 23 °C in July (Bueckert and Clarke 2013). The textural class of the area is sandy loam. Southern Manitoba is the center of agricultural activities in the province, accounting for more than 60% of its agricultural employment in 2017 (Statistics Canada, 2017).



Figure 6.1: Location of the study area form Manitoba Agriculture (https://agrimaps.gov.mb.ca/agrimaps/#)

# 6.2.2 The FAO-Penman-Monteith (FAO-PM) Equation

The FAO-PM is the standard equation for determining reference crop evapotranspiration, and it was used for estimating daily  $ET_0$  in this study. All other empirical  $ET_0$  models were compared to the FAO-PM equation. According to Allen et al. (1998), it is given as:

$$ETo = \frac{0.408\Delta(R_n - G) + \gamma \frac{900}{T + 273} [e_s - e_a] u_2}{\Delta + \gamma * (1 + 0.34 * u_2)}$$
(6.1)

where  $ET_0$  is the reference crop evapotranspiration [mm day<sup>-1</sup>];  $R_n$  is the net radiation [MJ m<sup>-2</sup> day<sup>-1</sup>]; G is the soil heat flux [MJ m<sup>-2</sup> day<sup>-1</sup>]; T is the average daily air temperature at the height of 2 m [°C];  $u_2$  is the wind speed at a height of 2 m [m s<sup>-1</sup>];  $e_s$  is the saturation vapour pressure [kPa];  $e_a$  is the actual vapour pressure [kPa];  $e_s - e_a$  is the vapour pressure deficit [kPa];  $\Delta$  is the slope of the saturation vapour pressure-temperature curve [kPa °C<sup>-1</sup>]; and  $\gamma$  is the psychrometric constant [kPa °C<sup>-1</sup>].

## **6.2.3 Empirical ETo models**

Based on input requirements, Tabari et al. (2013) classified empirical  $ET_o$  models into temperature-based, radiation-based, mass-transfer based, and combination-based models. As the name implies, the major requirement for temperature and radiation-based models are temperature and solar radiation data, respectively. The mass transfer-based models are based on Dalton's gas law (Tabari et al., 2013). The combination based  $ET_o$  models combine aerodynamic and energy balance components.

In this study, fourteen common, simple, and some recent empirical  $ET_0$  models (Table 6.1) were evaluated and compared with the FAO-PM equation. Temperature-based models include Hargreaves Samani, Trajkovic, Ravazzani, and Schendel model. The radiation-based model includes Irmak, Priestly-Taylor, Tabari 1, and Tabari 2 model. Mass transfer-based models include Romanenko and Penman, while combination-based models are Alexandris (Copais), Valiantzas 1, Valiantzas 2, and Valiantzas 3.

S/No	Model	Equation	Inputs
1	Hargreaves-Samani (1985)	$ET_{o} = 0.0023 * R_{a} * 0.408 * (T_{a} + 17.8)^{0.5}$	Ta
2	Trajkovic (2007)	$ET_{o} = 0.0023 * R_{a} * 0.408 * (T_{a} + 17.8)^{0.424}$	Ta
3	Ravazzani et al. (2012)	$ET_{o} = (0.817 + 0.00022z)0.0023 * R_{a}$ $* 0.408 * (T + 17.8)^{0.5}$	Ta
4	Schendel (1967)	$ET_{o} = 16 * \left(\frac{T_{a}}{RH}\right)$	Ta
5	Irmak et al. (2003)	$ET_{o} = -0.611 + 0.149R_{s} + 0.079T_{a}$	$T_{a,}R_{s}$
6	Tabari 1 et al. (2013)	$ET_{o} = -0.642 + 0.174R_{s} + 0.0353T_{a}$	$T_{a,}R_{s}$
7	Tabari 2 et al. (2013)	$\begin{split} ET_{o} &= -0.478 + 0.156 R_{s} - 0.0112 T_{max} \\ &+ 0.0733 T_{min} \end{split}$	$T_{max,}T_{min,}R_s$
8	Priestly-Taylor (1972)	$ETo = 1.26 \frac{\Delta + R_n}{2.45(\Delta + \gamma)}$	
9	Romanenko (1961)	$ET_o = (0.0018 * (T_a + 25)^2 * (100 - RH))$	$T_{a,}RH$
10	Penman (1948)	$ETo = \left(\frac{0.000479}{U_2} + 2.625\right)(e_s - e_a)$	RH, U <sub>2</sub>
11	Alexandris and Kerkides (2003) Copias approach	$ET_{o} = 0.057 + 0.227c2 + 0.643c1 + 0.0124c1 * c2 \text{ where } c1 = 0.646 - 0.078RH + 0.372R_{s} - 0.00264R_{s}RH, c2 = -0.0033 + 0.00812T_{a} + 0.101R_{s} - 0.00584R_{s}T_{a}$	T <sub>a,</sub> RH, R <sub>s</sub>
12	Valiantzas 1	$ET_{o} = 0.0393R_{s}\sqrt{T_{a} + 9.5} - 0.19R_{s}^{0.6}\phi^{0.15} + 0.048(T_{a} + 20)\left(1 - \frac{RH}{100}\right)U_{2}^{0.7}$	T <sub>a</sub> , U <sub>2</sub> , RH, R <sub>s</sub>
13	Valiantzas 2	$ET_{o} = 0.0393 * R_{s}\sqrt{T_{a} + 9.5} - 0.19R_{s}^{0.6}\varphi^{0.15} + 0.078(T_{a} + 20)\left(1 - \frac{RH}{100}\right)$	T <sub>a</sub> , RH, R <sub>s</sub>
14	Valiantzas 3	$ETo = 0.0393 * R_s \sqrt{T_a + 9.5} - 0.19 R_s^{0.6} \phi^{0.15} + 0.0059 (T_a + 20) (T_a - Tmin - 0.45 (T_{max} + T_{min}) + 3.45)^{0.8}$	Ta,Tmin,Tmax Rs,

Table 6.1: ETo models and their input requirements

Where T<sub>min</sub> is minimum temperature (°C), T<sub>max</sub> is maximum temperature (°C), T<sub>a</sub> is average daily air temperature (°C), Rs is Solar radiation (MJ/m-<sup>2</sup> day<sup>-1</sup>), RH is relative humidity (%), U<sub>2</sub> wind speed corrected to 2 m, Ra is extraterrestrial radiation (MJ m<sup>-2</sup> day<sup>-1</sup>), φ is latitude (rad)
#### 6.2.4 ET<sub>0</sub> estimates under limiting conditions

In the absence of complete weather variables, Allen et al. (1998) gave recommendations to estimate inputs for the FAO-PM equation. Solar radiation is usually estimated from the Hargreaves solar radiation equation; actual vapour pressure is related to relative humidity and is estimated using the assumption that dew point temperature is approximately equal to the minimum temperature, while a global average of 2 m/s or an average wind speed data from a close station can be used. In this study, a global average of 2 m/s was used for wind speed. The equation for estimating missing solar radiation and actual vapour pressure is presented in equation 2 and 3, respectively

$$R_{s} = K_{rs} * R_{a} * 0.408 * (T_{max} - T_{min})^{0.5}$$
(6.2)

$$e_a = 0.611 \exp\left[\frac{17.27T_{min}}{T_{min} + 237.3}\right]$$
(6.3)

where,  $K_{rs}$  is an adjustment coefficient having a value of 0.16 and 0.19 for inland and coastal locations, respectively. In this study,  $K_{rs}$  was calibrated for the study location. Empirically,  $K_{rs}$  is determined as stated by (Allen 1997)

$$K_{rs} = K_{Ra} * \left(\frac{P}{P_o}\right)^{0.5} \tag{6.4}$$

where,  $K_{Ra}$  is an empirical coefficient equal to 0.17 for interior regions and 0.20 for coastal areas, P is mean atmospheric pressure at the weather station site (kPa); P<sub>o</sub> = mean atmospheric pressure at the sea level (101.3 kPa)

#### 6.2.5 Model calibration and validation

Two-thirds of the weather data (2012 to 2017) were used to determine new empirical constants (calibration), while the remaining one-third (2018-2019) was used for validation of the calibrated equation. Model calibration was done using the non-linear generalized reduced gradient (GRG) solving method in Microsoft excel. The solver tool determines empirical constants by minimizing the sum of squared residual. Numerous studies have adopted this optimization technique (Bogawski and Bednorz 2014, Djaman et al. 2018).

### 6.2.6 Sensitivity analysis

Sensitivity analysis is usually carried out to determine the most influential or sensitive variable (Irmak et al. 2006). Across the globe, weather stations are limited, expensive to establish and maintain, and measure few variables. Therefore, it is important to identify climatic variables that are most sensitive to  $ET_o$  so that efforts can be geared towards their accurate measurements. Under scarce data conditions, Jersurki et al. (2019) and Shafieiyoun et al. (2020) have argued to apply the  $ET_o$  model based on the sensitivity to  $ET_o$  to climatic variables. Sensitivity analysis is also important in understanding the variability of climatic variables and  $ET_o$ , especially under increased climate uncertainties (Nouri et al. 2017).

In this study, a simple method that involved plotting the relative change in each of the climatic variables ( $T_{max}$ ,  $T_{min}$ ,  $R_s$ , and VPD) against relative change in ET<sub>o</sub>. Numerous researchers have applied this method (Irmak et al. 2006, Yin et al. 2010, Gao et al. 2016, Ndulue et al. 2020, Shafieiyoun et al. 2020). The VPD was used because RH exceeded the maximum value (100%) when increased by 15%, thereby rendering it physically meaningless (Biazar et al. 2019). The plot was drawn by determining the base FAO PM ET<sub>o</sub> (that is, ET<sub>o</sub> calculated from the average daily value of each climatic variable for the given study period, April to October, n = 215). New sets of ET<sub>o</sub> were determined by increasing and decreasing each climatic variable by ±5 to ±25% while keeping other variables constant. Then, a plot showing the percent change in each variable to ET<sub>o</sub> is developed. Variables with the steepest slope and highest average values are the most sensitive.

#### 6.2.7 Model performance assessment

A qualitative and statistical test assessed model performance. Qualitative assessment involves a graphical plot of empirical  $\text{ET}_{0}$  models against the FAO-PM  $\text{ET}_{0}$  model on a 1:1 scale. The slope coefficient (*b*) and the coefficient of determination ( $\mathbb{R}^{2}$ ) indicate model accuracy and precision.  $\mathbb{R}^{2}$  ranges from 0 to 1. An  $\mathbb{R}^{2}$  =1 means a perfect agreement between observed and predicted. It is calculated as:

$$R^{2} = \left(\frac{\sum_{i=1}^{n} (o_{i} - \overline{o})(P_{i} - \overline{P})}{\sqrt{\sum_{i=1}^{n} (o_{i} - \overline{o})^{2}} \sqrt{\sum_{i=1}^{n} (P_{i} - \overline{P})^{2}}}\right)^{2}$$
(6.5)

Statistical tests commonly used for model assessment (Moriasi et al. 2015, Johnson et al. 2019) to quantify mean differences between measured and predicted values were also used, including: Root Mean Square Error (RMSE)

$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (O_i - P_i)^2}{n}}$$
(6.6)

Mean Percent Error (MPE)

$$MPE = \frac{\sum_{i}^{n} (P_i - O_i)}{n} \times 100 \tag{6.7}$$

MPE is used to quantify model over or underestimation. Positive and negative values indicate overestimation and underestimation, respectively.

Mean absolute error (MAE) is a measure of the magnitude of the average error.

$$MAE = \frac{\sum_{i}^{n} |O_i - P_i|}{n} \tag{6.8}$$

Nash Sutcliffe efficiency (NSE)

NSE range form  $-\infty$  to 1. A negative NSE value is an indication that the observed mean is a better predictor than the model. NSE equal to 0 implies that the model estimates are as accurate as the observed mean. In contrast, a positive NSE value means that the model is a better predictor than the observed mean (Martel et al. 2018).

$$NSE = 1 - \frac{\sum_{i=1}^{n} (o_i - P_i)^2}{\sum_{i=1}^{n} (o_i - \overline{o})^2}$$
(6.9)

where  $O_i$  is ET<sub>0-FAO-PM</sub> estimates,  $P_i$  is the *ET<sub>0-EMPIRICAL</sub>*,  $\overline{O}_i$  is the mean of ET<sub>0-FAO-PM</sub>, and *n* is the total number of observed data points. A perfect model is obtained when NSE, R<sup>2</sup>, and *b* is equal to 1, and RMSE, MPE and MAE is equal to 0 (Moriasi et al. 2015, Johnson et al. 2019).

## **6.3 Results and Discussions**

### 6.3.1 Climatic conditions of the study area

Table 6.2 summarizes the study area's measured daily climate variables during the study period (April to October). The average monthly analysis showed that the highest Tmax, Tmin, and Rs were recorded in July, corresponding to the period of highest water use for most field crops such as canola (Alberta Agriculture and Rural Development, 2011). Lowest  $T_{max}$ ,  $T_{min}$ , were also observed in April, whereas the lowest Rs was observed in October.

	T <sub>max</sub> (°C)	T <sub>min</sub> (°C)	RH (%)	Rs (MJ/m <sup>2</sup> /day)	$U_2^2$ (m/s)
Minimum	3.56	-6.48	49.08	4.71	1.03
Maximum	29.22	16.05	80.52	26.21	3.76
Average	19.87±7.12	$6.88 \pm 5.88$	$67.78 \pm 6.28$	16.69±5.17	2.15±0.53
30yr-average <sup>3</sup> (1981-2010)	19.23±6.36	7.59±5.95	-	-	-

Table 6.2: Daily average climatic variables at the study site from 2012-2019<sup>1</sup>

<sup>1</sup>2016 was excluded from the analysis because of incomplete solar radiation data

 $^{2}$  wind speed corrected to 2 m height.

<sup>3</sup> Historical data from the Environment Canada website

(https://climate.weather.gc.ca/climate\_normals/results\_1981\_2010\_e.html?stnID=3626&autofwd=1)

# 6.3.2 Relationship between ETo and weather variables

The relationship between the FAO-PM ETo estimates and meteorological variables (solar radiation, vapour pressure deficit, average temperature, relative humidity, and wind speed) is presented in Figure 6.2. Figure 6.2 shows that  $\text{ET}_0$  has a strong positive correlation with vapour pressure deficit, solar radiation, and temperature ( $\mathbb{R}^2 > 0.5$ ). In contrast, a weak correlation was observed with relative humidity and wind speed ( $\mathbb{R}^2 < 0.5$ ). The strength of correlation among the weather variables could also indicate factors controlling the evapotranspiration process. The results suggest that the area's evapotranspiration process is more energy-driven than the water vapour transfer process. This agrees with other studies in arid and semi-arid climates (Preeyaphorn 2011, Martel et al. 2018, Djaman et al. 2018, Koudahe et al. 2018). However, Martel et al. (2018) observed a negative relationship between  $\text{ET}_0$  and wind speed. The relationship between  $\text{ET}_0$  and weather variables indicates the accuracy of  $\text{ET}_0$  estimates under missing data conditions. Thus, it is likely that under missing solar radiation,  $\text{ET}_0$  estimates would be of lesser accuracy when compared with missing wind speed.



Figure 6.2: Linear regression of ETo and (a). solar radiation (b). wind speed (c). vapor pressure deficit (d). average temperature (e). relative humidity

## 6.3.3 ET<sub>0</sub> assessment under missing data

As stated earlier, the recommendations of Allen et al. (1998) were followed under the missing data scenario. In this study,  $ET_o$  estimates under missing wind speed, relative humidity, solar radiation is represented as  $ETo-U_2$ ,  $ET_o$ -RH and  $ET_o$ -Rs, respectively. Other combinations and statistical results are presented in Table 6.3 and shown in Figure 6.3.



Figure 6.3: Regression of ET<sub>o</sub> computed with complete and incomplete datasets.

Fig. 6.3 shows the linear regression of the FAO PM ET<sub>o</sub> computed with full datasets and FAO PM ET<sub>o</sub> computed with missing data. As expected, the accuracy of FAO PM ET<sub>o</sub> estimates decreased as the number of missing variables increased. Statistical results showed that ET<sub>o</sub> computed when only one variable is missing performed best with R<sup>2</sup> and NSE > 0.9 and RMSE < 0.35 mm/day. Under this scenario, ET<sub>o</sub>-U<sub>2</sub> performed best, followed by ET<sub>o</sub>-RH and ET<sub>o</sub>-Rs. The above results suggest that ET<sub>o</sub> estimate under limiting data condition is acceptable. However, a comparison of measured Rs and Rs computed using the Hargreaves equation showed that the Hargreaves equation

accounted for 63% ( $R^2 = 0.63$ ) of the variation in measured Rs, and an overwhelming overestimation. Calibration of the solar radiation constant, Krs, has been reported to improve Rs and ET<sub>o</sub> estimates (Aladenola & Madramootoo 2014, Tabari and Talaee 2016). The result herein agrees with the findings of Ngogondo et al. (2013), Popova et al. (2006), Jabloun and Sahli (2008), C'ordova et al. (2015), and Djaman et al. (2018). Under the ETo-U<sub>2</sub> scenario, Sentelhas et al. (2010) reported precision of 96 to 99% and RMSE < 0.3 mm/day across all Ontario stations. Under semi-arid climates, Jabloun and Sahli (2008), Koudahe et al. (2018), and Djaman et al. (2018) reported a precision ranging from 87 to 91%. On the other hand, Shafieiyoun et al. (2020) found that ETo computed under data-scarce conditions were unacceptable.

When two variables are missing, the FAO PM ET<sub>o</sub> decreased, although ET<sub>o</sub>-Rs,U<sub>2</sub> and ET<sub>o</sub>-RH, U<sub>2</sub> performed relatively well with NSE > 0.8 and RMSE of about 0.55 mm/day. Under this condition, ET<sub>o</sub>-Rs, U<sub>2</sub> and ET<sub>o</sub>-RH, U<sub>2</sub> gave similar results and performed better than ET<sub>o</sub>-Rs, RH. This may be because RH has a stronger correlation with ET<sub>o</sub> than U<sub>2</sub> (Figure 6.2). The poorest performance was obtained when all three weather variables were missing. This condition overestimated ET<sub>o</sub> by more than 36% and RMSE > 0.82 mm/day. The results of this study agree with Djaman et al. (2018), who reported an R<sup>2</sup> ranging from 0.723 to 0.84, slope coefficient of 0.723 to 1.104, RMSE of 0.85 to 1.10 mm/day, and MAE 0.7 to 1.43 mm/day across five stations in New Mexico, while Sentelhas et al. (2010) reported low regression coefficient between 0.08 to 0.49 and an average RMSE of 1.194 mm/day across sites in Southern Canada.

# 6.3.4 Comparison of ET<sub>0</sub> models

All the selected empirical  $ET_0$  equations were compared with the FAO-PM equation, the standard  $ET_0$  equation. Regression and statistical analyses showing their performance are presented in Figure 6.4. The empirical  $ET_0$  models exhibited varying performance in the study area. Overall, among the 14 models, the Valiantzas-1 model performed the best while the Romanenko model performed the worst.

MODELS	RMSE (mm/day)	MPE (%)	MAE (mm/day)	NSE
ETo-U <sub>2</sub>	0.335	17.41	0.234	0.951
ETo-RH	0.367	13.67	0.276	0.941
ETo-Rs	0.467	19.65	0.328	0.905
ETo-Rs,U <sub>2</sub>	0.554	16.92	0.416	0.866
ETo-RH,U <sub>2</sub>	0.555	18.14	0.429	0.866
ETo-RH,Rs	0.683	31.9	0.516	0.796
ETo-RH,Rs,U <sub>2</sub>	0.822	36.19	0.641	0.705

Table 6.3: ETo computed with complete data sets Versus ETo computed with missing parameters

#### 6.3.4.1 Combination-based models

The statistical results showed that combination-based models performed best for estimating ET<sub>o</sub>. Of the combination-based models evaluated, the Valiantzas-1 model performed the best based on its statistical indices. It can also be observed that almost all the scatter plots' points fall on the regression line (Figure 6.4). Although the Valiantzas-1 model has the same weather data requirement as the FAO-PM, it is simple and involves the direct substitution of variables, unlike the FAO PM model. Thus, the Valiantzas-1 model could be used in place of the FAO-PM equation, which agrees with Valiantzas (2013), who used global data sets, Djaman et al. (2018) in New Mexico, Ahooghalandari et al. (2017) in Australia, Djaman et al. (2015) in Senegal, Valipour (2014) in Iran, and Bourletsikas et al. (2017) in Greece. These studies have reported precision and accuracy greater than 98% and low RMSE values. In contrast, Peng et al. (2017) and Valle Júnior et al. (2020) reported poor performance of the Valiantzas ET<sub>o</sub> models in China and Brazil, respectively. Of the four models, the Copias model had the lowest performance, with the lowest NSE (0.77) and the highest RMSE (0.73 mm/day). This disagrees with Valle Júnior et al. (2020), who reported that the Copias model performed better than Valiantzas-2 and Valiantzas-3 models. In Australia, Alexandris et al. (2008) reported a slope coefficient of 1.005 and an RMSE of 0.281 mm/day with the Copais equation. Bourletsikas et al. (2017) also reported similar performance. From Figure 6.4, it is also seen that Valiantzas-3 performed better than Valiantzas-2 despite requiring lesser inputs (solar radiation and temperature). However, Valipour (2014) reported that the Valiantzas-2 (Rs, T, RH) gave accurate ET<sub>o</sub> estimates under Iran's limited data. The MPE

values indicated that Valiantzas-3, Copias, and Valiantzas-1 models overestimated  $ET_0$  while Valiantzas-2 underestimated  $ET_0$ .

### 6.3.4.2 Radiation-based models

The radiation-based models closely follow the combination-based models with good model performance. It is seen that  $R^2$  is greater than 0.8, slope coefficient ranged between 0.78 to 0.98, and had low RMSE values. The Irmak model performed best with an  $R^2$  of 0.87, NSE of 0.86, RMSE of 0.55 mm/day and MAE of 0.4 mm/day, while Tabari 2 performed least with RMSE and NSE of 0.89 mm/day and 0.648, respectively. This result agrees with Irmak et al. (2003), Tabari et al. (2013), and Bourletsikas et al. (2017). Valle Júnior et al. (2020) evaluated 21 models and reported that 3 out of the 5-best selected  $ET_0$  were radiation-based models. The good performance of radiation-based models is likely related to the strong relationship between  $ET_0$  and Rs (Fig. 2) in the study area. All the models in this group underestimated  $ET_0$ , with the Priestly-Taylor model have the least MPE. This is consistent with the report of Celestin et al. 2020, who reported an underestimation of up to 78.6%. In Iran, Tabari et al. (2013) reported an overestimation of 18.10% by the Irmak model. Tabari 1 & 2 models use minimum and maximum temperature in place of average temperature, which improved their performance (Tabari et al. 2013). A similar result was also reported by Xystrakis & Matzarakis (2011).

#### **6.3.4.3** Temperature-based models

The temperature-based models showed acceptable and moderate results ( $R^2 > 0.5$ ), as indicated in Figure 6.4. From the RMSE and MAE values, the temperature-based models ranked third behind combination-based and radiation-based model classification. The Ravazanni model performed best in this category with NSE= 0.789, RMSE of 0.727 mm/day and MAE = 0.55 mm/day. The worst performing model in this category was the Schendel model with an NSE of 0.459 and RMSE of 1.11 mm/day. Thus, the Ravazanni model could be a promising ET<sub>0</sub> model under limited data conditions in the study area. It should be noted that the Trajkovic and Ravazanni models are variants of the Hargreaves-Samani model, which is one of the widely used ET<sub>0</sub> models. While numerous studies have calibrated the Hargreaves-Samani coefficients, Ravazanni et al. (2012) introduced a correction factor for elevation since ET<sub>0</sub> is also affected by non-weather related factors such as latitude, elevation, and topography (Baizer et al. 2019). This result agrees with numerous studies including (Razzagahi and Sepaskhah 2011, Bourletsikas et al. 2017, Peng et al.

2017, Sentalhas et al. 2010, Jabloun and Sahli 2008, Martel et al. 2018) that found the performance of the temperature-based models to be acceptable; but disagrees with Pandey et al. (2016), Shafieiyoun et al. (2020), and Valle Júnior et al. (2020). The result also showed that only the Trajkovic model underestimated  $ET_0$  by 9.8%, which agrees with Celestin et al. (2020). Djaman et al. (2015) reported a percent of error estimate (PE) of 38.81%, RMSE of 2.65 mm/day for the Schendel model, a PE of 70.18% and RMSE of 4.67 mm/day for the Trajkovic model, and PE of 69.58% and RMSE of 4.66 mm/day, for Ravazzani model at Ndiaye in Senegal.

### 6.3.4.4 Mass transfer-based models

The mass transfer-based models also showed average performance ( $R^2 > 0.5$ ) but inferior precision, as shown in the scatter plot in Figure 6.4. The mass transfer-based models performed worse based on model classification. Of the two-mass transfer-based model, the Penman model performed better. The Romanenko model performed worst, having the least NSE and highest RMSE of -1.603 and 2.44 mm/day, respectively. Both models overwhelmingly underestimated ET<sub>o</sub>, which is seen in the scatter plots of each model. The regression line is greatly deflected to the FAO-PM axis. Both models, especially the Romanenko model, underestimated ET<sub>o</sub> when ET<sub>o</sub> values were > 4 mm/day. The results disagree with the findings of Djaman et al. (2015), Bourletsikas et al. (2017), Tabari et al. (2013), and Celestin et al. (2020). Djaman et al. (2015) reported that the Romanenko model second and recommended its use when temperature and relative humidity are available. Although Tabari et al. (2013) reported worse performance of the mass transfer models, the Romanenko model performed well with R<sup>2</sup>, RMSE, and PE of 0.92, 0.66 mm/day, and 11.99%, respectively, while the Penman model yielded an R<sup>2</sup> of 0.8, RMSE of 0.81 mm/day, MAE of 0.41 and an overestimate of 17.59%.

From the above, the combination-based model classification performed best. This is due to their robust nature and the number of required data inputs. Specifically, based on the NSE and RMSE values, the performance of the fourteen empirical ET<sub>o</sub> models evaluated in this study is in the following descending order: Valiantzas 1, Valiantzas 3, Irmak, Valiantzas 2, Priestly-Taylor, Copais, Tabari 1, Ravazzani, Trajkovic, Tabari 2, Hargreaves-Samani, Schendel, Penman, and Romanenko model. However, for models with average to poor performance, calibration was used to improve model performance.





Figure 6.4: Regression analysis between empirical ET<sub>o</sub> models versus FAO PM equation

## 6.3.5 Model calibration and validation

## 6.3.5.1 Determination of Hargreaves solar radiation adjustment coefficient, Krs

A new empirical coefficient was developed for the study area using the procedure outlined in section 6.2.4. After multiple runs, the calibrated solar radiation constant is 0.146, which is 10.4% lower than the recommended 0.16 (Allen et al. 1998). This agrees with Aladenola & Madramootoo (2014) and Tabari et al. (2016). In a related study, Aladenola & Madramootoo (2014) reported a coefficient of 0.15 for Winnipeg, which is close to the study area. In Iran, Tabari et al. (2016) obtained a coefficient of 0.14 for the semi-arid climate. Under tropical climates, Adaramola (2012) and Ndulue et al. (2019) found coefficients greater than 0.19. When comparing calibrated and original Rs with measured Rs, it is observed that the statistical indices improved (Table 4), especially the MPE values, which decreased from 256.4 to 62.06%. Similarly, Tabari et al. (2016) reported a 25% reduction in RMSE.

Rs estimate	<b>R</b> <sup>2</sup>	RMSE (mm/day)	MPE (%)	MAE (mm/day)	NSE
Calibrated Krs=0.146	0.688	4.208	62.06	3.29	0.681
Original Krs=0.16	0.688	4.962	256.41	3.821	0.557
ЕТо					
Calibrated Krs	0.934	0.409	6.636	0.289	0.933
Original Krs	0.929	0.502	25.18	0.348	0.899

Table 6.4: Statistics indices using original and calibrated Krs for Rs and ETo computation

Furthermore, the FAO-PM  $ET_o$  equation was calculated using measured Rs, Rs computed using the calibrated coefficient, and Rs using the original Krs coefficient. Results showed a significant reduction in the magnitude of RMSE values when compared to the Rs computation. This can be attributed to the cumulative influence of other climatic variables. Overall, the calibrated Krs of 0.146 resulted in improved ETo estimates (Table 6.4). For example, NSE increased by 3.2% while MAE, MPE, and RMSE decreased by 16.9, 73.6, and 18.4%, respectively.

# 6.3.5.2 Determination of empirical coefficients for ETo models

Table 6.5 shows developed empirical coefficients for 7 out of the 14 evaluated  $ET_o$  models. These models were chosen because they are widely used, simple, and require fewer climate inputs. They included two temperature-based (Hargreaves Samani and Schendel), three radiation-based (Priestly-Taylor, Irmak, and Tabari), and 2 mass transfer-based models (Penman and Romanenko). It is worthy to note that the Trajvokic and Ravazzani models are variants of the Hargreaves-Samani model, while Tabari 1 and Irmak models are the same in structure and data requirements.

<b>A</b> AE
m/day)
.514
.698
.362
.354
.611
.534
.354

Table 6.5 also shows the calibrated models and their statistical performance. Using the Hargreaves Samani model, the calibrated solar radiation coefficient, temperature constant, and temperature exponent are 0.00146, 24.76, and 0.56, respectively, against the original constants of 0.0023, 17.78, and 0.5. In Texas, Awal (2020) reported constants of 0.00138, 24.49, and 0.685. Sentelhas et al. (2010) obtained coefficients ranging from 0.0017 to 0.022 for various locations in Ontario, while Djaman et al. (2018) reported coefficients ranging from 0.00202 to 0.00346 in New Mexico. Overall, the calibrated models resulted in improved model performance and better ETo estimates compared to the original coefficients (Figure 6.5). For the Hargreaves-Samani model, results showed that NSE increased from 0.7 to 0.75, RMSE decreased from 0.5 to 0.2 mm/day, MAE decreased from 0.6 to 0.3, MPE decreased from 55 to 25%. Similar results were obtained for Irmak, Schendel, Romanenko, Penman, Priestly-Taylor, and Tabari 2 models. Among the calibrated models, the greatest model improvement

was observed for Romanenko and Penman models. RMSE decreased from 2.35 to 0.7 mm/day, MPE decreased from 2.068 to 0.078, MAE decreased from 2.068 to 0.734 mm/day, and NSE increased from -1.22 to 0.803 for the Romanenko model. For the Penman model, RMSE decreased from 1.56 to 0.803 mm/day, MPE decreased from 1.363 to 0.134, MAE decreased from 1.363 to 0.611 mm/day, and NSE increased from 0.021 to 0.741.



HG-Hargreaves-Samani, PT-Priestly Taylor, IRM-Irmak, SCH-Schendel, ROM-Romanenko, TAB2-Tabari2 Figure 6.5: Statistical indices for ETo computed using original and calibrated ETo equations.

#### 6.3.6 Changes of climatic variables to changes in ETo

Figure 6.6 shows the plot of percent changes in  $ET_o$  to percent changes in each meteorological variable (Tmax, Tmin, Rs, U2, and VPD). The figure showed that changes to  $ET_o$  to climatic variables were generally linear. The results also showed that Tmax and Rs happen to cause the most significant changes to  $ET_o$ , followed by VPD, U<sub>2</sub>, and Tmin in the study area. A 25% increase in Tmax and Rs resulted in an 11.45 and 11.07% increase in  $ET_o$ . This translates to an average rise of  $ET_o$  by 0.37mm, varying between 0.02 to 0.74 mm. The exponential relationship between temperature and saturation vapor deficit (SVP) and the linear relationship between vapor pressure deficit (VPD) and ETo explains the dominating influence of temperature under semi-arid and arid climate (Irmak et al. 2006). Also, from the Clausius-Clapeyron relationship, temperature increases SVP by approximately 7%/K (Held and Soden, 2006), resulting in an increased VPD, which ultimately leads to an increased ETo.

The results also showed that  $U_2$  had a higher impact than Tmin, with the former causing a  $\pm$  4.14% variation and the latter 2.16% in response to a  $\pm$ 25% change. A similar result was found

by Darshana et al. (2013), Patle and Singh (2015), Poddar et al. 2018, Biazar et al. (2019), and Ndulue et al. (2020). In the semi-arid climate of Iran, the wind speed was reported as the most sensitive variable (Tabari and Talaee 2014, Nouri et al. 2017)



Figure 6.6: Percent change in each climatic variable to percent change in ET<sub>o</sub>

The result also confirms the dominating influence of energy-related terms in the FAO-PM equation in the study area. Thus, a small increase in Tmax has a higher impact on  $ET_0$  than a larger increase in U<sub>2</sub>. For example, a 25% increase in Tmin and U<sub>2</sub> resulted in a 2.2 and 4.1% increase in  $ET_0$ , while a 10% increase in Tmax caused  $ET_0$  to increase by 4.4%. In the arid climate of Rajasthan, India, Goyal (2004) found that a 1% increase in temperature resulted in a 15 mm increase in ET. Overall, the sensitivity of ETo to climatic variables in the study area is ranked as Tmax>Rs>VPD>U<sub>2</sub>>Tmin. Therefore, accurate measurements of temperature and solar radiation are essential for reliable  $ET_0$  estimates.

## 6.4 Conclusion

An accurate estimate of  $ET_0$  is important in determining the water demand of field crops, irrigation scheduling, agricultural planning, water usage, and effective water resources management. The FAO-PM equation is the globally accepted equation for estimating ETo. Despite its accuracy, it is

limited by requiring too many inputs not readily available because most weather stations hardly report ET<sub>0</sub>. Thus, empirical ET<sub>0</sub> equations have been developed as an alternative. In this study, ET<sub>o</sub> derived from complete data sets was compared with missing data sets. Results showed that ET<sub>o</sub> estimates under missing wind speed, relative humidity, or solar radiation were acceptable with RMSE < 0.9 and  $R^2 > 0.8$ . However, their performance decreased with an increasing number of missing data yielding average to poor performance. Furthermore, the performance of fourteen empirical ET<sub>o</sub> models was compared with the FAO PM model. Based on their NSE and RMSE values, the performance of the fourteen empirical ET<sub>o</sub> models evaluated is in the following descending order: Valiantzas 1, Valiantzas 3, Irmak, Valiantzas 2, Priestly-Taylor, Copais, Tabari 1, Ravazzani, Trajkovic, Tabari 2, Hargreaves-Samani, Schendel, Penman, and Romanenko model. New empirical coefficients (calibration) were developed for simple ETo models requiring one or two climatic inputs. Results showed that calibration resulted in improved model performance. The sensitivity analysis result showed that ETo is most sensitive to changes in Tmax and Rs, followed by VPD, U<sub>2</sub>, and Tmin. Therefore, accurate measurements of temperature and solar radiation are important for reliable ET<sub>o</sub> estimates in Winkler. This study provides simple and alternative ETo models with reasonable ETo estimates for use in southern Manitoba and other areas with similar climate characteristics. Further studies can validate the calibrated ET<sub>o</sub> models using lysimeter, water balance, or modelling.

# **Chapter Seven**

## Conclusion

# 7.1 Main findings

- The impacts of three water management techniques, controlled drainage (CD), free drainage (FD), and no drainage (ND), on canola yield and oil qualities, were evaluated over three years (2019-2021). The result shows significant differences in crop yield between the years, suggesting that yield may have been influenced by weather variables (temperature and rainfall). In 2019 with relatively reasonable rainfall amounts (May to August) and normal average temperature, CD plots had consistently higher yields across all replicates. The CD yield was significantly higher compared to FD. As the drought continued in the following years, the impact of CD diminished, showing no significant differences between the three treatments.
- 2. The results showed that the prevailing weather may have masked the impacts of drainage on canola oil qualities and soil nutrient dynamics across the soil profile. There were no significant differences between the treatments in 2020 and 2021.
- 3. The DRAINMOD model was calibrated and validated by comparing simulated and observed water table depth. Based on graphical and statistical tests, the DRAINMOD model was found adequate to simulate water table depth and predict other hydrological variables. Thus, the validated DRAINMOD model of the field was used to make sound water management decisions to achieve maximum agricultural productivity and economic and environmental sustainability.
- 4. The model was used to assess different drainage designs and determine the optimal drain spacing that would maximize the return on investment. The simulation results suggest that the long-term average yield would be maximized by close drain spacing ≤ 15 m. Economic analysis showed that 10 m drain spacing would maximize the return on investment.
- 5. The validated DRAINMOD model was run using downscaled GCMs to assess climate change effects on the hydrology of an agricultural field and canola yield under two water table management systems: free and controlled drainage. DRAINMOD simulation results suggest that subsurface drainage is expected to decrease during the mid-century while ET and surface runoff are expected to increase.

- 6. Even though CD can decrease the drainage outflow from subsurface drainage in the future, comparable canola yield did not improve due to drought stress during the growing season. The presence of heavy clay at the study area, with low hydraulic conductivity would limit water movement, which could have minimized CD impacts on crop yield.
- 7. The results presented in this study represent a wide range of options for water managers and producers to develop strategies that would mitigate heat and drought stress in the future. Water could be drained to facilitate field operations before planting. This drained water could be stored for later use to overcome the anticipated drought stress.
- 8. The ET is an essential input for most hydrologic models. Due to limited weather variables, alternative ET models with low inputs have been developed. However, these models should be tested before use in a new environment as they could affect model results in simulations to determine adequate soil and water management strategies.

## 7.2 Practical applications and contributions

This research has practical applications in Manitoba and regions with similar soil and climatic characteristics. It has also contributed to existing literature and advanced the knowledge of soil and water management in southern Manitoba in the following ways

- This research has shown that weather has a significant impact on the performance of water management systems on canola yield and quality parameters in southern Manitoba. This study showed that depending solely on rainfall would not achieve the potential or target canola yield.
- 2. The results presented in this study indicate that existing drainage systems should be reexamined in design and operation. During springtime, drainage water allowed to leave the field could be stored and reused for irrigating crops during prolonged water deficits. Farmers and producers should channel the drainage water into water conservation structures such as retention ponds and reuse the captured water for irrigation during acute development stages.
- 3. This research study has also shown that drainage contractors adopting drainage design variables from other regions without considering local factors such as soil, climate, and crop type may not be suitable to achieve maximum agronomic, economic, and

environmental gains. Long-term modelling results from this study have shown that closer draining spacing would maximize canola yield for the PESAI site with heavy clay soil.

- 4. Under climate change, rainfall is projected to decrease during the growing season, and temperature will increase in the future. From a water management perspective, using the DRAINMOD model, simulation results showed that canola yield losses due to excess water would be lower in the future. In comparison, yield losses due to drought stress would be higher. Stakeholders and water managers can use these results from future impacts of climate change on canola yields to adapt, mitigate, and develop other management strategies to cope with the effects predicted.
- 5. Limited and incomplete weather data pose a severe challenge to long-term studies needed for soil and water management in southern Manitoba. ET is one of the most critical inputs for hydrologic modelling. Researchers have used empirical ET models with fewer inputs as an alternative to the recommended FAO-PM model. Since the choice of an ET model could affect model results, the empirical ET model should first be parameterized using local data. In this study, new empirical constants applicable to southern Manitoba were developed. For example, the calibrated solar radiation constant was 0.146, rather than the recommended 0.16. Also, the calibrated empirical ET coefficients for the Hargreaves-Samani ET equations are 0.00146, 24.76 and 0.56 instead of the original coefficients: 0.0023, 17.8, and 0.5.

### 7.3 Recommendations for future studies

- 1. Future studies should improve the observed water table depth model simulation by extensive data collection, including growing and non-growing seasons. Also, for robust results, measured data for model application should include soil water content, drainage outflow, runoff, nitrate content, and phosphate
- 2. Future studies could evaluate soil water dynamics and crop yield under pulse irrigation. Canola is mainly grown in the Canadian prairies under a rainfed system. However, pulse irrigation using the drain tile could be used for irrigation under high temperature and drought conditions as experienced in the 2021 season. The prevalence of flat topography and seasonal shallow water tables in southern Manitoba provides an opportunity for this modified form of subirrigation. This method works by applying water in pulses to wet the

soil profile, and the crop ET is met through capillary flow. The conventional practice of raising the water table by subirrigation by pumping water back through the tile may lead to more water loss to deep percolation. Therefore, pulsing the water application can minimize this loss.

3. Long-term field studies should be conducted to clearly determine the impacts of subsurface drainage, especially CD on canola yield and quality parameters.

## References

- AAFC 2000– Agriculture and Agri-Food Canada. Prairie Agricultural Landscapes: Aland Resources Review. <u>https://publications.gc.ca/Collection/A98-3-4-2000E.pdf</u>. Assessed Accessed June 21, 2018
- Abbas, H. (2015). Impact of overhead irrigation on nitrogen dynamics and marketable yield of potato. Master's Thesis, Department of Biosystems Engineering University of Manitoba, Winnipeg, Canada.
- Abd-Elaty, I., Sallam, G. A., Straface, S., & Scozzari, A. (2019). Effects of climate change on the design of subsurface drainage systems in coastal aquifers in arid/semi-arid regions: Case study of the Nile delta. *Science of The Total Environment*, 672, 283-295.
- Abdullah AS. (2014). Minimum tillage and residue management increase soil water content, soil organic matter and canola seed yield and seed oil content in the semiarid areas of Northern Iraq. Soil Tillage Res.144:150–5.
- Acharya, U., Chatterjee, A., Daigh, A.L.M., (2019). Effect of subsurface drainage spacing and depth on crop yield. *Agron. J.* 111, 1675–1681.
- Adaramola M. S. (2012). Estimating global solar radiation using common meteorological data in Akure, Nigeria. Renewble Energy, 47:38–44.
- Adhikari, N., Davidson, P. C., Cooke, R. A., & Book, R. S. (2020). DRAINMOD-linked Interface for Evaluating Drainage System Response to Climate Scenarios. *Applied Engineering in Agriculture*, 36(3), 303-319.
- Ahooghalandari, M., Khiadani, M., & Jahromi, M. E. (2017). Calibration of Valiantzas' reference evapotranspiration equations for the Pilbara region, Western Australia. Theoretical and Applied Climatology, 128(3–4), 845–856.
- Aksouh-Harradj, N. M., Campbell, L. C. and Mailer, R. J. 2006. Canola response to high and moderately high temperature stresses during seed maturation. Can. J. Plant Sci. 86: 967–980.
- Aladenola, O.O. & Madramootoo, C.A. (2014). Evaluation of solar radiation estimation methods for reference evapotranspiration estimation in Canada. Theoretical Applied Climatology 118(3): 377-385.
- Alberta Agriculture and Rural Development, (2011). Crop Water Use and Requirements. Revised November 2011. Downloaded from 5485851-2011-agri-facts-crop-water-userequirements-revised-100-561-1-2011-11.pdf (Accessed September 25, 2020)
- Allerhand Joanna E., Klang, James A., P.E. Mark S. Kieser (2013). Drainage Water Management Yield Effects and Farm Profitability. Kieser & Associates
- Ale, S., Bowling, L. C., Brouder, S. M., Frankenberger, J. R., & Youssef, M. A. (2009). Simulated effect of drainage water management operational strategy on hydrology and crop yield for

Drummer soil in the Midwestern United States. *Agricultural Water Management*, 96(4), 653-665.

- Ale, S., Bowling, L. C., Brouder, S. M., Frankenberger, J. R., & Youssef, M. A. (2009). Simulated effect of drainage water management operational strategy on hydrology and crop yield for Drummer soil in the Midwestern United States. *Agricultural Water Management*, 96(4), 653-665.
- Ali Zeshan, Waheed Hina, Gul Alvina, Farzana Afzal, Kiran Anwaar, and Saiqa Imran (2017). Chapter 11 Brassicaceae plants: Metal accumulation and their role in phytoremediation in Oilseed Crops: Yield and Adaptations under Environmental Stress, First Edition. Edited by Parvaiz Ahmad.
- Alexandris, S., Kerkides, P., (2003). New empirical formula for hourly estimations of reference evapotranspiration. Agric. Water Manag. 60, 157–180. <u>https://doi.org/10.1016/S0378-3774(02)00172-5</u>.
- Alexandris, S., Stricevic, R., & Petkovic, S. (2008). Comparative analysis of reference evapotranspiration from the surface of rainfed grass in central Serbia, calculated by six empirical methods against the Penman-Monteith formula. *European Water*, 21(22), 17-28.
- Allen, R.G. (1997). A Self-Calibrating Method for Estimating Solar Radiation from Air Temperature. J. Hydrologic Engineering, ASCE 2(2):56-67.
- Allen, R. G., & Pereira, Dirk Raes, M. S. et al. (1998). Crop evapotranspiration Guidelines for computing crop water requirements - FAO Irrigation and drainage paper 56. Retrieved from <u>http://www.fao.org/docrep/X0490E/X0490E00.html</u> (Assessed 2018/09/14)
- Almazroui, M., Islam, M. N., Saeed, F., Saeed, S., Ismail, M., Ehsan, M. A., ... & Barlow, M. (2021). Projected changes in temperature and precipitation over the United States, Central America, and the Caribbean in CMIP6 GCMs. *Earth Systems and Environment*, 5(1), 1-24.
- Amatya, D. M., Fialkowski, M., & Bitner, A. (2019). A daily water table depth computing model for poorly drained soils. *Wetlands*, *39*(1), 39-54.
- Angadi, S. V., Cutforth, H. W., Miller, P. R., McConkey, B. G., Entz, M. H., Brandt, S. A., & Volkmar, K. M. (2000). Response of three Brassica species to high temperature stress during reproductive growth. Canadian Journal of Plant Science, 80, 693– 701.10.4141/P99-152
- Angadi, S. V., Cutforth, H. W., McConkey, B. G., & Gan, Y. (2004). Early seeding improves the sustainability of canola and mustard production on the Canadian semiarid prairie. *Canadian Journal of Plant Science*, 84(3), 705-711.
- Ashraf, M., & Mehmood, S. (1990). Effects of waterlogging on growth and some physiological parameters of four Brassica species. Plant and Soil, 121(2), 203–209. https://doi.org/10.1007/BF00012313
- Askar, M. H., Youssef, M. A., Chescheir, G. M., Negm, L. M., King, K. W., Hesterberg, D. L., ... & Skaggs, R. W. (2020). DRAINMOD Simulation of macropore flow at subsurface

drained agricultural fields: Model modification and field testing. *Agricultural Water Management*, 242, 106401.

- Assefa, Y.; Prasad, V.; Foster, C.; Wright, Y.; Young, S.; Bradley, P.; Stamm, M.; Ciampitti, I.A. (2018). Major management factors determining spring and winter canola yield in North America. Crop Sci., 58, 1–16.
- Awad, A., Luo, W., & Zou, J. (2021). DRAINMOD simulation of paddy field drainage strategies and adaptation to future climate change in lower reaches of the Yangtze River basin. *Irrigation and Drainage*.
- Awal, R.; Habibi, H.; Fares, A.; Deb, S. (2020). Estimating reference crop evapotranspiration under limited climate data in West Texas. J. Hydrol. Reg. Stud., 28, 100677.
- Awale, R., Chatterjee, A., Kandel, H. and Ransom, J.K. (2015) Tile Drainage and Nitrogen Fertilizer Management Influences on Nitrogen Availability, Losses, and Crop Yields. Open Journal of Soil Science, 5, 211-226.
- Ayars, J. E., Christen, E. W., & Hornbuckle, J. W. (2006). Controlled drainage for improved water management in arid regions irrigated agriculture. *Agricultural water* management, 86(1-2), 128-139.
- Azhar, A.H., Perera, B.J.C., 2011. Evaluation of reference evapotranspiration estimation methods under Southeast Australian Conditions. Journal of Irrigation and Drainage Engineering 137 (5), 268–279
- Bailey CD, Koch MA, Mayer M, Mummenhoff K, O'Kane Jr SL, et al. (2006) Toward a global phylogeny of the Brassicaceae. Mol Biol Evol 23: 2142–2160.
- Barnett, T., Malone, R., Pennell, W., Stammer, D., Semtner, B., & Washington, W. (2004). The effects of climate change on water resources in the west: introduction and overview. *Climatic Change*, 62(1), 1-11.
- Barrett-Lennard, E. G. 2003. The interaction between waterlogging and salinity in higher plants: causes, consequences and implications. Plant Soil 253: 3554
- Barthet, V.J. (2016). Canola: Overview. Canadian Grain Commission. Module in Food Science
- Barthet, V.J. (2020). Quality of western Canadian canola 2020, <u>https://www.grainscanada.gc.ca/en/grain-research/export-</u> <u>quality/oilseeds/canola/2020/pdf/Quality-western-Canadian-canola-2020.pdf</u>. Accessed 13-02-2022
- Baulch, H. M., Elliott, J. A., Cordeiro, M. R., Flaten, D. N., Lobb, D. A., & Wilson, H. F. (2019). Soil and water management: Opportunities to mitigate nutrient losses to surface waters in the northern Great Plains. *Environmental Reviews*, 27(4), 447-477.
- Bedard-Haughn, A. (2009). Managing excess water in Canadian prairie soils: A review. Can. J. Soil Sci., 89, 157–168.
- Beven, K. (2012). Rainfall-runoff modelling, Wiley. DOI:10.1002/9781119951001
- Biazar, S. M., Dinpashoh, Y. & Singh, V. P. (2019) Sensitivity analysis of the reference crop evapotranspiration in a humid region. Environmental Science and Pollution Research 26:32517–32544.

- Bilibio, C., Carvalho, J. D. A., Hensel, O., & Richter, U. (2011). Effect of different levels of water deficit on rapeseed (Brassica napus L.) crop. Ciência e Agrotecnologia, 35(4), 672– 684. https://doi.org/10.1590/S1413-70542011000400005
- BirunAra, A., Shekari, F., Hassanpouraghdam, M. B., Khorshidi, M. B., & Esfandyari, E. (2011). Effects of water deficit stress on yield, yield components and phenology of canola (Brassica napus L.) at different growth stages. Journal of Food, Agriculture and Environment, 9(3–4), 506–509.
- Blann KL, Anderson JL, Sands GR, Vondracek B. 2009. Effects of agricultural drainage on aquatic ecosystems: A review. Critical Reviews in Environmental Science and Technology, 39: 909-1001. DOI: 10.1080/10643380801977966
- Bogawski P, Bednorz E (2014) Comparison and validation of selected evapotranspiration models for conditions in Poland (Central Europe). Water Resour Manag 28(14):5021–5038
- Bonaiti, G., & Borin, M. (2010). Efficiency of controlled drainage and subirrigation in reducing nitrogen losses from agricultural fields. *Agricultural Water Management*, 98(2), 343–352. https://doi.org/10.1016/j.agwat.2010.09.008
- Bonsal, B.R., Peters, D.L., Seglenieks, F., Rivera, A., and Berg, A. (2019): Changes in freshwater availability across Canada; Chapter 6 in Canada's Changing Climate Report, (ed.) E. Bush and D.S. Lemmen; Government of Canada, Ottawa, Ontario, p. 261–342.
- Bouman, B.A.M., H. van Keulen, H.H. van Laar, and R. Rabbinge. 1996. The 'School of de Wit' crop growth simulation models: A pedigree and historical overview. *Agric. Syst.* 52:171–198.
- Bourletsikas, A., Argyrokastritis, I., & Proutsos, N. (2017). Comparative evaluation of 24 reference evapotranspiration equations applied on an evergreen-broadleaved forest. Hydrology Research, 49 (4): 1028-1041.
- Brevé, M. A., Skaggs, R. W., Parsons, J. E., & Gilliam, J. W. (1997). DRAINMOD-N, a nitrogen model for artificially drained soil. *Transactions of the ASAE*, 40(4), 1067-1075.
- Brown, L.C. and A.D. Ward. 1997. Understanding Agricultural Drainage. Food, Agricultural and Biological Engineering. Ohio State University Extension Fact Sheet. AEX-320-97. http://ohioline.osu.edu/aex-fact/0320.htm (Chapter 6).
- Bueckert A.R. and Clarke, J.M. (2013). Review: Annual crop adaptation to abiotic stress on the Canadian prairies: Six case studies. Canadian Journal of Plant Science, 2013, 93(3): 375-385 <u>https://doi.org/10.4141/cjps2012-184</u>
- Cannell, R. Q., & Belford, R. K. (1980). Effects of waterlogging at different stages of development on the growth and yield of winter oilseed rape (Brassica napus L.). *Journal of the Science of Food and Agriculture*, *31*(9), 963-965.
- CCC Canola Council of Canada 2012. Seeding based on soil temperature. https://www.canolacouncil.org/canola-watch/2012/04/25/seeding-based-on-soil-temperature/
- CCC Canola Council of Canada. 2021. Effects of soil characteristics. [Onlinehttps://www.canolacouncil.org/canola-encyclopedia/field-characteristics/effects-of-soil-

characteristics/#:~:text=Canola%20is%20considered%20moderately%20tolerant,than%2 0later%20growth%20stages%2030. (Accessed May 21, 2021)

- CCC-Canola Council of Canada. 2021. Nutrient management. <u>https://www.canolacouncil.org/canola-encyclopedia/fertility/</u>. Accessed May 21, 2018
- CCC Canola Council of Canada 2021. About Canola. <u>https://www.canolacouncil.org/about-canola/</u>. Accessed 13-08-2021
- CCC Canola Council of Canada 2021. History of canola seed development. <u>https://www.canolacouncil.org/canola-encyclopedia/history-of-canola-seed-development/</u>. Accessed 13-08-2021
- CCC Canola Council of Canada 2021. Industry overview. <u>https://www.canolacouncil.org/about-canola/industry/</u>. Accessed February 02, 2022
- CCC Canola Council of Canada 2021. Markets and statistics <u>https://www.canolacouncil.org/markets-stats/</u>. Accessed February 02, 2022
- CCC Canola Council of Canada 2021. Production. Canadian canola production statistics. https://www.canolacouncil.org/markets-stats/production/. Accessed 13-02-2022
- Cathcart R J, Topinka A K, Kharbanda P, Lange R, Yang R-C, Hall L M. 2006. Rotation length, canola variety and herbicide resistance system affect weed populations and yield. Weed Sci 54: 726-734
- Celestin, S.; Qi, F.; Li, R.; Yu, T.; Cheng, W. (2020). Evaluation of 32 Simple Equations against the Penman–Monteith Method to Estimate the Reference Evapotranspiration in the Hexi Corridor, Northwest China. *Water*, *12*, 2772.
- Chen, N., Peng, B., Hong, H., Turyaheebwa, N., Cui, S., & Mo, X. (2013). Nutrient enrichment and N: P ratio decline in a coastal bay–river system in southeast China: the need for a dual nutrient (N and P) management strategy. *Ocean & coastal management*, *81*, 7-13.
- Chaganti, V. N., Ganjegunte, G., Niu, G., Ulery, A., Enciso, J. M., Flynn, R., ... & Kiniry, J. R. (2021). Yield response of canola as a biofuel feedstock and soil quality changes under treated urban wastewater irrigation and soil amendment application. *Industrial Crops and Products*, 170, 113659.
- Champolivier, L. and Merrien, A. (1996). Effects of water stress applied at different growth stages to Brassica napus L. var. oleifera on yield, yield components and seed quality. Eur. J. Agron., 5: 153-160.
- Chartzoulakis, K., & Bertaki, M. (2015). Sustainable water management in agriculture under climate change. *Agriculture and Agricultural Science Procedia*, *4*, 88-98.
- Chow, V. T., Maidment, D. R., & Mays, L. W. (1988). *Applied hydrology*. New York: McGraw-Hill.
- Clayton, G. W., Harker, K. N., O'Donovan J. T., Blackshaw, R. E., Dosdall, L. M., Stevenson, F. C. and Ferguson, T. 2004. Fall and spring seeding date effects on herbicide-tolerant canola (Brassica napus L.) cultivars. Can. J. Plant Sci. 84: 419–430.

- Cliff Jamieson 2021. Canada Markets. Canada's Canola and the Global Trade https://www.dtnpf.com/agriculture/web/ag/blogs/canada-markets/blogpost/2021/07/14/canadas-canola-globaltrade#:~:text=The%20brown%20line%20with%20markers,global%20exports%20of%20 16.099%20mmt.
- Climatedata.ca. <u>https://climatedata.ca/explore/location/?loc=GAAYG&location-select-temperature=tx\_max&location-select-precipitation=r1mm&location-select-other=frost\_days</u>
- C'ordova, M., Carrillo-Rojas, G., Crespo, P., Wilcox, B., and Célleri, R. (2015). Evaluation of the Penman-Monteith (FAO 56 PM) method for calculating reference evapotranspiration using limited data. Application to the Wet Páramo of Southern Ecuador. Mt. Res. Dev., 35(3), 230–239.
- Colwell H.T.M. 1978. The economics of increasing crop productivity in Ontario and Quebec by tile drainage installation. Canadian Farm Economics. 13(3):1-7.
- Confortin, T.C. Todero, I., Luft, L. Ugalde, G.A., Mazutti, M.A., Oliveira, Z.B., Bottega, E.L., Knies, A.E., Zabot, G.L., Tres, M.V., (2019). Oil yields, protein contents, and cost of manufacturing of oil obtained from different hybrids and sowing dates of canola, J. Environ. Chem. Eng. 7.
- Cooke, R., Nehmelman, J., & Kalita, P. (2002). Effect of tile depth on nitrate transport from tile drainage systems, ASAE Paper number 022017. 2002 ASAE Annual International Meeting / CIGR XVth World Congress, 0300(02).
- Cooke, R., & Verma, S. (2012). Performance of drainage water management systems in Illinois, United States. *Journal of Soil and Water Conservation*, 67(6), 453-464.
- Cordeiro, M.R.C. 2014. Agronomic and environmental impacts of corn production under different management strategies in the Canadian prairies. PhD Thesis. University of Manitoba, Canada.
- Cordeiro, M., & Ranjan, R. S. (2015). DRAINMOD Simulation of Corn Yield under Different Tile Drain Spacing in the Canadian Prairies. *Transactions of the ASABE*, 58(6), 1481–1491. https://doi.org/10.1177/107110070502600211
- Cordeiro, M. R., Rotz, A., Kroebel, R., Beauchemin, K. A., Hunt, D., Bittman, S., ... & McKenzie, D. B. (2019). Prospects of Forage Production in northern regions under climate and land-use changes: A case-study of a dairy farm in Newfoundland, Canada. *Agronomy*, 9(1), 31.
- Cosgrove, W. J., & Loucks, D. P. (2015). Water management: Current and future challenges and research directions. Water Resources Research, 4823–4839.
- Crabbé, P., Lapen, D. R., Clark, H., Sunohara, M., & Liu, Y. (2012). Economic benefits of controlled tile drainage: Watershed Evaluation of Beneficial Management Practices,

South Nation River basin, Ontario. Water Quality Research Journal, 47(1), 30–41. https://doi.org/10.2166/wqrjc.2012.007

- Cutforth, H.W., S.V. Angadi, B.G. McConkey, P.R. Miller, D. Ulrich, R. Gulden, K.M. Volkmar, M.H. Entz, and S.A. Brandt. 2013. Comparing rooting characteristics and soil water withdrawal patterns of wheat with alternative oilseed and pulse crops grown in the semiarid Canadian prairie. Can. J. Soil Sci. 93:147–160.
- Darshana, D., Pandey, A. & Pandey, R. P. (2013). Analysing trends in reference evapotranspiration and weather variables in the Tons River Basin in Central India. Stochastic Environmental Research and Risk Assessment 27, 1407–1421.
- Darzi, A., Ejlali, F., Ahmadi. M. Z., Najafi, Gh. H. 2007. The suitability of controlled drainage and subirrigation in paddy fields. Pakistan journal of Biological Science10 (3):492-497
- Daun, J. K. (2011). Chapter 1: Origin, Distribution, and Production AOCS press (J. K. Daun, N. A. M. Eskin, & D. Hickling, Eds.).
- Davidson, N. C. (2014). How much wetland has the world lost? Long-term and recent trends in global wetland area. *Marine and Freshwater Research*, 65(10), 934-941.
- Davoodi, K., Darzi-Naftchali, A., & Aghajani-Mazandarani, G. (2019). Evaluating DRAINMOD-S to Predict Drainage Water Salinity and Groundwater Table Depth During Winter Cropping in Heavy-Textured Paddy Soils. *Irrigation and Drainage*, 68(3), 559-572.
- Dayyani, S., Madramootoo, C. A., Prasher, S. O., Madani, A., & Enright, P. (2010). Modeling water table depth, drain outflow, and nitrogen losses in a cold climate using DRAINMOD 5.1. *Transactions of the ASABE*, 53(2), 385-395.
- Dayyani, S., Prasher, S. O., Madani, A., & Madramootoo, C. A. (2012). Impact of climate change on the hydrology and nitrogen pollution in a tile-drained agricultural watershed in eastern Canada. *Transactions of the ASABE*, *55*(2), 389-401.
- Djaman K, Balde AB, Sow A, Muller B, Irmak S, N'Diaye MK, Manneh B, Moukoumbi YD, Futakuchi K, Saito K (2015). Evaluation of sixteen reference evapotranspiration methods under sahelian conditions in the Senegal River Valley. J Hydrology: Regional Studies 3:139–159
- Djaman, K., O'Neill, M., Diop, L. (2018). Evaluation of the Penman-Monteith and other 34 reference evapotranspiration equations under limited data in a semiarid dry climate. Theoretical and Applied Climatology, 133, 1-9.
- Dogan, E., Copur, O., Kahraman, A., Kirnak, H., & Guldur, M. E. (2011). Supplemental irrigation effect on canola yield components under semiarid climatic conditions. Agricultural Water Management, 98(9), 1403–1408.
- Dou, X., Shi, H., Li, R., Miao, Q., Tian, F., Yu, D., ... & Wang, B. (2021). Effects of Controlled Drainage on the Content Change and Migration of Moisture, Nutrients, and Salts in Soil and the Yield of Oilseed Sunflower in the Hetao Irrigation District. *Sustainability*, 13(17), 9835.
- Dreccer, M. F., Fainges, J., Whish, J., Ogbonnaya, F. C., & Sadras, V. O. (2018). Comparison of sensitive stages of wheat, barley, canola, chickpea and field pea to temperature and water

stress across Australia. Agricultural and Forest Meteorology, 275–294. https://doi.org/10.1016/j.agrformet.2017.10.006

- Drew, M.C. 1997. Oxygen deficiency and root metabolism: Injury and acclimation under hypoxia and anoxia. Annual Review of Plant Physiology and Plant Molecular Biology 48:223-250.
- Dring, C., Devlin, J. F., Boag, G., Sunohara, M. D., Fitzgibbon, J., Topp, E., & Lapen, D. R. (2016). Incentives and disincentives identified by producers and drainage contractors/experts on the adoption of controlled tile drainage in eastern Ontario, Canada. *Water Quality Research Journal of Canada*, 51(1), 1-16.
- Drury, C.F., C.S. Tan, W.D. Reynolds, T.W. Welacky, T.O. Oloya, and J.D. Gaynor. 2009. Managing tile drainage, subirrigation, and nitrogen fertilization to enhance crop yields and reduce nitrate loss. Journal of Environment Quality 38(3): 1193-1204.
- Dumanski, S., Pomeroy, J. W., & Westbrook, C. J. (2015). Hydrological regime changes in a Canadian Prairie basin. Hydrological Processes, 29(18), 3893-3904
- Duran-Encalada, J. A., Paucar-Caceres, A., Bandala, E. R., & Wright, G. H. (2017). The impact of global climate change on water quantity and quality: A system dynamics approach to the US–Mexican transborder region. European Journal of Operational Research, 256(2), 567–581.
- Ebrahimian, E., Bybordi, A., & Seyyedi, S. M. (2017). How Nitrogen and Zinc Levels Affect Seed Yield, Quality, and Nutrient Uptake of Canola Irrigated with Saline and Ultra-Saline Water. Communications in Soil Science and Plant Analysis, 48(3), 345–355. https://doi.org/10.1080/00103624.2016.1269804
- Elferjani, R., & Soolanayakanahally, R. (2018). Canola Responses to Drought, Heat, and Combined Stress: Shared and Specific Effects on Carbon Assimilation, Seed Yield, and Oil Composition. Frontiers in plant science, 9, 1224. doi:10.3389/fpls.2018.01224
- Elmi, A. A., Madramootoo, C., & Hamel, C. (2000). Influence of water table and nitrogen management on residual soil NO<sub>3</sub><sup>-</sup> and denitrification rate under corn production in sandy loam soil in Quebec. Agriculture, Ecosystems and Environment, 79(2–3), 187–197. <u>https://doi.org/10.1016/S0167-8809(99)00157-7</u>
- ECCC- Environment and Climate Change Canada 2013. Water in Canada. https://www.canada.ca/en/environment-climate-change/services/wateroverview/publications/water-in-canada.html
- Environment Canada, 2019. 1981-201 climate normal and averages <u>https://climate.weather.gc.ca/climate\_normals/index\_e.html</u>
- Environment Canada, 2019. 1981-201 climate normal and averages <u>https://climate.weather.gc.ca/climate\_normals/index\_e.html</u>

Environment canada 2020

(https://climate.weather.gc.ca/climate\_normals/results\_1981\_2010\_e.html?stnID=3626& autofwd=1). (Accessed October 5, 2020)

Environment Canada. 2021. Canadian Climate Normals 1981-2010. National Climate Data and Information Archive. Arborg: Environment Canada. Available at: <u>https://climate.weather.gc.ca/climate\_normals/results\_1981\_2010\_e.html?searchType=st\_nName&txtStationName=arborg&searchMethod=contains&txtCentralLatMin=0&txtCen</u> tralLatSec=0&txtCentralLongMin=0&txtCentralLongSec=0&stnID=3717&dispBack=1.
Accessed 20 August 2021.

- Evans, R. O., Gilliam, J. W., and Skaggs, R. W. (1989). "Managing water table management systems for water quality." ASAE Paper No. 89-2129, American So- ciety of Agricultural Engineers, St. Joseph, Mich.
- Evans RO, and Skagss RW (1991) Stress Day index models to predict corn and soybean relative yield under high water table conditions. Trans ASAE 5, 1997–2005. doi:0001-2351/91/3405-1997
- Evans, R.O., Skaggs, R.W., Gilliam, J.W., 1995. Controlled versus conventional drainage effects on water quality. J. Irrig. Drain. Eng. 121, 271–276.
- Evans, R. O. and Fausey, N. R. 1999. Effects of inadequate drainage on crop growth and yield. Pages 13-54 in R. W. Skaggs and J. van Schilfgaarde, eds. Agricultural Drainage. ASA-CSSA-SSSA, Madison, WI.
- Faraji, A., N. Latifi, A. Soltani, and A.H.S. Rad. 2009. Seed yield and water use efficiency of canola (Brassica napus L.) as affected by high temperature stress and supplemental irrigation. Agric. Water Manage. 96:132–140. doi:10.1016/j. agwat. 2008. 07. 014
- Fausey, N.R. 2005. Drainage management for humid regions. Int. Agric. Eng. J. 14(4): 209–214.
- FCC Farm Credit Canada 2020. TRADE RANKINGS 2020 Opportunities and challenges to diversify Canada's food exports. Accessed 13-02-2022. https://www.fccfac.ca/fcc/resources/trade-rankings-report-2020-e.pdf
- Fayrap, A., & Koc, C. (2012). Comparison of drainage water quality and soil salinity in irrigated areas with surface and subsurface drainage systems. *Agricultural Research*, 1, 280–284.
- Fore SR, Porter P, Lazarus W (2011) Net energy balance of small-scale on-farm biodiesel production from canola and soybean. Biomass and Bioenergy 35(5): 2234–2244
- Francois, L.E. (1994). Growth, Seed Yield, and Oil Content of Canola Grown under Saline Conditions.
- Franklin, J. A., Kav, N. N. V, Yajima, W., & Reid, D. M. (2005). Root temperature and aeration effects on the protein profile of canola leaves. Crop Science, 45(4), 1379–1386. https://doi.org/10.2135/cropsci2004.0352
- Fraser, H., Fleming, R., & Eng, P. (2001). Environmental benefits of tile drainage. Prepared for: LICO. Land Improvement Contractors of Ontario. Ridgetown College, University of Guelph.
- French, R. J., Seymour, M., & Malik, R. S. (2016). Plant density response and optimum crop densities for canola (Brassica napus L.) in Western Australia. Crop and Pasture Science, 67(4), 397. <u>https://doi.org/10.1071/CP15373</u>
- Gaj, N., & Madramootoo, C. A. (2017). Long-term simulations of the hydrology for sugarcane fields in the humid tropics: case study on Guyana's coastland. *Journal of Irrigation and Drainage Engineering*, 143(8), 05017002.

- Gan, Y., Angadi, S. V., Cutforth, H., Potts, D., Angadi, V. V. and McDonald, C. L. 2004. Canola and mustard response to short periods of temperature and water stress at different developmental stages. Can. J. Plant Sci. 84: 697–704.
- Gan, Y., Malhi, S.S., Brandt, S., Katepa-Mupondwa, F., and Kutcher, H.R. 2007. Brassica juncea canola in the northern Great Plains: responses to diverse environments and nitrogen fertilization. Agron. J 99: 1208–1218. doi:10.2134/agronj2006.0296.
- Gan, Y., Harker, K. N., Kutcher, H. R., Gulden, R. H., Irvine, B., May, W. E., & O'Donovan, J. T. (2016). Canola seed yield and phenological responses to plant density. Canadian Journal of Plant Science, 96(1), 151–159. https://doi.org/10.1139/cjps-2015-0093
- Gao, Z., He, J., Dong, K., Bian, X. & Li, X. (2016). Sensitivity study of reference crop evapotranspiration during growing season in the West Liao River basin, China. Theoretical and Applied Climatology 124 (3–4), 865–881.
- Gao, F.; Feng, G.; Ouyang, Y.; Wang, H.; Fisher, D.; Adeli, A.; Jenkins, J. 2017. Evaluation of Reference Evapotranspiration Methods in Arid, Semiarid, and Humid Regions. J. Am. Water Resour. Assoc., 53, 791–808, doi:10.1111/1752-1688.12530.
- Gentry, L. E., M. B. David, T. V. Royer, C. A. Mitchell, and K. M. Starks. 2007. Phosphorus Transport Pathways to Streams in Tile-Drained Agricultural Watersheds. J. Environ. Qual. 36:408-415. doi:10.2134/jeq2006.0098
- Gibbs, J., and Greenway, H. (2003). Mechanisms of anoxia tolerance in plants. I. Growth, survival and anaerobic catabolism. *Funct. Plant Biol.* 30, 1–47. doi: 10.1071/PP98095
- Grigg, B. C., Southwick, L. M., Fouss, J. L., & Kornecki, T. S. (2003). drainage system impacts on surface runoff, nitrate loss, and crop yield on a southern alluvial soil. *ASABE 46:*1531–1537.
- Ghane, E., Askar, M. H., & Skaggs, R. W. (2021). Design drainage rates to optimize crop production for subsurface-drained fields. *Agricultural Water Management*, 257, 107045.
- Ghobadi M., Bakhshandeh M., Fathi G., Gharineh M.H., Alami-Said K., Naderi A, and Ghobadi, M.E. (2006). Short and Long Periods of Water Stress During Different Growth Stages of Canola (Brassica napus L.): Effect on Yield, Yield Components, Seed Oil and Protein Contents. Journal of Agronomy, 5: 336-341.
- Gramlich, A., Stoll, S., Stamm, C., Walter, T., & Prasuhn, V. (2018). Effects of artificial land drainage on hydrology, nutrient and pesticide fluxes from agricultural fields A review, 266: 84–99. https://doi.org/10.1016/j.agee.2018.04.005
- Gelderman, R.H., and D. Beegle. 2015. Nitrate-nitrogen. In: Nathan, M.V., and R.H. Gelderman, editors, Recommended Chemical Soil Testing Procedures for the North Central Region. North Central Reg. Res. Publ. no. 221 (revised). Missouri Agric. Exp. Stn. SB 1001. Columbia, MO. p. 5.1-5.4.
- Golmohammadi, G., Rudra, R. P., Prasher, S. O., Madani, A., Goel, P. K., & Mohammadi, K. (2016). Modeling the impacts of tillage practices on water table depth, drain outflow and

nitrogen losses using DRAINMOD. *Computers and Electronics in Agriculture*, 124, 73–83. <u>https://doi.org/10.1016/j.compag.2016.03.031</u>

- Golmohammadi, G., Rudra, R. P., Parkin, G. W., Kulasekera, P. B., Macrae, M., & Goel, P. K. (2021). Assessment of Impacts of Climate Change on Tile Discharge and Nitrogen Yield Using the DRAINMOD Model. *Hydrology*, 8(1), 1.
- Goyal, R. K. (2004). Sensitivity of evapotranspiration to global warming: a case study of arid zone of Rajasthan (India). Agricultural Water Management 69,1–11.
- Grenon, G., Singh, B., De Sena, A., Madramootoo, C. A., von Sperber, C., Goyal, M. K., & Zhang, T. (2021). Phosphorus fate, transport and management on subsurface drained agricultural organic soils: A review. *Environmental Research Letters*, *16*(1), 013004.
- Gunn, K. M., Fausey, N. R., Shang, Y., Shedekar, V. S., Ghane, E., Wahl, M. D., & Brown, L. C. (2015). Subsurface drainage volume reduction with drainage water management: Case studies in Ohio, USA. Agric. Water Mgmt., 149, 131-142. https://doi.org/10.1016/j.agwat.2014.10.014
- Gunn, K. M., Baule, W. J., Frankenberger, J. R., Gamble, D. L., Allred, B. J., Andresen, J. A., & Brown, L. C. (2018). Modeled climate change impacts on subirrigated maize relative yield in northwest Ohio. *Agricultural Water Management*, 206, 56-66.
- Haghnazari, F., Karandish, F., Darzi-Naftchali, A., & Šimůnek, J. (2020). Dynamic assessment of the impacts of global warming on nitrate losses from a subsurface-drained rainfed-canola field. *Agricultural water management*, 242, 106420.
- Hammac W. Ashley, Maaz Tai M., Koenig Richard T., Burke Ian C., and William L. Pan (2017). Water and Temperature Stresses Impact Canola (Brassica napus L.) Fatty Acid, Protein, and Yield over Nitrogen and Sulfur. Journal of Agricultural and Food Chemistry 65 (48), 10429-10438. DOI: 10.1021/acs.jafc.7b02778
- Hang, A.N., H.P. Collins, and K.E. Sowers. 2009. Irrigated Spring and Winter Canola Production in Washington. Prosser, WA. WSU Extension Manual EM006E.
- Hanke K (2018). Impacts of Climate Change and Controlled Tile Drainage on Water Quality and Quantity in Southern Ontario, Canada. UWSpace. <u>http://hdl.handle.net/10012/13757</u>
- Hargreaves, G.H., Samani, Z.A., (1985). Reference crop evapotranspiration from temperature. Applied Engineering in Agriculture 1 (2), 96–99.
- Harker, K. N., O'Donovan, J. T., Blackshaw, R. E., Johnson, E. N., Lafond, G. P., & May, W. E. (2012). Seeding depth and seeding speed effects on no-till canola emergence, maturity, yield and seed quality. Canadian Journal of Plant Science, 92(4), 795–802. <u>https://doi.org/10.4141/cjps2011-189</u>
- Hassanpour, B., Parsinejad, M., Yazdani, M.R., Salahshour Dalivand, F. and Kossari, H. (2011). Evaluation of modified DRAINMOD in predicating groundwater table fluctuations and yield of canola in paddy fields under snowy conditions (case study: RASHT, IRAN). *Irrigation and Drainage*, 60, 660–667

- Hashemi, S. Z., Darzi-Naftchali, A., & Qi, Z. (2020). Assessing water and nitrate-N losses from subsurface-drained paddy lands by DRAINMOD-N II. *Irrigation and Drainage*, 69(4), 776-787.
- Hawley, N., T.H. Johengen, Y.R. Rao, S.A. Ruberg, D. Beletsky, S.A. Ludsin, B.J. Eadie, D.J. Schwab, T.E. Croley, and S.B. Brandt. 2006. Lake Erie hypoxia prompts Canada-U.S. study. EOS Trans. Am. Geophys. Union 87:313–319.
- Held, I. M. & Soden, B. J. (2006). Robust responses of the hydrological cycle to global warming. J. Clim. 19, 5686–5699.
- Helmers, M., Christianson, R., Brenneman, G., Lockett, D., & Pederson, C. (2012). Water table, drainage, and yield response to drainage water management in southeast Iowa. *Journal of soil and Water Conservation*, 67(6), 495-501.
- Helmers, Matt; Pederson, Carl; Craft, Kristina; and Schott, Linda (2017) "Impact of Drainage Water Management on Crop Yield, Drainage Volume, and Nitrate Loss," Farm Progress Reports: Vol. 2016 : Iss. 1, Article 144. DOI: https://doi.org/10.31274/farmprogressreports-180814-1708
- Helmers, M. J., Abendroth, L., Reinhart, B., Chighladze, G., Pease, L., Bowling, L., ... & Strock, J. (2022). Impact of controlled drainage on subsurface drain flow and nitrate load: A synthesis of studies across the US Midwest and Southeast. *Agricultural Water Management*, 259, 107265.
- Hergert G.W., Margheim G.F., Pavlista A.D., Martin D.L., Supalla R.J., Isbell T.A. (2016). Yield, irrigation response, and water productivity of deficit to fully irrigated spring canola. Agric. Water Manage., 168:96-103
- Hossain, Z., Johnson, E. N., Wang, L., Blackshaw, R. E., & Gan, Y. (2019). Comparative analysis of oil and protein content and yield of five Brassicaceae oilseeds on the Canadian prairie. *Industrial Crops and Products*, *136*, 77-86.
- Hornbuckle, J. W., Christen, E. W., Ayars, J. E., & Faulkner, R. D. (2005). Controlled water table management as a strategy for reducing salt loads from subsurface drainage under perennial agriculture in semi-arid Australia. *Irrigation and Drainage Systems*, 19(2), 145– 159. https://doi.org/10.1007/s10795-005-4420-0
- Hu, W., Schoenau, J. J., Cutforth, H. W., & Si, B. C. (2015). Effects of row-spacing and stubble height on soil water content and water use by canola and wheat in the dry prairie region of Canada. Agricultural Water Management, 153, 77–85. https://doi.org/10.1016/J.AGWAT.2015.02.008
- Huang, B. 2000. Waterlogging responses and interaction with temperature, salinity, and nutrients. Pages 263-283 in R. E. Wilkinson, ed. Plant-environment interactions. Marcel Dekker, New York, NY.
- Hudson, J.A., & Roberts, G.A. (1982). The effect of a tile drain on the soil moisture content of peat.
- Huffman, R. L., Fangmeier, D. D., Elliot, W. J., Workman, S. R., & Schwab, G. O. (2011). Soil and Water Conservation Engineering, Fifth Edition (7th ed.). ASABE.

- IPCC, 2013. Climate change 2013: the physical science basis. Working Group I Contribution to the IPCC Fifth Assessment Report. Cambridge, United Kingdom.
- Irmak S, Irmak A, Allen RG, Jones JW (2003) Solar and net radiation-based equations to estimate reference evapotranspiration in humid climates. J Irrig Drain Eng ASCE 129(5):336–347
- Irmak, S., Payero, J. O., Martin, D. L., Irmak, A. & Howell, T. A.(2006). Sensitivity analyses and sensitivity coefficients of standardized daily ASCE-Penman–Monteith equation. Journal of Irrigation and Drainage Engineering 132, 564–578.
- Irwin, R. (1989). Land Drainage Technology Canada's Leadership Role. Scientia Canadensis, 13(2), 102–109. https:// doi.org/10.7202/800288ar
- Ismail, S. M. (2016). Maximizing Production and Irrigation Water Productivity of Canola Crop (Brassica Napus L.) Under Arid Land Conditions. Irrigation and Drainage.
- Jabloun, M., & Sahli, A. (2008). Evaluation of FAO-56 methodology for estimating reference evapotranspiration using limited climatic data. Application to Tunisia. Agricultural Water Management, 95(6), 707–715
- Jackson, M. B. and Drew, M. C. 1984. Effects of flooding on growth and metabolism of herbaceous plants. Pages 47-128 in T. T. Kozlowski, ed. Flooding and plant growth. Academic Press, Orlando, FL
- Jackson, M. B. (2004). *The Impact of Flooding Stress on Plants and Crops*. Available at: <u>http://www.plantstress.com/Articles/waterlogging\_i/waterlog\_i.htm</u>
- Jackson, E. K., Roberts, W., Nelsen, B., Williams, G. P., Nelson, E. J., and Ames, D. P. (2019). Introductory overview: error metrics for hydrologic modelling–A review of common practices and an open source library to facilitate use and adoption. Environ. Modell. Softw. 119, 32–48. doi: 10.1016/j.envsoft.2019.05.001
- Jafari-Talukolaee, M., H. Ritzema, A. Darzi-Naftchali and A. Shahnazari. 2016. Subsurface drainage to enable the cultivation of winter crops in consolidated paddy fields in northern Iran. Sustainability 8(3): 249.
- Jafari Talukolaee, M., Darzi Naftchali, A., Zare Parvariji, L., & Ahmadi, M. Z. (2018). Investigating long-term effects of subsurface drainage on soil structure in paddy fields. *Soil and Tillage Research*, 177(May 2017), 155–160. https://doi.org/10.1016/j.still.2017.12.012
- Jalota, S. K., Vashisht, B. B., Sharma, S., & Kaur, S. (2018). Climate Change Impact on Crop Productivity and Field Water Balance. In Understanding Climate Change Impacts on Crop Productivity and Water Balance (pp. 87–148). Elsevier. <u>https://doi.org/10.1016/b978-0-12-809520-1.00003-3</u>
- Jerszurki, D., de Souza, J. L. M., & Silva, L. D. C. R. (2019). Sensitivity of ASCE-Penman– Monteith reference evapotranspiration under different climate types in Brazil. *Climate Dynamics*, 53(1), 943-956.
- Jiang, Q., Qi, Z., Xue, L., Bukovsky, M., Madramootoo, C. A., & Smith, W. (2020). Assessing climate change impacts on greenhouse gas emissions, N losses in drainage and crop production in a subsurface drained field. *Science of the Total Environment*, 705, 135969.

- Jin, C.- X. and G. R. Sands, 2003. The long-term field-scale hydrology of subsurface drainage systems in a cold climate. Trans. *ASAE* 46(4):1011-1021.
- Jouni, H. J., Liaghat, A., Hassanoghli, A., & Henk, R. (2018). Managing controlled drainage in irrigated farmers' fields: A case study in the Moghan plain, Iran. Agricultural water management, 208, 393-405.
- Kashyap, D., & Agarwal, T. (2020). Climate Change, Water Resources, and Agriculture: Impacts and Adaptation Measures. In *Global Climate Change and Environmental Policy* (pp. 227-256). Springer, Singapore.
- Kamkar, B., Daneshmand, A. R., Ghooshchi, F., Shiranirad, A. H., & Langeroudi, A. S. (2011). The effects of irrigation regimes and nitrogen rates on some agronomic traits of canola under a semiarid environment. *Agricultural Water Management*, 98(6), 1005-1012.
- Kandil, H. M., Skaggs, R. W., Dayem, S. A., & Aiad, Y. (1995). DRAINMOD-S: Water management model for irrigated arid lands, crop yield and applications. *Irrigation and Drainage Systems*, 9(3), 239-258.
- Kang, Y., Khan, S., & Ma, X. (2009). Climate change impacts on crop yield, crop water productivity and food security - A review. In *Progress in Natural Science* (Vol. 19, Issue 12, pp. 1665–1674). Science Press. https://doi.org/10.1016/j.pnsc.2009.08.001
- Karandish, F., & Šimůnek, J. (2016). A comparison of numerical and machine-learning modeling of soil water content with limited input data. *Journal of Hydrology*, 543, 892–909. https://doi.org/10.1016/j.jhydrol.2016.11.007
- King, K. W., Williams, M. R., Macrae, M. L., Fausey, N. R., Frankenberger, J., Smith, D. R., ... & Brown, L. C. (2015). Phosphorus transport in agricultural subsurface drainage: A review. *Journal of environmental quality*, 44(2), 467-485.
- Khalatbari, A., Shirani Rad, A., Valadabady, S. A., Sayfzadeh, S., & Zakerin, H. (2021). Yield Components and Fatty Acids Variation of Canola Cultivars Under Different Irrigation Regimes and Planting Dates. *Gesunde Pflanzen*, 1-11.
- Kirkegaard J. A., Lilley J. M., Brill R. D., Sprague S. J., Fettell N. A., Pengilley G. C. (2016). Re-evaluating sowing time of spring canola (Brassica napus L.) in south-eastern Australia—how early is too early? Crop and Pasture Science 67, 381-396.
- Kirkland, K. J. and Johnson, E. N. 2000. Alternative seeding dates (fall and April) affect Brassica napus canola yield and quality. Can. J. Plant Sci. 80: 713–719
- Keam, D. (2011). The sustainability of municipal wastewater irrigation in the Interlake Region of Manitoba as a means of nitrogen and phosphorous abatement for Lake Winnipeg.
- Kladivko, E. J., Frankenberger, J. R., Jaynes, D. B., Meek, D. W., Jenkinson, B. J., & Fausey, N. R. (2004). Nitrate leaching to subsurface drains as affected by drain spacing and changes in crop production system. *Journal of environmental quality*, *33*(5), 1803-1813.

- Kokulan, V., Macrae, M. L., Lobb, D. A., & Ali, G. A. (2019). Contribution of overland and tile flow to runoff and nutrient losses from Vertisols in Manitoba, Canada. *Journal of environmental quality*, 48(4), 959-965.
- Koocheki, A.; Nassiri Mahallati, M.; Moradi, R.; Mansoori, H. .2014. Optimizing water, nitrogen and crop density in canola cultivation using response surface methodology and central composite design. Soil Sci. Plant Nutr, 60, 286–298.
- Koudahe K, Djaman K, Adewumi JK (2018) Evaluation of the Penman-Monteith reference evapotranspiration under limited data and its sensitivity to key climatic variables under humid and semiarid conditions. Model Earth Syst Environ 4:1239–1257
- Kresovic, B., Matovic, G., Gregoric, E., Djuricin, S., & Bodroza, D. (2014). Irrigation as a climate change impact mitigation measure: An agronomic and economic assessment of maize production in Serbia. *Agricultural Water Management*, *139*, 7-16.
- Kross, A., Lapen, D. R., McNairn, H., Sunohara, M., Champagne, C., & Wilkes, G. (2015). Satellite and in situ derived corn and soybean biomass and leaf area index: Response to controlled tile drainage under varying weather conditions. *Agricultural Water Management*, 160, 118-131.
- Kumar S, Nakajima T, Kadono A, Lal R, Fausey N (2014) Long-term tillage and drainage influences on greenhouse gas fluxes from a poorly drained soil of central Ohio. J Soil Water Conserv 69(6):553–563
- Kunz, N. C., & Moran, C. J. (2016). The utility of a systems approach for managing strategic water risks at a mine site level. *Water Resources and Industry*, *13*, 1–6. <u>https://doi.org/10.1016/J.WRI.2016.02.001</u>
- Kurki-Fox, J. J., Burchell, M. R., & Kamrath, B. J. (2019). The Potential Long-Term Impacts of Climate Change on the Hydrologic Regimes of North Carolina's Coastal Plain Non-Riverine Wetlands. *Transactions of the ASABE*, 62(6), 1591-1606.
- Lal R. (2005) Climate change, soil carbon dynamics, and global food security. In: Lal R, Stewart B, Uphoff N, et al., editors. Climate change and global food security. Boca Raton (FL): CRC Press; p. 113–43.
- Lancashire, P. D., Bleiholder, H., Boom, T. V. D., Langelüddeke, P., Stauss, R., Weber, E., & Witzenberger, A. (1991). A uniform decimal code for growth stages of crops and weeds. *Annals of applied Biology*, 119(3), 561-601.
- Lauretti, E., & Praticò, D. (2017). Effect of canola oil consumption on memory, synapse and neuropathology in the triple transgenic mouse model of Alzheimer's disease. Scientific reports, 7(1), 17134. doi:10.1038/s41598-017-17373-3
- Li, G., Zhang, X., Cannon, A. J., Murdock, T., Sobie, S., Zwiers, F., Anderson, K., Qian, B. (2018). Indices of Canada's future climate for general and agricultural adaptation applications. Climatic Change, 148(1–2), 249–263. <u>https://doi.org/10.1007/s10584-018-2199-x</u>
- Lin, L., Allemekinders, H., Dansby, A., Campbell, L., Durance-Tod, S., Berger, A., & Jones, P. J. (2013). Evidence of health benefits of canola oil. Nutrition reviews, 71(6), 370-85.
- Liu, J., Elliott, J. A., Wilson, H. F., Macrae, M. L., Baulch, H. M., & Lobb, D. A. (2021). Phosphorus runoff from Canadian agricultural land: A cross-region synthesis of edge-offield results. *Agricultural Water Management*, 255, 107030.
- Lohani, N., Singh, M. B., & Bhalla, P. L. (2021). Short-term heat stress during flowering results in a decline in Canola seed productivity. *Journal of Agronomy and Crop Science*.
- López-Urrea, R., Martín de Santa Olalla, F., Fabeiro, C., & Moratalla, A. (2006). Testing evapotranspiration equations using lysimeter observations in a semiarid climate. Agricultural Water Management, 85(1–2), 15–26
- Luo W, Skaggs RW, Chescheir GM. 2000. DRAINMOD modifications for cold conditions. Transactions of the ASAE 43(6): 1569-1582
- Luo W., Sands G.R., Youssef M.A., Strock J.S., Song I., Canelon D. 2010. Modeling the impact of alternative drainage practices in the northern Corn-Belt with DRAINMOD-NII. Agricultural Water Management 97(3):389–398.
- LMC International 2020. The Economic Impact of Canola on the Canadian Economy: 2020 Update Report for: Canola Council of Canada Winnipeg, Canada. Accessed on 8<sup>th</sup> March 2021 <u>https://www.canolacouncil.org/download/131/economic-impact/17818/economicimpact-report-canada\_december-2020</u>
- Maaz, T., Pan, W., & Hammac, W. (2016). Influence of soil nitrogen and water supply on canola nitrogen use efficiency. *Agronomy Journal*, *108*(5), 2099-2109.
- Madani, A., & Brenton, P. (1995). Effect of drain spacing on subsurface drainage performance in a shallow, slowly permeable soil. *Canadian Agricultural Engineering*, *37*, 9–12.
- Madramootoo, C. A., G. T. Dodds, and A. Papadopoulos. 1993. Agronomic and environmental benefits of water table management. *Journal of Irrigation and Drainage Engineering* 119 (6): 1052–1065.
- Madramootoo, C. A., Helwig, T. G., & Dodds, G. T. (2001). Managing water tables to improve drainage water quality in Quebec, Canada. *Transactions of the ASAE*, 44(6), 1511.
- Madramootoo CA, Johnston WR, Ayars JE, Evans RO, Fausey NR (2007) Agricultural drainage management, quality and disposal issues in North America. Irrig Drain 56:S35–S45
- Malota, M.; Senzanje, A. Modelling mid-span water table depth and drainage discharge dynamics using DRAINMOD 6.1 in a sugarcane field in Pongola, South Africa. J. Water Sa 2015, 41, 325–334.
- Manik, S. M. N., Pengilley, G., Dean, G., Field, B., Shabala, S., & Zhou, M. (2019). Soil and Crop Management Practices to Minimize the Impact of Waterlogging on Crop Productivity. Frontiers in Plant Science, 10(February), 1–23.
- Manitoba agricultural services corporation (MASC, 2015). <u>https://www.gov.mb.ca/agriculture/crops/production/seeding-early-manitoba.html</u>

- <u>Manitoba agriculture (2019). https://www.gov.mb.ca/agriculture/crops/insects/print,grasshopper-forecast.html. Assessed 15<sup>th</sup> November 2019</u>
- MASC Manitoba Agricultural Services Corporation. 2020. Annual report 20019-2020.<u>https://www.masc.mb.ca/masc.nsf/annual\_report\_2019\_20.pdf</u>. Assessed 26 October 2021
- Manitoba Soil Fertility Advisory Committee. 2007. Manitoba soil fertility guide. Gov. Manitoba, Winnipeg, MB, Canada.
- Manitoba Agriculture, 2022. Manitoba Agriculture and Agri-food Statistics Farm. Downloaded from <u>https://www.manitoba.ca/agriculture/markets-and-statistics/publications-and-proceedings/pubs/mb-ag-stats-factsheet-2021.pdf on 12-02-2022</u>
- Manitoba agriculture (2021). Soil Management Guide Soil Compaction. <u>https://www.gov.mb.ca/agriculture/environment/soil-management/soil-management/soil-management/soil-management-guide/soil-compaction.html. Assessed 26 October 2021</u>
- MASC Manitoba Agricultural Services Corporation. 2020. Annual report 20019-2020.<u>https://www.masc.mb.ca/masc.nsf/annual\_report\_2019\_20.pdf</u>. Assessed 26 October 2021
- MARD (Manitoba Agriculture and Resource Development), 2021. Cost of Production <u>https://www.gov.mb.ca/agriculture/farm-management/production-economics/cost-of-production.html</u>. Assessed 26 October 2021
- Mante, A. A., Ranjan, R. S., & Bullock, P. (2018). Subsurface drainage for promoting soil strength for field operations in southern Manitoba. *Soil and Tillage Research*, *184*, 261-268.
- Martel, M., Glenn, A., Wilson, H., & Kröbel, R. (2018). Simulation of actual evapotranspiration from agricultural landscapes in the Canadian Prairies. *Journal of Hydrology: Regional Studies*, 15, 105-118.
- Masasi, B., Taghvaeian, S., Gowda, P. H., Marek, G., & Boman, R. (2020). Validation and application of AquaCrop for irrigated cotton in the Southern Great Plains of US. *Irrigation Science*, *38*(5), 593-607.
- Maulé, C., Helgason, W., McGinn, S., & Cutforth, H. (2006). Estimation of standardized reference evapotranspiration on the Canadian Prairies using simple models with limited weather data. Canadian Biosystems Engineering. 48, 1(1)–1-11.
- McGinn, S. M. (2010). Weather and climate patterns in Canada's prairie grasslands. *Arthropods* of Canadian grasslands, 1, 105-119.
- McKenzie, R. H., Bremer, E., Middleton, A. B., Pfiffner, P. G. and Woods, S. A. 2011. Optimum seeding date and rate for irrigated cereal and oilseed crops in southern Alberta. Can. J. Plant Sci. 91: 293–303.
- Mehring, G.H. Kandel,H.J., Ransom,J.K., Schoch,A., Steele D.D. (2015) Spring Wheat Response to Disease Control and Subsurface Drainage Management in the Red River of

the North Valley, USA. *Agricultural Sciences*,06,1220-1231. doi: <u>10.4236/as.2015.610117</u>

- Meng T, Carew R, Florkowski W, Klepacka A (2017) Analyzing temperature and precipitation influences on yield distributions of Canola and spring wheat in Saskatchewan. J Appl Meteorol Climatol 56(4):897–913
- Mkhabela, M. S., & Bullock, P. R. (2012). Performance of the FAO AquaCrop model for wheat grain yield and soil moisture simulation in Western Canada. *Agricultural Water Management*, 110, 16-24.
- Modi, P. A., Fuka, D. R., & Easton, Z. M. (2021). Impacts of climate change on terrestrial hydrological components and crop water use in the Chesapeake Bay watershed. *Journal of Hydrology: Regional Studies*, *35*, 100830.
- Moghadam, H. R. T., Zahedi, H., & Ghooshchi, F. (2011). Oil quality of canola cultivars in response to water stress and super absorbent polymer application. *Pesquisa Agropecuária Tropical*, *41*, 579-586.
- Mohammadighavam, S., & Kløve, B. (2016). Evaluation of DRAINMOD 6.1 for hydrological simulations of peat extraction areas in northern Finland. *Journal of Irrigation and Drainage Engineering*, *142*(11), 04016053.
- Mood, N. S., Parsinejad, M., & Mirzaei, F. (2009). Controlled Drainage Effects on Crop Yield and Water Use Efficiency under Semi-Arid Condition of Iran. In World Environmental and Water Resources Congress 2009: Great Rivers (pp. 4330-4337). ASCE.
- Moriasi, D. N., Gitau, M. W., Pai, N., & Daggupati, P. (2015). Hydrologic and water quality models: performance measures and evaluation criteria. ASABE, 58(6), 1763–1785. https://doi.org/10.13031/trans.58.10715
- Morrison, M. J., & Stewart, D. W. (2002). Heat stress during flowering in summer Brassica. Crop Science, 42(3), 797–803. https://doi.org/10.2135/crops ci2002.0797
- Morrison, J. C.A. Madramootoo, and M. Chikhaoui (2014). Modeling agricultural land drainage under spring snowmelt conditions with DRAINMOD. Can. J. Civ. Eng. 41: 275-284,
- Morrison, M. J., Harker, K. N., Blackshaw, R. E., Holzapfel, C. J., & O'Donovan, J. T. (2016). Canola yield improvement on the Canadian Prairies from 2000 to 2013. Crop and Pasture Science, 67(4), 245. https://doi.org/10.1071/CP15348
- Mourtzinis, S., Andrade, J. F., Grassini, P., Edreira, J. I. R., Kandel, H., Naeve, S., ... & Conley, S. P. (2021). Assessing benefits of artificial drainage on soybean yield in the North Central US region. *Agricultural Water Management*, 243, 106425.
- Mousavi SF, Soltani-Gerdefaramarzi S, Mostafazadeh-Fard B (2010) Effects of partial rootzone drying on yield, yield components and irrigation water use efficiency of canola (*Brassica napus* L.). Paddy Water Environ 8:157–163
- Naghavi MR., Khalili M. and Aboughadareh AP. (2015), Effect of water deficit stress on yield and yield components of canola (Brassica napus L.) cultivars. International Journal of Agronomy and Agricultural Research, 7(2): 8-13.

- Nangia, V., Sunohara, M. D., Topp, E., Gregorich, E. G., Drury, C. F., Gottschall, N., & Lapen, D. R. (2013). Measuring and modeling the effects of drainage water management on soil greenhouse gas fluxes from corn and soybean fields. *Journal of Environmental Management*, 129, 652–664. https://doi.org/10.1016/j.jenvman.2013.05.040
- Ndulue, E., Onyekwelu, I., Ogbu, K. N. & Ogwo, V. (2019). Performance evaluation of solar radiation equations for estimating reference evapotranspiration (ETo) in a humid tropical environment. Journal of Water and Land Development. 42 (7–9), 124–135.
- Ndulue, E., Onyekwelu, I., Okechukwu, M., Anyadike, C., Echiegu, E. (2020). Sensitivity of FAO Penman–Monteith reference evapotranspiration (ET<sub>o</sub>) to climatic variables under different climate types in Nigeria, Journal of Water and Climate Change. https://doi.org/10.2166/wcc.2020.200
- Ndulue, E., & Sri Ranjan, R. S. (2021). Performance of the FAO Penman-Monteith equation under limiting conditions and fourteen reference evapotranspiration models in southern Manitoba. *Theoretical and Applied Climatology*, 143(3), 1285-1298.
- Ndulue E. and Sri Ranjan (2022). DRAINMOD simulation of drain spacing impact on canola yield in heavy clay soils in the Canadian Prairies. Irrigation and Drainage (Accepted). DOI: 10.1002/ird.2683
- Negm, L. M., Youssef, M. A., Skaggs, R. W., Chescheir, G. M., & Kladivko, E. J. (2014). DRAINMOD-DSSAT simulation of the hydrology, nitrogen dynamics, and plant growth of a drained corn field in Indiana. *Journal of Irrigation and Drainage Engineering*, 140(8), 04014026.
- Ng, H.Y. F., Tan C S, Drury C F, Gaynor J D. 2001. Controlled drainage and subirrigation influences tile nitrate loss and corn yields in a sandy loam soil in Southwestern Ontario. Agric. Ecosyst. Environ. 1758: 1-8
- Nistor, M., Cheval, S., Gualtieri, A. F., Dumitrescu, A., Boţan, V. E., Berni, A., Hognogi, G., Irimuş, I. A. and Porumb-Ghiurco, C. G. (2017), Crop evapotranspiration assessment under climate change in the Pannonian basin during 1991–2050. Meteorological Applications, 24: 84-91.
- Ngezimana, W., & Agenbag, G. A. (2015). The effect of nitrogen and sulphur on the agronomical and water use efficiencies of canola (Brassica napus L.) grown in selected localities of the Western Cape province, South Africa. South African Journal of Plant and Soil, 32(2), 71–76. https://doi.org/10.1080/02571862.2014.988302
- Ngongondo Cosmo, Chong-Yu Xu, Lena M. Tallaksen, Berhanu Alemaw (2013). Evaluation of the FAO Penman-Monteith, Priestley–Taylor and Hargreaves models for estimating reference evapotranspiration in southern Malawi. *Hydrology Research*; 44 (4): 706–722.
- Nouri, M., Homaee, M. & Bannayan, M. (2017) Quantitative trend, sensitivity and contribution analyses of reference evapotranspiration in some arid environments under climate change. Water Resources Management 31, 2207–2224

- NRCS Natural Resources Conservation Service. 2001. Water Management (Drainage). National Engineering Handbook, Part 650, Chapter 14. Retrieved from https://directives.sc.egov.usda.gov/OpenNonWebContent.aspx?content=17551.wba
- O'Donovan, J. T. (1994). Canola (Brassica rapa) Plant Density Influences Tartary Buckwheat (Fagopyrum tataricum) Interference, Biomass, and Seed Yield. Weed Science. 42: 385– 389
- Omidi, H., Tahmasebi, Z., Badi, H. A. N., Torabi, H., & Miransari, M. (2010). Fatty acid composition of canola (Brassica napus L.), as affected by agronomical, genotypic and environmental parameters. *Comptes rendus biologies*, *333*(3), 248-254.
- Orth, R., Staudinger, M., Seneviratne, S. I., Seibert, J., & Zappa, M. (2015). Does model performance improve with complexity? A case study with three hydrological models. *Journal of Hydrology*, *523*, 147-159
- Ozer H 2003: Sowing date and nitrogen rate effects on growth, yield and yield components of two summer rapeseed cultivars. Europ. J. Agron., 19, 453–463. doi:10.1016/S1161-0301(02)00136-3
- Pagán, B. R., Ashfaq, M., Rastogi, D., Kendall, D. R., Kao, S. C., Naz, B. S., ... & Pal, J. S. (2016). Extreme hydrological changes in the southwestern US drive reductions in water supply to Southern California by mid-century. *Environmental Research Letters*, 11(9), 094026.
- Paparrizos, S., Maris, F., & Matzarakis, A. (2014). Estimation and Comparison of Potential Evapotranspiration Based on Daily and Monthly Data From Sperchios Valley in Central Greece. Global NEST Journal, 16(2), 204–217.
- Pandey, P.K.; Dabral, P.P.; Pandey, V. 2016. Evaluation of reference evapotranspiration methods for the northeastern region of India. Int. Soil Water Conserv. Res., 4, 52–63.
- Pavlista, A. D., Santra, D. K., Isbell, T. A., Baltensperger, D. D., Hergert, G. W., Krall, J., ... & Berrada, A. (2011). Adaptability of irrigated spring canola oil production to the US High Plains. *Industrial Crops and Products*, 33(1), 165-169.
- Pavlista, A.D., Hergert, G.W., Margheim, J.F., Isbell, T.A., 2016. Growth of spring Canola (Brassica napus) under deficit irrigation in Western Nebraska. Ind. Crops Prod. 33, 451– 456, http://dx.doi.org/10.1016/j.indcrop.2015.12.059.
- Patle, G. T., & Singh, D. K. (2015). Sensitivity of annual and seasonal reference crop evapotranspiration to principal climatic variables. *Journal of Earth System Science*, *124*(4), 819-828.
- Pease, L. A., Fausey, N. R., Martin, J. F., & Brown, L. C. (2017). Projected climate change effects on subsurface drainage and the performance of controlled drainage in the Western Lake Erie Basin. *Journal of Soil and Water Conservation*, 72(3), 240-250.
- Pease, L.A., King, K.W., Williams, M.R., LaBarge, G.A., Duncan, E.W., Fausey, N.R., 2018. Phosphorus export from artificially drained fields across the Eastern Corn Belt. J. Gt. Lakes Res. 44 (1), 43–53. <u>https://doi.org/10.1016/j.jglr.2017.11.009</u>.

- Peng, L., Li, Y., & Feng, H. (2017). The best alternative for estimating reference crop evapotranspiration in different sub-regions of mainland China. *Scientific reports*, 7(1), 1-19.
- Penman L.H. 1948. Natural evaporation from open water, bare soil, and grass. Proceedings of Royal Meteorological Society 193: 120–145.
- Ploschuk, R. A., Miralles, D. J., Colmer, T. D., Ploschuk, E. L., & Striker, G. G. (2018).
   Waterlogging of Winter Crops at Early and Late Stages: Impacts on Leaf Physiology, Growth and Yield. Frontiers in Plant Science, 9, 1863.
   <u>https://doi.org/10.3389/fpls.2018.01863</u>
- Phogat, V., M.A. Skewes, J.W. Cox, G. Sanderson, J. Alam, and J. Šimůnek. 2014. Seasonal simulation of water, salinity, and nitrate dynamics under drip irrigated mandarin (Citrus reticulata) and assessing management options for drainage and nitrate leaching. J. Hydrol. 513:504–516. doi:10.1016/j.jhydrol.2014.04.008
- Poddar, A., Gupta, P., Kumar, N., Shankar, V. & Ojha, C. S. P. (2018). Evaluation of reference evapotranspiration methods and sensitivity analysis of climatic parameters for sub-humid sub-tropical locations in western Himalayas (India). ISH Journal of Hydraulic Engineering 1–11. doi:10.1080/09715010.2018.1551731.
- Podolsky, G. (1982) Soils of the West Interlake area. CANADAMANITOBA Soil Survey. Winnipeg, MB: Agriculture Canada, Manitoba Department of Agriculture, Department of Soil Science. ENCE, University of Manitoba. <u>https://sis.agr.gc.ca/cansis/publications/surveys/mb/mbd36/mbd36\_report.pdf</u>
- Poole, C. A., Skaggs, R. W., Cheschier, G. M., Youssef, M. A., & Crozier, C. R. (2013). Effects of drainage water management on crop yields in North Carolina. *Journal of Soil and Water Conservation*, 68(6), 429–437. https://doi.org/10.2489/jswc.68.6.429
- Popova Z, Kercheva M, Pereira L (2006). Validation of the FAO methodology for computing ETo with limited data. Application to south Bulgaria. Irrigation and Drainage, 55(2):201–215
- Pokharel, M., Chiluwal, A., Stamm, M., Min, D., Rhodes, D., & Jagadish, S. K. (2020). High night-time temperature during flowering and pod filling affects flower opening, yield and seed fatty acid composition in canola. *Journal of Agronomy and Crop Science*, 206(5), 579-596.
- Potter TD , Kay JR , Ludwig IR , Frischke BM (2007) Effect of row spacing and sowing rate on canola cultivars in low rainfall areas. J of Oil Seeds Res 52(3): 189-195
- Prasad, P.V.V., Staggenborg, S.A., and Ristic, Z. 2008. Pages 301–355 in L.R. Ahuja, V.R. Reddy, S.A. Saseendran, and Q. Yu, eds. Response of crops to limited water: understanding and modeling water stress effects on plant growth processes. Impacts of drought and/or heat stress on physiological, developmental, growth, and yield processes of crop plants. Adv. Agric. Syst. Model. 1. ASA, CSSA, SSSA, Madison, WI.

- Preeyaphorn Kosa (2011). The Effect of Temperature on Actual Evapotranspiration based on Landsat 5 TM Satellite Imagery, Evapotranspiration, Prof. Leszek Labedzki (Ed.), ISBN: 978-953-307-251-7, InTech, Available from http://www.intechopen.com/books/evapotranspiration/the-effect-of-temperature-onactualevapotranspiration-based-on-landsat-5-tm-satellite-imagery
- Priestley CHB, Taylor R.J. (1972). On the assessment of surface heat flux and evaporation using large-scale parameters. Monthly Weather Reviews 100:81–92.
- Pritchard, F. M., Eagles, H. A., Norton, R. M., Salisbury, P. A., and Nicolas, M. (2000). Environmental effects on seed composition of Victorian canola. *Aust. J. Exp. Agric.* 40, 679–685.
- Qian, B., Gameda, S., Zhang, X., & De Jong, R. (2012). Changing growing season observed in Canada. *Climatic Change*, *112*(2), 339-353.
- Qian, B., De Jong, R., Huffman, T., Wang, H., & Yang, J. (2016). Projecting yield changes of spring wheat under future climate scenarios on the Canadian Prairies. *Theoretical and* applied climatology, 123(3-4), 651-669.
- Qian, B., Jing, Q., Bélanger, G., Shang, J., Huffman, T., Liu, J., & Hoogenboom, G. (2018). Simulated canola yield responses to climate change and adaptation in Canada. *Agronomy journal*, 110(1), 133-146.
- Qian, B., Zhang, X., Smith, W., Grant, B., Jing, Q., Cannon, A. J., ... & Zhao, J. (2019). Climate change impacts on Canadian yields of spring wheat, canola and maize for global warming levels of 1.5 C, 2.0 C, 2.5 C and 3.0 C. *Environmental Research Letters*, 14(7), 074005.
- Qiu, J., Shen, Z., Hou, X., Xie, H., & Leng, G. (2020). Evaluating the performance of conservation practices under climate change scenarios in the Miyun Reservoir Watershed, China. *Ecological Engineering*, 143, 105700.
- Rabalais, N.N., Turner, R.E., Sen, B.K., Gupta, D.F., Boesch, P., Chapman Murrell, M.C. 2007. Hypoxia in the northern Gulf of Mexico: does the science support the plan to reduce, mitigate and control hypoxia? Estuaries Coasts 30, 753–772.
- Ranatunga, K., Nation, E. R., & Barratt, D. G. (2008). Review of soil water models and their applications in Australia. *Environmental Modelling & Software*, 23(9), 1182-1206.
- Randall, G.W., and M.J. Goss. 2001. Nitrate losses to surface water through subsurface, tile drainage. In Nitrogen in the Environment: Sources, Problems and Management, eds. R.F. Follet and J.L. Hatfield,95-122. Amsterdam: Elsevier B.V.
- Randall, G. W., & Vetsch, J. A. (2005). Nitrate losses in subsurface drainage from a cornsoybean rotation as affected by fall and spring application of nitrogen and nitrapyrin. *Journal of Environmental Quality*, *34*(2), 590-597.
- Randall, G. W., and Goss, M. J.,2008. Chapter 6. Nitrate Losses to Surface Water Through Subsurface in Nitrogen in the Environment: Sources, Problems, and Management J.L. Hatfield and R.F. Follett (Eds)145–175. https://doi.org/10.1177/0047287510382295

- Rashid, M., Hampton, J. G., Rolston, M. P., Khan, K. M., & Saville, D. J. (2018). Heat stress during seed development affects forage brassica (Brassica napus L.) seed quality. *Journal* of agronomy and crop science, 204(2), 147-154.
- Rathke, G. W., Christen, O., & Diepenbrock, W. (2005). Effects of nitrogen source and rate on productivity and quality of winter oilseed rape (Brassica napus L.) grown in different crop rotations. *Field crops research*, *94*(2-3), 103-113.
- Ravazzani, G., Corbari C., Morella, S., Gianoli, P., Mancini, M., 2012. Modified Hargreaves-Samani equation for the assessment of reference evapotranspiration in Alpine River Basins Journal of Irrigation and Drainage Engineering 138 (7), 592–599.
- Raza, A., Razzaq, A., Mehmood, S. S., Zou, X., Zhang, X., Lv, Y., & Xu, J. (2019). Impact of climate change on crops adaptation and strategies to tackle its outcome: A review. *Plants*, 8(2), 34.
- Razzaghi, F., and Sepaskhah, A. R. 2011. Calibration and validation of four common ETo estimation equations by lysimeter data in a semi-arid environment. Archives of Agronomy and Soil Science, 58(3), 303-319.
- Romanenko VA (1961) Computation of the autumn soil moisture using a universal relationship for a large area. In: Proceedings, Ukrainian Hydrometeorological Research Institute, no. 3. Kie
- Romero-Lankao, P., J.B. Smith, D.J. Davidson, N.S. Diffenbaugh, P.L. Kinney, P. Kirshen, P. Kovacs, and L. Villers Ruiz, 2014: North America. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Barros, V.R., C.B. Field, D.J. Dokken, M.D. Mastrandrea, K.J. Mach, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L. White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 1439-1498
- Rosa, L., Chiarelli, D. D., Sangiorgio, M., Beltran-Peña, A. A., Rulli, M. C., D'Odorico, P., & Fung, I. (2020). Potential for sustainable irrigation expansion in a 3 C warmer climate. *Proceedings of the National Academy of Sciences*, 117(47), 29526-29534.
- Rosenzweig C, Parry ML (1994) Potential impact of climate change on world food supply. Nature 367:133–138
- Saadat, S.; Frankenberger, J.; Bowling, L.; Ale, S. (2020). Evaluation of Surface Ponding and Runoff Generation in a Seasonally Frozen Drained Agricultural Field. J. Hydrol. 588, 124985.
- Salazar, O., Wesström, I., & Joel, A. (2008). Evaluation of DRAINMOD using saturated hydraulic conductivity estimated by a pedotransfer function model. *Agricultural water management*, 95(10), 1135-1143.

- Satchithanantham, S., & Ranjan, R. S. (2015). Evaluation of DRAINMOD for Potato Crop under Cold Conditions in the Canadian Prairies. *Transactions of the ASABE*, (October), 307–317. https://doi.org/10.1002/pssr.200903283
- Satchithanantham, S. 2013. Agronomic and environmental impacts of potato production under different management strategies in the Canadian Prairies. PhD Thesis. University of Manitoba, Canada
- Satchithanantham, S., Sri Ranjan, R., & Bullock, P. (2014). Protecting water quality using controlled drainage as an agricultural BMP for potato production. *Transactions of the ASABE*, *57*(3), 815-826.
- Sands, G. R., Canelon, D., & Talbot, M. (2015). Developing optimum subsurface drainage design procedures. Acta Agriculturae Scandinavica, Section B—Soil & Plant Science, 65(sup1), 121-127.
- Sauchyn, D. and Kulshreshtha, S. (2008): Prairies; in From impacts to Adaptation: Canada in a changing climate 2007. Edited by D.S. Lemmen, F.J. Warren, J. Lacroix and E. Bush; Government of Canada, Ottawa, ON., p.275-328
- Schaap, M. G., Leij, F. J., & Van Genuchten, M. T. (2001). Rosetta: A computer program for estimating soil hydraulic parameters with hierarchical pedotransfer functions. *Journal of hydrology*, 251(3-4), 163-176.
- Schendel U (1967) Vegetationswasserverbrauch und -wasserbedarf. Habilitation, Kiel, p 137
- Scherer, T. F., Sands, G. R., Kandel, H., & Hay, C. H. (2015). *Frequently asked questions about subsurface (tile) drainage*. NDSU Extension Service.
- Schilling, J., Hertig, E., Tramblay, Y., & Scheffran, J. (2020). Climate change vulnerability, water resources and social implications in North Africa. *Regional Environmental Change*, 20(1), 1-12.
- Schindler DW, Hecky RE, McCullough GK. 2012. The rapid eutrophication of Lake Winnipeg: Greening under global change. Journal of Great Lakes Research, 38: 6-13. DOI: 10.1016/j.jglr.2012.04.003.
- Schneider K., Ketzer B., Breuer, L., Vache, K.B., Bernhofer, C., and Frede, H.G. (2007). Evaluation of evapotranspiration methods for model validation in a semi-arid watershed in northern China. Advanced Geosciences, 11, 37–42
- Schott, L., Lagzdins, A., Daigh, A. L. M., Craft, K., Pederson, C., Brenneman, G., & Helmers, M. J. (2017). Drainage water management effects over five years on water tables, drainage, and yields in southeast Iowa. Journal of Soil and Water Conservation, 72(3), 251–259. <u>https://doi.org/10.2489/jswc.72.3.251</u>
- Seiller, G., & Anctil, F. (2016). How do potential evapotranspiration formulas influence hydrological projections?. *Hydrological Sciences Journal*, *61*(12), 2249-2266.

- Sentelhas, P. C., Gillespie, T. J., & Santos, E. A. (2010). Evaluation of FAO Penman-Monteith and alternative methods for estimating reference evapotranspiration with missing data in Southern Ontario, Canada. Agricultural Water Management, 97(5), 635–644.
- Setyanto, P., Pramono, A., Adriany, T. A., Susilawati, H. L., Tokida, T., Padre, A. T., & Minamikawa, K. (2018). Alternate wetting and drying reduces methane emission from a rice paddy in Central Java, Indonesia without yield loss. *Soil Science and Plant Nutrition*, 64(1), 23-30.
- Shukla, M. B., Prasher, S. O., Madani, A., & Gupta, G. P. (1994). Field validation of DRAINMOD in Atlantic Canada. Canadian Agric. Eng., 36(4), 205-213.
- Shafieiyoun E, Gheysari M, Khiadani M, Koupai JA, Shojaei P, Moomkesh M. (2020). Assessment of reference evapotranspiration across an arid urban environment having poor data monitoring system. Hydrological Processes, 34:4000–4016. https://doi.org/10.1002/hyp.13851
- Sharma V, and Irmak S. (2012). Mapping Spatially Interpolated Precipitation, Reference Evapotranspiration, Actual Crop Evapotranspiration, and Net Irrigation Requirements in Nebraska. Part II. Actual Crop Evapotranspiration and Net Irrigation Requirements. *Transactions of ASABE*, 55(3) 907-921
- Shuttleworth, W.J. (2008). Evapotranspiration measurement methods. Southwest Hydrology 7, 22–23.
- Singh, M., Prasher, S.O., Tan, C., Tejawat, C. (1994). Evaluation of DRAINMOD for Southern Ontario conditions. Can. Water Resour., 19(4):313-326.
- Singh, V. P., & Xu, C.Y. (1997). Evaluation and Generalization of 13 Mass-Transfer Equations for Determining Free Water Evaporation. Hydrological Processes, 11(3):311–323.
- Singh, R., Helmers, M. J., & Qi, Z. (2006). Calibration and validation of DRAINMOD to design subsurface drainage systems for Iowa's tile landscapes. Agricultural water management, 85(3), 221-232.
- Singh B, Rengel Z, Bowden JW 2006: Carbon, nitrogen and sulphur cycling following incorporation of canola residue of different sizes into a nutrient-poor sandy soil. Soil Biol. Biochem., 38, 32–42. doi:10.1016/j. soilbio.2005.03.025
- Singh, R., Helmers, M. J., Kaleita, A. L., & Takle, E. S. (2009). Potential impact of climate change on subsurface drainage in Iowa's subsurface drained landscapes. *Journal of irrigation and drainage engineering*, 135(4), 459-466.
- Singh, G. & Nelson, K.A. (2021) Long-term drainage, subirrigation, and tile spacing effects on maize production. Field Crops Research, 262, 108032. Available from: https://doi.org/10.1016/j.fcr.2020.108032
- Skaggs R.W., 1978. A water management model for shallow water Table soils. In: Tech. Rep. 134. University of North Carolina Water Resource Research Institute, Raleigh, NC.

- Skaggs RW, Breve MA, Gilliam JW. 1994. Hydrologic and water quality impacts of agricultural drainage. Crit Rev Environ Sci Tech. 24:1–32
- Skaggs, R. W., M. A. Youssef, G. M. Chescheir, and J. W. Gilliam. 2005. Effect of drainage intensity on nitrogen losses from drained lands. Trans. ASABE 48(6): 2169-2177
- Skaggs, R. W., M. A. Mohamed, and G. M. Chescheir. 2006. Drainage design coefficients for eastern United States. Agric. Water Mgmt. 86(1-2): 40-49.
- Skaggs, R. W., Youssef, M. A., Gilliam, J. W., & Evans, R. O. (2010). Effect of controlled drainage on water and nitrogen balances in drained lands. *Transactions of the* ASABE, 53(6), 1843-1850.
- Skaggs, R. W., Fausey, N. R., & Evans, R. O. (2012). Drainage water management. Journal of soil and water conservation, 67(6), 167A-172A.
- Skaggs, R. W., Youssef, M. A., & Chescheir, G. M. (2012). DRAINMOD: Model use, calibration, and validation. Trans. ASABE, 55(4), 1509-1522. <u>http://dx.doi.org/10.13031/2013.42259</u>.
- Sharpley, A.N., and Menzel, R.G. (1987). The impact of soil and fertilizer phosphorus on the environment. Adv. Agron., 41, 297–324
- Singer, S. D.; Zou, J. T.; Weselake, R. J. (2016). Abiotic factors influence plant storage lipid accumulation and composition. Plant Sci., 243, 1–9
- Smith, R. E., Michalyna, W., & Wilson, G. (1973). Soils of the MQRDEN-WINKLER AREA. Manitoba soil survey report, No. 18.
- Smith, W. N., Grant, B. B., Desjardins, R. L., Kroebel, R., Li, C., Qian, B., ... & Drury, C. F. (2013). Assessing the effects of climate change on crop production and GHG emissions in Canada. Agriculture, ecosystems & environment, 179, 139-150.
- Smith, D. R., King, K. W., Johnson, L., Francesconi, W., Richards, P., Baker, D., & Sharpley, A. N. (2015). Surface Runoff and Tile Drainage Transport of Phosphorus in the Midwestern United States. Journal of Environment Quality, 44(2), 495. <u>https://doi.org/10.2134/jeq2014.04.0176</u>
- Smith, E., Vosman, A., Kellman, L., & Rodd, V. (2019). Impact of drainage type on simultaneous nitrogen losses in Atlantic Canada. *Canadian Journal of Soil Science*, 79(January), 70–79. https://doi.org/10.1139/cjss-2018-0115
- Sojka, M., Kozłowski, M., Kęsicka, B., Wróżyński, R., Stasik, R., Napierała, M., ... & Liberacki, D. (2020). The effect of climate change on controlled drainage effectiveness in the context of groundwater dynamics, surface, and drainage outflows. Central-Western Poland case study. *Agronomy*, 10(5), 625.
- Stamm, C., Fluhler, H., Gachter, R., Leuenberger, J., and Wunderli, H. (1998). Preferential transport of phosphorus in drained grassland soils. J. Environ. Qual., 27, 515–522

Statistics Canada 2016 <u>https://www150.statcan.gc.ca/n1/daily-quotidien/170510/dq170510a-eng.htm</u> Assessed 30th March 2019

Statistics Canada 2017. World Water Day... by the numbers

https://www.statcan.gc.ca/eng/dai/smr08/2017/smr08\_215\_2017.Assessed 30th March 2019

## Statistics Canada 2018. <u>https://grainscanada.gc.ca/en/grain-research/export-</u> <u>quality/oilseeds/canola/2018/03-weather.html</u>. Assessed 15<sup>th</sup> October 2019

- Statistics Canada, 2018. Canola: a Canadian success story <u>https://www150.statcan.gc.ca/n1/pub/96-325-x/2007000/article/10778-eng.htm</u> (23/09/2018)
- Statistics Canada. <u>Table 32-10-0359-01</u> Estimated areas, yield, production, average farm price and total farm value of principal field crops, in metric and imperial units
- Stolzy, L. H. and Sojka, R. E. 1984. Effects of flooding on plant disease. Pages 221264 in T. T. Kozlowski, ed. Flooding and plant growth. Academic Press, Orlando, FL.
- Strock, J.S., P.J.A. Kleinman, K.W. King, and J.A. Delgado. 2010. Drainage water management for water quality protection. J. Soil Water Conserv. 65:131A–136A.
- Sunohara, M.D., Craiovan, E., Topp, E., Gottschall, N., Drury, C.F., and Lapen, D.R. 2014. Comprehensive nitrogen budgets for controlled tile drainage fieldsinEastern Ontario, Canada. J. Environ. Qual. 43:617–630.
- Sunohara, M. D., Gottschall, N., Craiovan, E., Wilkes, G., Topp, E., Frey, S. K., & Lapen, D. R. (2016). Controlling tile drainage during the growing season in Eastern Canada to reduce nitrogen, phosphorus, and bacteria loading to surface water. *Agricultural Water Management*, 178, 159-170.
- Suzuki, N., Rivero, R. M., Shulaev, V., Blumwald, E., & Mittler, R. (2014). Abiotic and biotic stress combinations. *New Phytologist*, 203(1), 32-43.
- Tabari, H., & Grismer, M. E. & S. T. (2013). Comparative analysis of 31 reference evapotranspiration methods under humid conditions. Irrigation Science 31(2):107–117
- Tabari, H. & Talaee, P. H. (2014). Sensitivity of evapotranspiration to climatic change in different climates. Global and Planetary Change 115,16–23
- Tabari H, Hosseinzadeh Talaee P, Willems P, Martinez C (2016). Validation and calibration of solar radiation equations for estimating daily reference evapotranspiration at cool semiarid and arid locations. Hydrol Sci J. doi:10.1080/02626667.2014.947293
- Tan, C. S., Drury, C. F., Ng, H. Y. F. and Gaynor, J. D. 1999. Effect of controlled drainage and sub-irrigation on subsurface tile drainage nitrate loss and crop yield at the farm scale. Can. Water Resour. J. 24: 177?186.

- Tan, C.S., Drury, C.F., Soultani, M., van Wesenbeeck, I.J., Ng, H.Y.F., Gaynor, J.D., Welacky, T.W., 1998. Effect of controlled drainage and tillage on soil structure and tile drainage nitrate loss at the field scale. Wat. Sci. Tech. 38, 103-110.
- Tan, C. S., & Zhang, T. Q. (2011). Surface runoff and sub-surface drainage phosphorus losses under regular free drainage and controlled drainage with sub-irrigation systems in southern Ontario. *Canadian Journal of Soil Science*, 91(3), 349–359.
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. *Bulletin of the American meteorological Society*, *93*(4), 485-498.
- Tesfamariam, E.H., Annandale, J.G., Steyn, J.M., 2010. Water stress effects on winter Canola growth and yield. Agron. J. 102, 658–666.
- Thiery, W., Visser, A. J., Fischer, E. M., Hauser, M., Hirsch, A. L., Lawrence, D. M., ... & Seneviratne, S. I. (2020). Warming of hot extremes alleviated by expanding irrigation. *Nature communications*, 11(1), 1-7.
- Thornthwaite, C.W., 1948. An approach toward a rational classification of climate. Geographical Review 38, 55–94.
- Thorp, K. R., Jaynes, D. B., & Malone, R. W. (2008). Simulating the long-term performance of drainage water management across the Midwestern United States. *Transactions of the ASABE*, *51*(3), 961-976.
- Tohidi-Moghadam HR, Zahedi H, Ghooshchi F (2011) Oil quality of colza cultivars in response to water stress and superabsorbent polymer application. Pesqui Agropecuária Trop 41:579– 586.https://doi.org/10.5216/pat.v41i4.13366
- Tolomio, M., & Borin, M. (2018). Water table management to save water and reduce nutrient losses from agricultural fields: 6 years of experience in North-Eastern Italy. Agricultural Water Management, 201(June 2017), 1–10. https://doi.org/10.1016/j.agwat.2018.01.009
- Trajkovic S (2007) Hargreaves versus Penman–Monteith under Humid Condition. J Irrig Drain Eng ASCE 133:38–42
- Trajkovic, S. and Kolakovic, S. (2009) Evaluation of Reference Evapotranspiration Equations under Humid Conditions. Water Resource Management, 23, 3057-3067. <u>https://doi.org/10.1007/s11269-009-9423-4</u>
- UNESCO, 1987. The process of water resources project planning: a systems approach. Haimes, Y.Y. et al. Studies and reports in hydrology, No. 44, Paris
- USDA 2022 United States Department of Agriculture Foreign Agricultural Service Oilseeds: World Markets and Trade. Downloaded https://apps.fas.usda.gov/psdonline/circulars/oilseeds.pdf 13-02-2022.
- USGCRP (2018) Impacts, risks, and adaptation in the United States:Fourth National Climate Assessment, Vol II . In: Reidmiller DR, Avery CW, Easterling DR, Kunkel KE, Lewis

KLM, Maycock TK, Stewart BC (eds) US Global Change Research Program, Washington, DC, pp 1515. https://doi.org/10.7930/NCA4.2018

- Vadas, P., P. Kleinman, A. Sharpley, and B. Turner. 2005. Relating soil phosphorus to dissolved phosphorus in runoff. J. Environ. Qual. 34:572–580. doi:10.2134/jeq2005.0572
- Valiantzas, J. D. (2013). Simple ETo Forms of Penman's Equation without Wind and/or Humidity Data. II: Comparisons with Reduced Set-FAO and Other Methodologies. Journal of Irrigation and Drainage Engineering, 139(January), 9–19.
- Valiantzas, J. D. (2013). Simplified forms for the standardized FAO-56 Penman-Monteith reference evapotranspiration using limited weather data. Journal of Hydrology, 505, 13–23.
- Valipour, M. (2014). Importance of solar radiation, temperature, relative humidity, and wind speed for calculation of reference evapotranspiration. Archives of Agronomy and Soil Science, 340, 1–17.
- Valle Júnior, L. C. G., Ventura, T. M., Souza, R. S. R., de Nogueira, J. S., de Lobo, F. A., Vourlitis, G. L., & Rodrigues, T. R. (2020). Comparative assessment of modelled and empirical reference evapotranspiration methods for a Brazilian savanna. Agricultural Water Management, 232, 106040. <u>https://doi.org/10.1016/j.agwat.2020.106040</u>.
- Van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., ... & Rose,
  S. K. (2011). The representative concentration pathways: an overview. *Climatic change*, *109*(1), 5-31.
- Van Zandvoort A, Lapen DR, Clark LD, Flemming C, Craiovan E, Sunohara MD, Boutz R, Gottschall N (2017) Soil CO<sub>2</sub>, CH<sub>4</sub>, and N<sub>2</sub>O fluxes over and between tile drains on corn, soybean, and forage fields under tile drainage management. Nutr Cycl Agroecosyst 109:115–132
- Vanderleest, C. P., Caron, J., & Bland, W. L. (2016). Water table level management as an irrigation strategy for cranberry (Vaccinium macrocarpon Aiton). *Canadian Journal of Soil Science*, 97(1), 11-19.
- Vepraskas, M. J., Skaggs, R. W., & Caldwell, P. V. (2020). Method to assess climate change Impacts on hydrologic boundaries of individual wetlands. *Wetlands*, 40(2), 365-376.
- Vigil MF, Anderson RL, Beard WE. Base temperature and growing-degree-hour requirements for the emergence of canola. Crop Sci. 1997. 37:844-9.
- Wagena, M. B., & Easton, Z. M. (2018). Agricultural conservation practices can help mitigate the impact of climate change. *Science of The Total Environment*, 635, 132-143
- Wahba MAS, Christen EW. 2006. Modeling subsurface drainage for salt load management in southeastern Australia. Irrigation and Drainage Systems 20(2–3): 267–282

- Wanchuk, R. N., & Apedaile, L. P. (1988). Economic Performance of Surface and Subsurface Drainage. *Canadian Water Resources Journal*, 13(2), 48–55. https://doi.org/10.4296/cwrj1302048
- Wang, X., Mosley, C.T., Frankenberger, J.R., Kladivko, E.J., (2006). Subsurface drain flow and crop yield predictions for different drain spacing using DRAINMOD. Agric. Water Manage. 79 (2), 113–136
- Wang, H., He, Y., Qian, B., McConkey, B., Cutforth, H., McCaig, T., ... & Hoogenboom, G. (2012). Climate change and biofuel wheat: A case study of southern Saskatchewan. *Canadian Journal of plant science*, 92(3), 421-425.
- Wang, Z., Qi, Z., Xue, L., Bukovsky, M., & Helmers, M. J. (2015). Modeling the impacts of climate change on nitrogen losses and crop yield in a subsurface drained field. *Climatic Change*, 129(1), 323-335.
- Wang, Z., Shao, G., Lu, J., Zhang, K., Gao, Y., & Ding, J. (2020). Effects of controlled drainage on crop yield, drainage water quantity and quality: A meta-analysis. *Agricultural Water Management*, 239, 106253.
- Wang, Z., Shao, G., Lu, J., Zhang, K., Gao, Y., & Ding, J. (2020). Effects of controlled drainage on crop yield, drainage water quantity and quality: A meta-analysis. Agricultural Water Management, 239, 106253.
- Werner, A. T., & Cannon, A. J. (2016). Hydrologic extremes–an intercomparison of multiple gridded statistical downscaling methods. *Hydrology and Earth System Sciences*, 20(4), 1483-1508.
- Wesström, I., Joel, A., & Messing, I. (2014). Controlled drainage and subirrigation–A water management option to reduce non-point source pollution from agricultural land. *Agriculture, ecosystems & environment, 198,* 74-82.
- Wiesler, F., 1998. Comparative assessment of the efficacy of various nitrogen fertilizers. J. Crop Prod. 1, 81–114
- Wilby RL, Wigley TML (1997) Downscaling general circulation model output: a review of methods and limitations. Prog Phys Geogr 21 (4): 530-548.
- Willmott, C. J., & Matsuura, K. (2005). Advantages of the mean absolute error (MAE) over the root mean square error (RMSE) in assessing average model performance. *Climate research*, 30(1), 79-82.
- Wilson, H. F., Satchithanantham, S., Moulin, A. P., & Glenn, A. J. (2016). Soil phosphorus spatial variability due to landform, tillage, and input management: A case study of small watersheds in southwestern Manitoba. *Geoderma*, 280, 14-21.

- Wilson, H., Elliott, J., Macrae, M., & Glenn, A. (2019). Near-surface soils as a source of phosphorus in snowmelt runoff from cropland. *Journal of environmental quality*, 48(4), 921-930.
- Wilson, G. L., Mulla, D. J., Vetsch, J. A., & Sands, G. R. (2020). Predicting nitrate-nitrogen loads in subsurface drainage as a function of fertilizer application rate and timing in southern Minnesota. J. Environ. Qual. 49(5):1347-1358).
- Wollmer, A. C., Pitann, B., & Mühling, K. H. (2018). Waterlogging events during stem elongation or flowering affect yield of oilseed rape (Brassica napus L.) but not seed quality. *Journal* of Agronomy and Crop Science, 204(2), 165-174.
- Wright, G.C., Smith, C.J. & Woodroofe, M.R. The effect of irrigation and nitrogen fertilizer on rapeseed (*Brassica napes*) production in South-Eastern Australia. Irrig Sci (1988) 9: 1. <u>https://doi.org/10.1007/BF00292139</u>
- Wu, W., Ma, B. L., & Whalen, J. K. (2018). Enhancing rapeseed tolerance to heat and drought stresses in a changing climate: perspectives for stress adaptation from root system architecture. Advances in agronomy, 151, 87-157.
- Xystrakis, F., & Matzarakis, A. (2011). Evaluation of 13 Empirical Reference Potential Evapotranspiration Equations on the Island of Crete in Southern Greece. Journal of Irrigation and Drainage Engineering, 137(4), 211–222.
- Yang, C. C., Prasher, S. O., Wang, S., Kim, S. H., Tan, C. S., Drury, C., & Patel, R. M. (2007). Simulation of nitrate-N movement in southern Ontario, Canada, with DRAINMOD-N. Agric. Water Mgmt., 87(3), 299-306.
- Yang, X. (2008). Evaluation and application of DRAINMOD in an Australian sugarcane field. Agric. Water Manage., 95(4), 439–446.
- Yin, Y., Wu, S., Chen, G. & Dai, E. (2010). Attribution analyses of potential evapotranspiration changes in China since the 1960s. Theoretical and Applied Climatology 101,19–28.
- Yousaf, M., Ahmed, A., Nawaz, N., Sarwar, G., & Roidar, B. (2002). Effect of Different Planting Densities on the Grain Yield of Canola (Sarson) Varieties. Asian Journal of Plant Sciences, 1(4), 332–333. https://doi.org/10.3923/ajps.2002.332.333
- Youssef, M.A.; Skaggs, R.W.; Chescheir, G.M.; Gilliam, J.W. (2005). The nitrogen simulation model, DRAINMOD-N II. Trans. ASABE, 48, 611–626.
- Youssef, M. A., Abdelbaki, A. M., Negm, L. M., Skaggs, R. W., Thorp, K. R., & Jaynes, D. B. (2018). DRAINMOD-simulated performance of controlled drainage across the U.S. Midwest. Agricultural Water Management, 197, 54–66.
- Younes, S., Amir, H. S. R., Amir, A. B., Ghorban, N., & Hossein, Z. (2011). Yield and yield components of six canola (Brassica napus L.) cultivars affected by planting date and water deficit stress. African Journal of Biotechnology, 10(46), 9309–9313. <u>https://doi.org/10.5897/AJB11.048</u>

- Zhang, Y., Chen, W., & Riseborough, D. W. (2008). Transient projections of permafrost distribution in Canada during the 21st century under scenarios of climate change. *Global* and Planetary Change, 60(3-4), 443-456.
- Zhang, S., X. Liao, C. Zhang, and H. Xu. 2012. Influences of plant density on the seed yield and oil content of winter oilseed rape (*Brassica napus* L.). Ind. Crop. Prod. 40:27– 32. doi:https://doi.org/10.1016/j.indcrop.2012.02.016
- Zhou WJ, Lin XQ (1995) Effects of waterlogging at different growth stages on physiological characteristics and seed yield of winter rape (*Brassica napus* L.). Field Crops Res 44:103–110
- Zhou, W., Zhao, D., & Lin, X. (1997). Effects of waterlogging on nitrogen accumulation and alleviation of waterlogging damage by application of nitrogen fertilizer and mixtalol in winter rape (Brassica napus L.). Journal of Plant Growth Regulation, 16(1), 47–53. https://doi.org/10.1007/PL00006974
- Zhou Weijun and Lin Xianqing (1995). Effects of waterlogging at different growth stages on physiological characteristics and seed yield of winter rape (Brassica napus L.). Field Crops Research 44(2-3) 103-110
- Zhou, L., Yan, T., Chen, X., Li, Z., Wu, D., Hua, S., & Jiang, L. (2018). Effect of high night temperature on storage lipids and transcriptome changes in developing seeds of oilseed rape. *Journal of experimental botany*, 69(7), 1721-1733.
- Zou, X., Hu, C., Zeng, L., Cheng, Y., Xu, M., & Zhang, X. (2014). A comparison of screening methods to identify waterlogging tolerance in the field in Brassica napus L. during plant ontogeny. *PloS one*, 9(3), e89731. doi:10.1371/journal.pone.0089731
- Zou, X., Li, Y. E., Gao, Q., & Wan, Y. (2012). How water saving irrigation contributes to climate change resilience—a case study of practices in China. *Mitigation and Adaptation Strategies for Global Change*, *17*(2), 111-132